FRANCHISE DISCLOSURE DOCUMENT



The Glass Guru Enterprises, Inc. A Texas Corporation 5550 Granite Parkway Suite 280 Plano, TX 75024 Phone: 877-654-8507

Email: info@theglassguru.com Website: theglassgurufranchise.com

As a The Glass Guru® Franchisee, you will operate a business that will install, repair, and replace residential and commercial glass, mirrors, showers, windows, and doors, and provide other related services and sell other related products to residential and commercial customers.

The initial investment necessary to begin operation of a The Glass Guru franchised business ranges from \$126,305 to \$219,530. This includes \$36,000 to \$56,000 that must be paid to the Franchisor or its affiliates.

This disclosure document summarizes certain provisions of your Franchise Agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar-days before you sign a binding agreement with, or make any payment to, the Franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no government agency has verified the information contained in this document.**

You may wish to receive your disclosure document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact Daniel Frey, 5550 Granite Parkway, Suite 280, Plano, TX 75024 Telephone: (877) 654-8507.

The terms of your contract, will govern your franchise relationship. Do not rely on the disclosure document alone to understand your contract. Read all of your contract carefully. Show your contract and this disclosure document to a professional advisor, such as a lawyer or accountant.

Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this disclosure document, is available from the **Federal Trade Commission**. You may contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, DC 20580. You may also visit the FTC's website at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may be laws on franchising in your state. Ask your state agencies about them.

The issuance date: April 30, 2024

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, such as current and former franchisees. You can find their names and contact information in Item 20 or Exhibit F.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the Franchisor or at the Franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the Franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit E includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of affiliate-owned and franchised outlets.
Will my business be the only The Glass Guru business in my area?	Item 12 and the "territory" provisions in the Franchise Agreement describe whether the Franchisor and other franchisees can compete with you.
Does the Franchisor have a troubled legal history?	Items 3 and 4 tell you whether the Franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What is it like to be a The Glass Guru Franchisee?	Item 20 or Exhibit F lists current and former Franchisees. You may contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need to Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

<u>Business model can change</u>. The Franchise Agreement may allow the Franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions</u>. You may have to buy or lease items from the Franchisor or a limited group of suppliers the Franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions</u>. The Franchise Agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from Franchisor</u>. Even when the Franchise Agreement grants you a territory, the Franchisor may have the right to compete with you in your territory.

<u>Renewal</u>. Your Franchise Agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends</u>. The Franchise Agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires Franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit B.

Your state also may have laws that require special disclosures or amendments be made to your Franchise Agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About This Franchise

Certain states require that the following risk(s) be highlighted:

- 1. <u>Out-of-State Dispute Resolution</u>. The Franchise Agreement requires you to resolve disputes with the franchisor by mediation, arbitration, and/or litigation only in Texas. Out-of-state mediation, arbitration, or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to mediate, arbitrate, or litigate with the Franchisor in Texas than in your own state.
- 2. <u>Minimum Sales Performance</u>. You must maintain minimum sales performance levels. If you fail to do so, your territory size could be reduced, or the Franchisor could terminate your Agreement and you could lose your investment.
- 3. <u>Financial Condition</u>. The Franchisor's financial condition, as reflected in its financial statements (see Item 21), calls into question the Franchisor's financial ability to provide services and support to you.
- 4. **Spousal Liability.** Your spouse must sign a document that makes your spouse liable for all financial obligations under the franchise agreement even though/if your spouse has no ownership interest in the franchise. This guarantee will place both your and your spouse's marital and personal assets, perhaps including your house, at risk if your franchise fails.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

[BALANCE OF THIS PAGE INTENTIONALLY LEFT BLANK]

FOR THE STATE OF MICHIGAN

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU.

- 1. A prohibition of the right of a Franchisee to join an association of Franchisees.
- 2. A requirement that a Franchisee assent to a release, assignment, novation, waiver or estoppel that deprives a Franchisee of rights and protections provided in this act. This must not preclude a Franchisee, after entering into a Franchise Agreement, from settling any and all claims.
- 3. A provision that permits a Franchisor to terminate a franchise prior to the expiration of this term except for good cause. Good cause must include the failure of the Franchisee to comply with any lawful provision of the Franchise Agreement and to cure such failure after being given written notice and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- 4. A provision that permits a Franchisor to refuse to renew a franchise without fairly compensating the Franchisee by repurchase or other means for the fair market value at the time of expiration of the Franchisee's inventory, supplies, equipment, fixtures and furnishings. Personalized materials that have no value to the Franchisor and inventory, supplies, equipment, fixtures and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (a) the term of the franchise is less than 5 years, and (b) the Franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising or other commercial symbol in the same area subsequent to the expiration of the franchise or the Franchisee does not receive at least 6-months advance notice of Franchisor's intent not to renew the franchise.
- 5. A provision that permits the Franchisor to refuse to renew a franchise on terms generally available to other Franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- 6. A provision requiring that arbitration or litigation be conducted outside this state. This must not preclude the franchisee from entering into an agreement, at the time of arbitration or litigation, to conduct arbitration or litigation at a location outside this state.
- 7. A provision that permits a Franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. The subdivision does not prevent a Franchisor from exercising a right of first refusal to purchase the franchise. Good cause must include, but is

not limited to:

- A. The failure of the proposed transferee to meet the Franchisor's then-current reasonable qualifications or standards.
- B. The fact that the proposed transferee is a competitor of the Franchisor or Subfranchisor.
- C. The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- D. The failure of the Franchisee or proposed transferee to pay any sums owing to the Franchisor or to cure any default in the Franchise Agreement existing at the time of the proposed transfer.
- 8. A provision that requires the Franchisee to resell to the Franchisor items that are not uniquely identified with the Franchisor. This subdivision does not prohibit a provision that grants to a Franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the Franchisor the right to acquire the assets of a franchise for the market or appraised value and has failed to cure the breach in the manner provided in subdivision (C).
- 9. A provision that permits the Franchisor to directly or indirectly convey, assign or otherwise transfer its obligations to fulfill contractual obligations to the Franchisee unless a provision has been made for providing the required contractual services.

If the Franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000.00 the Franchisee may request the Franchisor to arrange for the escrow of initial investment and other funds paid by the Franchisee until the obligations, if any, of the Franchisor to provide real estate, improvements, equipment, inventory, training or other items included in the franchise offering are fulfilled. At the option of the Franchisor, a surety bond may be provided in place of escrow.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENFORCEMENT BY THE ATTORNEY GENERAL.

ANY QUESTIONS REGARDING THIS NOTICE MAY BE DIRECTED TO THE STATE OF MICHIGAN, DEPARTMENT OF ATTORNEY GENERAL, CONSUMER PROTECTION DIVISION,

ATTN: FRANCHISE SECTION, 525 W. OTTAWA ST., G MENNEN WILLIAMS BUILDING, 1st FLOOR, LANSING, MICHIGAN 48913, TELEPHONE (517) 373-7567.

TABLE OF CONTENTS

ITEM 1. TH	IE FRANCHISOR, AND ANY PARENTS, PREDECESSORS AND AFFILIATES	9
ITEM 2. BU	JSINESS EXPERIENCE	10
ITEM 3. LI	TIGATION	11
ITEM 4. BA	ANKRUPTCY	11
ITEM 5. IN	ITIAL FEES	11
ITEM 6. OTH	HER FEES	13
ITEM 7. ES	TIMATED INITIAL INVESTMENT	18
ITEM 8. RE	ESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	23
ITEM 9. FRA	NCHISEE'S OBLIGATIONS	24
ITEM 11. F	RANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND TRAINING	28
ITEM 12. T	ERRITORY	34
ITEM 13. T	RADEMARKS	36
ITEM 14. F	PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION	38
	DBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BU	
ITEM 16. F	RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL	40
ITEM 17. F	RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION	41
ITEM 18. F	PUBLIC FIGURES	46
ITEM 19. F	INANCIAL PERFORMANCE REPRESENTATIONS	47
ITEM 20.	OUTLETS AND FRANCHISEE INFORMATION	55
ITEM 21.	FINANCIAL STATEMENTS	59
ITEM 22.	CONTRACTS	59
ITEM 23.	RECEIPTS	60

EXHIBITS

A. FRANCHISE AGREEMENT

- 1. SPECIFICS
- 2. GENERAL RELEASE
- 3. NONDISCLOSURE AND NON-COMPETITION AGREEMENT
- 4. GUARANTY AND ASSUMPTION OF OBLIGATIONS
- 5. HOLDERS OF LEGAL OR BENEFICIAL INTEREST IN FRANCHISEE; OFFICERS; DIRECTORS
- 6. ASSIGNMENT OF TELEPHONE NUMBERS AND INTERNETTOOLS
- 7. ADDENDUM TO LEASE AND COLLATERAL ASSIGNMENT OF LEASE
- 8. MULTI-STATE ADDENDA
- B. LIST OF STATE ADMINISTRATORS
- C. LIST OF STATE AGENTS FOR SERVICE OF PROCESS
- D. TABLE OF CONTENTS OF CONFIDENTIAL OPERATIONS MANUAL
- E. FINANCIAL STATEMENTS
- F. LISTS OF CURRENT AND FORMER FRANCHISEES
- G. MULTI-STATE ADDENDA
- H. FINANCING DOCUMENTS

ITEM 1. THE FRANCHISOR, AND ANY PARENTS, PREDECESSORS AND AFFILIATES

To simplify the language in this Disclosure Document, the words "we," "our" and "us" refer to The Glass Guru Enterprises, Inc., a Texas corporation, the Franchisor of this business. "You" and "your" refer to the entity that buys the franchise, whether it be a corporation, limited liability company or other business entity. Certain provisions of this Disclosure Document also apply to your owners and will be noted.

The Franchisor

Our entity name and state of incorporation was changed from California to Texas effective April 9, 2021, by merger of our predecessor, The Glass Guru Franchise Systems, Inc., a California corporation into us with us surviving the merger. Our predecessor was formed on February 14, 2006, and we were formed on January 1, 2021. As the survivor of the merger, we are a successor in interest to our predecessor, and as such, will continue the operation of the business of offering the sale of and supporting The Glass Guru franchises. Our principal business address is 5550 Granite Parkway, Suite 280, Plano, Texas 75024. Our predecessor's address was 1382 Blue Oaks Blvd., Suite 213, Roseville, California 95678. We do business under our corporate entity name, The Glass Guru Enterprises, Inc. as well as the name "The Glass Guru."

Our predecessor offered The Glass Guru franchises from January 2007 up to the date of the merger and we began offering The Glass Guru franchises following the merger. Neither we nor our predecessor have conducted the type of business that you will operate; however, our affiliate has. Neither we nor our predecessor or our affiliate have offered franchises in any other line of business.

The agent for service of process is CT Corporation, 1999 Bryan Street, Suite 900, Dallas, Texas 75201. Please refer to **Exhibit C** for a list of other applicable agents for service of process.

Our Business Activities

We grant franchises to qualified persons or business entities in connection with the service mark "THE GLASS GURU" and other related trademarks and logos (collectively referred to as the "Marks"). We refer to these businesses as "The Glass Guru Business." We refer to The Glass Guru Business you will operate as the "Franchise Business."

The Glass Guru Business installs, repairs and replaces residential and commercial glass, mirrors, showers, windows and doors, and provides other related services and sells other related products to residential and commercial customers. You must operate the Franchise Business in accordance with our standards, methods, procedures, and specifications, which we refer to as our "System" and which is more particularly described in our Franchise Agreement attached as *Exhibit A* to this Disclosure Document. We (and our Affiliate) have custom designed, developed and/or acquired lines of products, such as certain window restoration tools and equipment, and other tools and equipment items especially suited for use in connection with the Franchise Business ("The Glass Guru Restoration Tools"). We are the supplier of The Glass Guru Restoration Tools to our franchisees.

Parents and Affiliates

We do not have a parent. We have no affiliates that provide products or services to our franchisees or that offer franchises in any line of business. We have one affiliate, Roseville Glass Partners, Inc., DBA The Glass Guru of Roseville, located at 198 Cirby Way, STE. 120, Roseville, California 95678, that has owned and operated one The Glass Guru Franchise Business in California since 2004.

General Description of the Market and Competition

Our concept is targeted to residential and commercial customers. As a Franchisee, you will compete with a variety of other businesses that install, repair, and replace residential and commercial glass, mirrors, showers, windows and

doors, and provide other related services and sell other related products to residential and commercial customers and sales are not seasonal. Your competition may be local independently owned businesses or may be part of a regional or national dealership chain or franchise system including companies such as Glass Doctor, Binswanger Glass, Window World or Window Medics. You may also compete with other The Glass Guru Businesses operated by us or other Franchisees.

Regulations Specific to the Industry

You must comply with all laws, rules, and regulations governing the operation of the Franchise Business and obtain all permits and licenses necessary to operate the Franchise Business. California, Nevada, Arizona, Florida, and other states require licensing to provide glass and window replacement or similar services. You should contact your state to determine if you must obtain a license. Some states, as well as local jurisdictions have enacted laws, rules, regulations, and ordinances which may apply to the operation of your Franchise Business, including those which: (a) regulate matters affecting the health, safety, and welfare of your customers, such as general health and sanitation requirements, restrictions on smoking, availability of and requirements for public accommodations, including restroom facilities and public access; (b) set standards pertaining to employee health and safety; (c) regulate matters affecting requirements for accommodations for disabled persons; (d) set standards and requirements for fire safety and general emergency preparedness; and (e) regulate the proper use, storage and disposal of waste or other hazardous materials. You should investigate the federal, state, and local laws, rules and regulations that apply in the geographic area in which you are interested in locating your Franchise Business and should consider both their effect and cost of ongoing compliance. Under the Franchise Agreement, you alone are responsible for knowing and complying with all applicable laws and regulations despite any advice or information that we may give you.

ITEM 2. BUSINESS EXPERIENCE

President | Daniel Frey

Mr. Frey is the Founder and has served as our and our predecessor's President/CEO since 2007. In addition, Mr. Frey is also the Founder, Director, and President/CEO of our Affiliate, Roseville Glass Partners, Inc., DBA The Glass Guru of Roseville since September 2004. Roseville Glass Partners, Inc. is located in Roseville, California.

VP Corporate Administration | D. Joy Frey

Mrs. Frey is the Co-Founder and has served as the VP of Corporate Administration since 2020. Additionally, she has held the position of Co-Founder and senior administrative professional of our Affiliate, Roseville Glass Partners, Inc., DBA The Glass Guru of Roseville since 2004. Roseville Glass Partners, Inc. is located in Roseville, California.

VP Franchise Operations | Dave Hull

Mr. Hull joined us as VP of Franchise Development and was appointed a Director in January 2011. Mr. Hull was elected as our Chief Development Officer in 2018. In May 2020, Mr. Hull's title was updated to VP of Franchise Operations. Mr. Hull is also the Co-Owner and managing owner of our Affiliate, Roseville Glass Partners, Inc., DBA The Glass Guru of Roseville from October 2009 – March 2011, and from September 2018 to present. Roseville Glass Partners, Inc. is located in Roseville, California.

VP Franchise Business Development | Rob Lopez

Mr. Lopez has served as our VP of Franchise Business Development since April 2020 and previously served as Director of Franchise Development from November 2015 to April 2020. Prior to joining us, Mr. Lopez appointed

as the President of Servicer Master-CCS, located in Charlotte and Lee County, Florida, an independently owned and operated franchise location business from 2007 to 2015.

VP Marketing & Communications | Jason Mitchell

Mr. Mitchell has served as our VP of Marketing & Communications since March 2023. Mr. Mitchell has served as the VP of Marketing for Spray-Net, headquartered in Boucherville, Canada, from August 2022 to March 2023. Prior to his tenure at Spray-Net, Mr. Mitchell served as the National VP of Mobile Modular for EZ-Access, headquartered in Auburn, Washington, from December 2020 to February 2022. Mr. Mitchell also served EZ-Access as the Western Regional VP of Sales from September 2020 to December 2020. Preceding his experience with EZ-Access, Mr. Mitchell contributed to Next Day Access, headquartered in Memphis, Tennessee, by serving as the Regional Director of Sales from March 2018 to September 2020.

Corporate Paralegal & Compliance Director | Annette M. Lopez

Ms. Lopez has served as our Corporate Paralegal and Compliance Director since July 2021. Prior to joining us, Ms. Lopez served as a Senior Civil Defense Paralegal for the United States Attorney's Office, Eastern District of California, from 2016 to 2021.

Director of Technology | Sean Young

Mr. Young has served as our Director of Technology since May 2020. Mr. Young previously served as the Director of Operations for our predecessor from May 2018 to May 2020.

<u>Director of Franchise Sales | S. Colby Howell</u>

Mr. Howell has served as our Director of Franchise Sales since February 2023. Mr. Howell serves as the Founder and member of Break Free Power, LLC, headquartered in Austin, Texas, since August 2021. Preceding that role, Mr. Howell served as a Lead Broker for Eastwood Energy Group, headquartered in Dallas, Texas, from January 2018 to August 2021.

ITEM 3. LITIGATION

No litigation information is required to be disclosed in this Item.

ITEM 4. BANKRUPTCY

No bankruptcy information is required to be disclosed in this Item.

ITEM 5. INITIAL FEES

You must pay us a non-refundable Initial Franchise Fee of \$30,000 to \$50,000, based on the population of the chosen territory a.k.a. Area of Primary Responsibility (APR) by reference to the most recent figures available from the United States Census Bureau at the time the territory/APR is designated, less applicable discounts when you sign the Franchise Agreement. The Franchise Fee may be paid either in a lump sum or if you meet our credit standards, by payment of a down payment plus the signing of a Promissory Note. We may finance up to 50% of the non-refundable Initial Franchise Fee. The Franchise Fee is payment, in part, for expenses incurred by us in furnishing assistance and

services to you as described in the Franchise Agreement and for costs incurred by us, including general sales and marketing expenses, training, legal, accounting, and other professional fees.

How the Initial Franchise Fee is Calculated

The non-refundable Initial Franchise Fee is determined by the aggregated population of the zip/postal codes included in the Area of Primary Responsibility (the Franchise Territory). This fee is calculated utilizing a sliding scale. The total non-refundable Initial Franchise Fee increases by approximately \$31 for every 1,000 persons increase in population. There is a minimum franchise fee of \$30,000 for any Franchise Territory equal to or less than 150,000 in population. The non-refundable Initial Franchise Fee is capped at \$50,000 for a Franchise Territory of 800,000 or greater in population. The table below provides examples of the non-refundable Initial Franchise Fee calculation for various sample territory populations:

POPULATION	\$ PER 1000	INITIAL FRANCHISE FEE
150,000	\$200.00	\$30,000
200,000	\$157.69	\$31,538
300,000	\$115.38	\$34,615
400,000	\$94.23	\$37,692
500,000	\$81.54	\$40,769
600,000	\$73.08	\$43,846
700,000	\$67.03	\$46,923
800,000	\$62.50	\$50,000

Minimum Franchise Fee = \$30,000 / Maximum Franchise Fee = \$50,000

Veteran Discount

If you are an honorably discharged veteran who meets our qualifications for purchasing a franchise, we will discount the non-refundable Initial Franchise Fee by an amount equal to 10% of the total non-refundable Initial Franchise Fee. This Veteran Discount is available only to United States and Canadian veterans honorably discharged and the term "veteran" shall be defined by us in our sole discretion. However, in determining who is included in the term "veteran" we may be guided, in whole or in part, by any definitions we deem appropriate, including definitions used by the federal government of the United States or Canada, as applicable, in determining who is eligible for federal benefits intended for veterans.

Incentives for Converted Businesses

Franchisor, at its sole discretion, may choose to award a franchise territory to a Franchisee who concurrently owns a business that offers/sells products and services that directly compete with The Glass Guru ("Competing Business") as defined in the Agreement. In these circumstances, Franchisee will be required to convert Competing Business with and into The Glass Guru Franchise Business. All subsequent business will be conducted by the Franchisee under the mark(s) of The Glass Guru, utilizing appropriate copyrighted materials, manuals, systems, etc., subject to the terms and conditions of the Agreement.

While Franchisor will discuss a business conversion with any Competing Business, a Competing Business is only eligible to receive "Incentives for Converted Businesses" when the Competing Business converting to a Franchise

Business can demonstrate Gross Sales equal to, or greater than \$500,000 for the most current 12-month operating period. In the case where Franchisee converts a Competing Business into a Franchise Business, the Converted Business discount schedule for the Franchise Fee and Royalty Fee will apply as follows:

Option 1

- A. Franchisee receives a 50% reduction in the non-refundable Initial Franchise Fee (as defined above); and,
- B. For a period of 12 months, Franchisee will receive a 50% reduction in the monthly Royalty Fee for the mutually agreed upon Gross Sales from the most current 12-month period prior to converting. Any Gross Sales exceeding this amount derived during the first 12-months of operations will be subject to the full monthly Royalty Fee amount defined above. This reduction in the monthly Royalty Fee will terminate following the Franchise Business's 1st full year (12-months) of operation.

Option 2

- A. Franchisee receives NO reduction in the non-refundable Initial Franchise Fee; and,
- B. For a period of 24 months, Franchisee will receive a 50% reduction in the monthly Royalty Fee (as defined above) for the mutually agreed upon Gross Sales from the most current 12-month period prior to converting. Any Gross Sales exceeding this amount derived during this 12-month period will be subject to the full Monthly Royalty Fee amount as defined in section 3.2 of the Agreement. This reduction in monthly Royalty Fees will reset and renew for a subsequent 12-month period during the Franchise Business's 2nd full year of operation. This reduction in the monthly Royalty Fee will terminate following the Franchise Business's 2nd full year (24-months) of operation.

Initial Purchases

Before you open the Franchise Business, you must purchase an initial supply of The Glass Guru Restoration Tools from us. The current cost of the Tool Kit is \$1,000 plus applicable shipping and sales tax. If the Franchise Agreement is terminated before you begin operations of the Franchise Business, you may return any tools or supplies purchased from us for a refund, provided that the items are unopened, unused, or remain in a saleable condition.

Before you open the Franchise Business, you must purchase The Glass Guru Brand Identity Package from us for a non-refundable fee currently assessed at\$5000 plus applicable shipping and sales tax. The items in the package promote the services offered by the Glass Guru Business.

ITEM 6. OTHER FEES

Name of Fee	<u>Amount</u>	<u>Due Date</u>	<u>Remarks</u>
Royalty Fee	5% of Gross Sales for all Products & Services.	Monthly by the 20th day of the calendar month	See Note 1
Non-APR Fee	2% of Gross Sales for all Products & Services provided outside of the Approved Area of Primary Responsibility.	20th day of the	See Note 1

Name of Fee	<u>Amount</u>	<u>Due Date</u>	<u>Remarks</u>
Marketing Fee	Franchisor charges a Marketing Fee. Franchisee is required to pay the Marketing Fee on a monthly basis equal to three percent (3%) of the corresponding monthly Gross Sales.	Monthly by the 20th day of the calendar month	See Definitions for Marketing Fee, and further clarifications. See Note 1
Local Advertising	Franchisee must spend the greater of five percent (5%) of the prior year's Gross Sales divided by the months of operation in the prior year, or the minimum monthly local advertising expenditure of \$1,250 ("Minimum Local Advertising Expenditure") on advertising, promotions, and public relations within the agreed upon Area of Primary Responsibility. A Franchise Business failing to meet the "Minimum Local Advertising Expenditure" will be assessed a fee ("Local Advertising Penalty") in the amount of the required spend minus the dollar amount spent aka the shortfall amount.	Monthly	You pay to third parties directly subject to our approval. Failing to meet the "Minimum Local Advertising Expenditure" will result in an additional fee. Your expenditures for Local Advertising are in addition to the required Marketing Fee listed above as well as Contributions and required Grand Opening Advertising expenditures (described in ITEM 7). Further information about all of our advertising programs is also included in ITEM 11 of this Disclosure Document. See Note 1 *The Local Advertising Penalty assessed will be the amount of the deficit of the amount required versus the amount actually spent.
Technology Fee	Rates as published in the Manual; currently, Franchisor charges a fee equaling the greater of \$360 or 1% of the corresponding monthly Gross Sales, with a maximum fee of \$760 being assessed.	Monthly	You pay us a technology fee that covers the direct costs of approved software, software licenses, subscriptions, development, maintenance, and hosting for all systems substantially benefiting franchise locations including but not limited to feature sets like; field service management (FSM), business productivity software, Intranet, learning management (LMS) system, and business intelligence systems (BI), and any additional systems/software otherwise authorized and/or required in the Operations Manual that are procured and managed by us on your behalf. See Note 1
Trade Associations	Rates as published in the Manual; currently, \$100. In addition to the direct fee, an additional charge of up to \$20 may be applied to this fee annually to compensate Franchisor for the administrative costs incurred of overseeing the collective memberships which provides an otherwise unavailable discount to the Franchise Business.	Yearly	We collect from you and pay to the National Glass Association on your behalf.

Name of Fee	<u>Amount</u>	<u>Due Date</u>	<u>Remarks</u>
Late Entry Fee	Rates as published in the Manual; currently, \$175	Upon demand	Failure to enter customer data within 48 hrs. of initial customer contact during the month will result in a Late Entry Fee. Non-compliance for three or more months during a calendar year will require an audit at Franchisees expense.
Audit Expenses	All costs and expenses associated with audit.	Upon demand	Audit costs payable only if the audit shows an understatement in amounts due of at least 3% or if audit is conducted due to your late entries or failure to report.
Late Fees	1.5% per month or the highest rate allowed by the law of the state where you are located, whichever is less.	Upon demand	Applies to all overdue Royalty Fees, Marketing Fees, Technology Fees, and any and all other amounts due to us. Also applies to any understatement in amounts due revealed by an audit.)
Insurance Reimbursement	Amount of unpaid premiums plus our actual expenses in obtaining the policies.	Upon demand	Payable only if you fail to maintain required insurance coverage and we elect to obtain coverage for you.
Transfer Fee	Rates as published in the Manual; currently, \$7,500.	Time of transfer	Does not apply to an assignment under <i>Section 18.3</i> of the Franchise Agreement.
Customer Service	All costs incurred in assisting your customers.	Upon demand	You must reimburse us if we determine it is necessary for us to provide service directly to your customers.
Substitute or New Manager (a.k.a. Designated Manager) Training	Rates as published in the Manual; currently, \$800 per day at Qualified Training Location (QTL) and \$500 per day at Franchisee's location, plus your expenses as well as your employees' expenses in attending.	As required	If your Designated Manager does not satisfactorily complete our Initial Training Program (or if you replace your Designated Manager), a substitute (or your replacement manager) must complete our Initial Training Program. We may charge you for additional training. Additionally, the Franchisee (Owner) and Designated Manger for all transferees will incur this expense. (See "Additional Training" in this table)

Name of Fee	<u>Amount</u>	<u>Due Date</u>	<u>Remarks</u>
Temporary Management Assistance	Rates as published in the Manual; currently, \$500 per day, plus expenses, or 25% of the Gross Sales during that period, whichever is greater.	Each month that it applies	Following the delivery of a Notice of Termination of the Franchise Agreement, we may assume operation of the Franchise Business until such time as you correct the breach resulting in the Notice of Termination. Additionally, following the death or incapacity of an owner of the Franchise Business, if necessary, and in our discretion, we may assume operation of the Franchise Business until the deceased or incapacitated owner's interest is transferred to a third party approved by us. We may charge a management fee during the time we are operating your Franchise Business and we will also be entitled to reimbursement of any expenses we incur that are not paid out of the operating cash flow of the Franchise Business. Washington Franchisees Only: The Franchisor will only charge the Temporary Management Fee to the extent it represents the Franchisor's actual and reasonable expenses in managing the franchise and will not charge the Fee for more than 180 days.
Ongoing Training Programs	Fees as determined by us.	Time of ongoing training program	You must pay your expenses as well as your employees' expenses in attending ongoing training programs. Attendance will not be required more than 2 times per year and collectively will not exceed 7 days in any year, including attendance at an Annual Convention.
Additional Training	Rates as published in the Manual; currently, \$800 at Qualified Training Locations (QTL) or \$500 at Franchisee's location, per day per person, plus your expenses as well as your employees' expenses in attending.	Time of training	We provide training for you and one office assistant. Up to 2 additional field or office assistants will be provided training at your request and with our prior authorization. You must pay for additional training if you request it.
Additional Operations Assistance	Rates as published in the Manual; currently, \$800 at Qualified Training Locations (QTL) or \$500 at Franchisee's location, per day, per staff member, plus our expenses.	Time of assistance	We provide assistance around the beginning of operations. You may pay for additional assistance if you request it.

Name of Fee	<u>Amount</u>	<u>Due Date</u>	<u>Remarks</u>
Appearance and Condition of the Franchise Business	All actual costs incurred by the Franchisor to bring the Franchise Business, vehicles, equipment, Location, and Signage in an attractive and safe condition and in good maintenance and repair in compliance with the standards set by the Franchisor in the Manual or required by any external agency, laws, or regulations.	Upon Demand	Upon Franchisee's failure to correct Notice of Default of specific standard(s). This fee is not the whole remedy available to franchisor under such circumstances.
New Technician Training	Rates as published in the Manual; currently, \$800 at Qualified Training Locations (QTL) or \$500 per day at Franchisee's location, per day per person. In addition, all expenses incurred by your technicians in attending these programs, including travel costs, room and board expenses and employees' salaries will be your sole responsibility.	Time of program	Optional for new trainees. Does not apply to any technicians attending the initial training we provide to you and your assistants.
National Convention	Rates as published in the Manual; currently, \$750. If you do not attend Convention, the fee is paid in any case and any training materials will be made available to you at the time and manner we determine.	Before date of convention	Payable if we hold a Convention for franchisees. You must pay the fee even if you fail to attend the Annual Convention. You must also pay your expenses as well as your employees' expenses in attending.
Cost of Enforcement	All costs including attorneys' fees.	Upon demand	You must reimburse us for all costs in enforcing obligations if we prevail.
Indemnification	All costs including attorneys' fees.	Upon demand	You must defend lawsuits at your cost and hold us harmless against lawsuits arising from your operation of the Franchise Business.
Administrative Fee for Improper Reporting	Rates as published in the Manual; currently, \$75 each occurrence.	Upon demand	Payable if you submit any Report in an improper or unapproved format.

Name of Fee	<u>Amount</u>	<u>Due Date</u>	<u>Remarks</u>
Administrative Fee for Failure to Submit Timely Reporting	Rates as published in the Manual; currently, \$50 each calendar day	Upon demand	Payable if you fail to transmit reports, records, or other information to Franchisor; or fail to timely close jobs and process open jobs; or fail to provide accurate information for each day data not transmitted properly.
NSF Fees and Interest	Rates as published in the Manual; currently, \$75 each occurrence + Interest	Upon demand	Payable as a result of ACH or other transfers returned to Franchisor for non-sufficient funds or any other reasons. Also, Franchisee must pay to Franchisor, on demand, interest on all overdue payments from the date the payment was due until paid equal to the greater of (i) 1.5% per month or (ii) the maximum rate of interest permitted by law.

NOTES

¹"Gross Sales" means the total gross revenues collected or received from whatever source (whether in the form of cash, credit, agreements to pay, barter, trade credits, good will or other consideration) that arise, directly or indirectly, from or in connection with the operation of your franchise, including but not limited to:

- a. the sale of goods and/or services offered by or through the franchise;
- b. the sale of goods and/or services by you or a third-party selling products and/or services on your behalf that are sold or that are required to be sold under the terms of the Franchise Agreement, no matter from what location or business the income is generated;
- c. the proceeds from any business interruption insurance and/or damages or settlement amounts received to compensate you for lost revenue of the franchise;
- d. any revenue generated from commissions, rebates or affiliated programs; and,
- e. except as provided below for charity activities, the value of any goods and/or services provided without compensation to you.

ITEM 7. ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT

Names of Expenditures	Actual or Amounts	Estimated s for You	Method of Payment	When Due	To Whom
Franchise Fee ¹	\$30,000	- \$50,000	Wire / Cashier's Check / Installments	Upon Signing Franchise Agreement	Us
Real Estate/Rent ²	\$3,500	- \$8,000	As Arranged	Before Beginning Operations	Lessor
Utility Deposits ³	\$0	- \$500	As Arranged	Before Beginning Operations	Utilities
Leasehold Improvements ⁴	\$500	- \$2,000	As Arranged	Before Beginning Operations	Third Parties
Insurance ⁵	\$500	- \$3,000	As Arranged	Before Beginning Operations	Third Parties
Office/Shop Equipment and Supplies ⁶	\$5,000	- \$8,500	As Arranged	Before Beginning Operations	Third Parties
Initial Glass Guru Tool Kit ⁷	\$1,000	- \$1,000	Wire / Cashier's Check	Upon Signing Franchise Agreement	Us
Training Expenses ⁸	\$1,500	- \$2,500	As Arranged	Before Beginning Operations	Third Parties
Signage ⁹	\$2,500	- \$4,000	As Arranged	Before Beginning Operations	Third Parties
Furniture, Fixtures & Equipment ¹⁰	\$500	- \$3,500	As Arranged	Before Beginning Operations	Third Parties
Computer Equipment ¹¹	\$1,250	- \$4,800	As Arranged	Before Beginning Operations	Third Parties
Software Systems ¹²	\$1,305	- \$2,730	As Arranged	First 3 months of Operation	Third Parties
Phone System ¹³	\$500	- \$2,500	As Arranged	Before Beginning Operations	Third Parties
Vehicles & Modifications ¹⁴	\$15,000	- \$25,000	As Arranged	Before Beginning Operations	Third Parties
Initial Local Advertising ¹⁵	\$3,750	- \$7,500	As Arranged	First 3 Months of Operation	Third Parties
Brand Identity Package ¹⁶	\$5,000	- \$5,000	Wire / Cashier's Check	Upon Signing Franchise Agreement	Us
Licenses & Permits ¹⁷	\$2,000	- \$4,000	As Arranged	Before Beginning Operations	Licensing Authority
Legal & Accounting ¹⁸	\$2,500	- \$5,000	As Arranged	First 3 Months of Operation, and As Necessary	Attorney, Accountant
Additional Funds/ (3 months) Working Capital ¹⁹	\$50,000	- \$80,000	As Arranged	As Necessary	You Determine
TOTAL ²⁰	\$126,305	- \$219,530			

NOTES

We anticipate that you will incur the initial expenditures estimated above in the establishment of a Franchise Business. Financing for the initial expenditures paid to us is offered by Franchisor under the terms described in Item 10.

- 1. <u>Franchise Fee.</u> The initial Franchise Fee is described in greater detail in ITEM 5 of this Disclosure Document. The terms for payment of a portion of the Franchise Fee in installments are described in greater detail in ITEM 10 of this Disclosure Document.
- 2. Real Estate/Rent. You must lease or otherwise provide a suitable facility for the operation of the Franchise Business. You must lease a space, your lease costs can vary based upon variance in square footage, cost per square foot and required maintenance costs. A suitable facility will have approximately 1,000 2,000 sf of combined office and warehouse space, with a minimum of 300 sf dedicated for office/showroom use, and 700 sf dedicated for warehouse use with a rolltop door available for deliveries. The facility must be located within the defined geographical boundaries of the licensed "Area of Primary Responsibility", and in an area, zone or district approved for retail and warehouse use. We assume that you will have to pay the first month's rent and a security deposit equal to one month's rent in advance. The rent you pay is typically not refundable, but your security deposit may be under certain circumstances.
- 3. <u>Utility Deposits.</u> If you are a new customer of your local utilities, you will generally have to pay deposits in connection with services such as electric, telephone, gas and water. The amount of the deposit and whether the deposit is refundable will vary depending upon the policies of the local utilities. You should contact your local utilities for more information.
- 4. <u>Leasehold Improvements.</u> If you lease space, you may need or choose to make minimal renovations. The cost of leasehold improvements will vary based upon size, condition, and location of the premises, local wage rates and material costs. The amounts you pay for leasehold improvements are typically not refundable.
- 5. <u>Insurance</u>. You must purchase the type and amount of insurance specified in Section 15 of the Franchise Agreement in addition to any other insurance that may be required by applicable law, any lender or lessor. Factors that may affect your cost of insurance include the location of the Franchise Business, value of the leasehold improvements, number of employees and other factors. The amounts you pay for insurance are typically not refundable. See ITEM 8 for more information.
- 6. Office/Shop Equipment and Supplies. You must purchase general office supplies including printers, stationery, business cards, and other typical office equipment. We do not know if the amounts you pay for office equipment and supplies are refundable. You must also purchase an initial supply of shop tools and consumables necessary for your Franchise Business to provide basic glass replacement services. You should inquire about the return and refund policy of the supplier at or before the time of purchase. Factors that may affect your cost of office equipment and supplies include local market conditions, competition among suppliers, and other factors.
- 7. <u>Initial Glass Guru Tool Kit.</u> You will be required to purchase the initial Glass Guru tool kit from us for the rate outlined in the Manual; currently, \$1000 (plus applicable shipping and sales taxes) prior to beginning operations. This "Kit" provides tools necessary for your Franchise Business to provide our proprietary Foggy Window Repair Service, including a small supply of micro-vents. This "Kit" is available through our online store.

- 8. <u>Training Expenses.</u> The cost of initial training is included in the Franchise Fee, but you are responsible for transportation and expenses for meals and lodging while attending training. The initial training program is also available to the Designated Manager and one office assistant. Up to two additional field or office assistants will be allowed to attend the initial training program at your request and with our prior consent for an additional fee as listed in ITEM 6. The total cost will vary depending on the number of people attending, how far you travel and the type of accommodations you choose. Your expenses to attend are typically not refundable.
- 9. <u>Signage</u>. This range includes the cost of all signage used in the Franchise Business. The signage requirements and costs will vary based upon the location of the Franchise Business, local zoning requirements and local wage rates for installation, among other things. The amounts you pay for signage are typically not refundable.
- 10. <u>Furniture, Fixtures & Equipment.</u> You will be required to purchase and/or lease and install furniture, fixtures and equipment such as desks, chairs, lamps, filing cabinets and other basic office items necessary to operate your Franchise Business. The cost of the furniture, fixtures and equipment will vary according to local market conditions, suppliers, and other related factors. We do not know if the amounts you pay for furniture, fixtures, or equipment may be refundable. You should inquire about the return and refund policy of the suppliers at or before the time of purchasing or leasing.
- 11. <u>Computer Equipment.</u> You will be required to have one computer workstation with Microsoft Office Suite and QuickBooks and other software as outlined in the Operations Manual. These will need to be purchased if you don't have them already. The amounts you pay for computer equipment are typically not refundable.
- 12. <u>Software Systems.</u> You will be required to pay us a technology fee that covers the direct costs of approved software, software licenses, subscriptions, development, maintenance, and hosting for all systems substantially benefiting franchise locations including but not limited to feature sets like; field service management (FSM), business productivity software, Intranet, learning management (LMS) system, and business intelligence systems (BI), and any additional systems/software otherwise authorized and/or required in the Operations Manual. The fee is as outlined in the Manual; currently, the monthly fee equals the greater of \$360 or 1% of Gross Sales, with a maximum fee of \$760 being assessed. You are also required to use Intuit QuickBooks Online for your accounting software. The QuickBooks license is not included as part of the technology fee paid to us, and you must acquire and license the software directly with Intuit or an approved reseller at a cost to you of \$75 to \$150 per month. The cost of QuickBooks is not considered in the charts above.
- 13. <u>Phone System.</u> You will be required to purchase a cell phone and a dedicated phone line to a commercial office. The number must be one with an area code and prefix that is local to the Area of Primary Responsibility. The amounts you pay for a phone system are typically not refundable.
- 14. <u>Vehicles.</u> You may be required to lease or purchase a utility vehicle, van or truck for use in the operation of the Franchise Business. The high and low estimates represent 3 months of estimated payments assuming full financing of the purchase price of the vehicles. Based on such factors as the location of the Franchise Business, competition among local dealers, financing options available to you and the time of year, these estimates may vary. These expenses are typically not refundable. Each vehicle is also required to be wrapped, and each vehicle transporting glass is required to have a glass rack for use in the operation of the

Franchise Business. The high and low estimates represent vehicle wrap costs and glass rack costs, which may vary based on such factors as the location of the Franchise Business, competition among local dealers and the time of year. These expenses are typically not refundable.

- 15. <u>Initial Local Advertising.</u> You will be required to spend the greater of five percent (5%) of the prior year's Gross Sales divided by the number of months in operation in the prior year, or minimum monthly local advertising expenditure of \$1250 ("Minimum Local Advertising Expenditure") on advertising, promotions, and public relations within the agreed upon Area of Primary Responsibility surrounding the Franchise Business ("Local Advertising"). The low estimate of \$3750 covers the minimum local advertising during the first 3 months of operation and the higher estimate assumes that you may choose to spend additional amounts to promote the opening of the business.
- 16. <u>Brand Identity Package.</u> You must purchase The Glass Guru Brand Identity Package from us at the rate outlined in the Manual; currently, \$5,000 (plus applicable shipping and sales taxes) prior to beginning operations of the Franchise Business. These items include store posters, displays, banners, trade show items, rack and other cards bearing the Marks and promote the services offered by the Glass Guru Business. You may choose to spend more. Factors that may affect the actual amount you spend include the, the size of the area you advertise to, location of the Franchise Business, time of year and customer demographics in the surrounding area. The amounts you spend for the Brand Identity Package are not refundable.
- 17. <u>Licenses & Permits.</u> Local government agencies typically charge fees for such things as contractors' licenses, business, and other operating licenses. Your actual costs may vary from the estimates based on the requirements of local government agencies. These fees are typically not refundable.
- 18. <u>Legal & Accounting.</u> We strongly suggest that you employ an attorney, an accountant and/or other consultants to assist you in establishing your Franchise Business. These fees may vary from location to location depending upon the prevailing rates of local attorneys, accountants, and consultants. These fees are typically not refundable. In conjunction with entering into the Franchise Agreement, you are required to obtain an independent third-party Bookkeeping service approved by us to facilitate your bookkeeping as outlined in the Manual; currently for the first 3 full months of operation. Our Approved and/or Preferred Vendors typically charge \$1500.00 for this service.
- 19. Additional Funds/Working Capital. We recommend that you have a minimum amount of money available to cover operating expenses, including all variable and fixed expenses without revenue for the first 3 months that the Franchise Business is open. The amounts referenced above have been determined through assessment of the historical break-even analysis of new franchise locations opened within the last year. However, we cannot guarantee that our recommendation will be sufficient. Additional working capital may be required if sales are low or operating costs are high. These expenses are typically not refundable.
- 20. <u>Total</u>. You should review these estimates carefully with an accountant or other business advisor before making any decision to buy a franchise.

The initial investment described in this Item relates to the development of a new franchise. If you are renewing your existing franchise for the next Term, you may not be required to pay a renewal fee, and you will not incur most of the expenses referenced in this Item. However, you may be responsible for upgrading your franchise and any related expenses. If you are acquiring an existing franchise by transfer, in addition to the price you negotiate for

the purchase of the franchise with the seller, you may be responsible for the transfer fee and you also may be responsible for upgrading the franchise and any related expenses. If you are a conversion franchise, you may not incur many of the expenses referenced in this item. However, you may be responsible for upgrading your business to become compliant with the terms of the Franchise Agreement.

ITEM 8. RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

Except as indicated below, you are not required to purchase or lease products or services from us or from suppliers approved by us or under our specifications. You may work with alternative suppliers that meet our criteria without additional approval, except as regards the items outlined below.

Approved Suppliers/Standards and Specifications

All products, supplies, signs, equipment, components, accounting services, insurance and other products or materials for use or sale in your Franchise Business must meet our specifications and quality standards and, if required by us, must be purchased only from an Approved Supplier, which may be us. Currently, we are the only Approved Supplier for The Glass Guru Restoration Tools. Certain components of the items in The Glass Guru Restoration Tools are proprietary equipment and supplies that you can only purchase from us. Currently proprietary equipment and supplies include micro vents & tip adapters for Foggy Window Repair.

We are also currently the only Approved Supplier for the Brand Identity Package with marketing and promotional materials bearing the Marks.

We will provide, in the Manual or by other written or electronic form, a list of The Glass Guru Restoration Tools as well as additional items you will need to purchase for resale or to operate your Franchise Business and, if required, a list of Approved Suppliers for some or all of these items, and from time to time we may revise this list. Our specifications may include minimum standards for performance, design, appearance, and quality. We formulate and modify our specifications and standards for products and services based upon our, and our Affiliate's industry knowledge and our operational experience in operating The Glass Guru Business since 2004. Neither the Franchisor, nor any officer of the Franchisor, owns any interest in any Approved Supplier (except for us).

If you would like to use any item, service, or supplier in establishing or operating the Franchise Business that we have not approved (for items or services that require supplier approval), you must first send us sufficient information, specifications and/or samples for us to determine whether the item, service or supplier complies with our standards and specifications or the supplier meets our Approved Supplier criteria. We will decide within a reasonable time (usually, 30 days) after receiving the required information whether you may purchase or lease such items or services or from such supplier. We apply the following and other general criteria in approving a proposed supplier: the ability to provide sufficient quantity of product; quality of products and/or services at competitive prices; production and delivery capability; and dependability and general reputation of the supplier. Presently, there are no costs or expenses associated with the evaluation of products or suppliers proposed by franchisees that must be reimbursed.

Periodically, we may review our approval of any item, service, or supplier. We will notify you if we revoke our approval of an item, service, or supplier, and you must immediately stop purchasing disapproved items or services or must immediately stop purchasing from a disapproved supplier.

The Glass Guru Restoration Tools

We, and our Affiliate, have designed, developed and/or acquired and continue to design and develop certain window restoration tools and equipment and other items especially suited for use in the operation of The Glass Guru Business ("The Glass Guru Restoration Tools"). In order to maintain the consistency, quality, and uniformity of the System, we will make The Glass Guru Restoration Tools available to you in accordance with the procedures for ordering, handling, and shipping that we may determine periodically, provided that you are in compliance with

the Franchise Agreement and all other agreements with us. We will provide The Glass Guru Restoration Tools supplied by us, on an as needed basis, at competitive prices; however, we have the right to earn a reasonable profit on the sale of The Glass Guru Restoration Tools. If properly maintained, the restoration kit is expected to last at least 3 years. We derive revenue from the sale of The Glass Guru Restoration Tools to our franchisees.

During the fiscal year ended December 31, 2023, we had revenue of \$38,154 or 0.78% of our total revenue of \$4,873,797 from all required purchases by franchisees. We currently do not derive revenue from any other required purchases by franchisees. Our Affiliate does not derive any revenue from purchases by franchisees.

Computer/Point of Sale System

You are required to purchase and use the computer and point of sale system described in ITEM 11 of this Disclosure Document.

Insurance

In addition to any other insurance that may be required by applicable law, any lender or lessor, you must purchase and maintain in effect during the term of the Franchise Agreement the type and amount of insurance specified in Section 15 of the Franchise Agreement - These insurance policies include a comprehensive general liability policy with minimum limits of \$1,000,000; automobile liability policy with minimum limits of \$1,000,000; workers compensation insurance with minimum limits of \$100,000 (or higher if we required by your state): and property insurance coverage at full replacement cost. Your insurance policies must name us as an additional insured and/or loss payee.

Miscellaneous

We may negotiate purchase agreements, including group rates and price terms with suppliers, for purchases of equipment and supplies necessary for the operation of the Franchise Business. Presently, there are no such purchase or supply agreements in effect and no purchasing or distribution cooperatives that you must join. During our most recent fiscal year, we did not receive any rebates from suppliers due to franchisee purchases. However, in the future we may receive volume rebates, discounts, or other benefits from suppliers.

We estimate that approximately 5% to 15% of your expenditures for purchases in establishing your Franchise Business will be for goods and services that must be purchased either from us or an Approved Supplier or in accordance with our standards and specifications. We estimate that approximately 30% to 40% of your expenditures on an ongoing basis will be for goods and services that must be purchased from either us or an Approved Supplier or in accordance with our standards and specifications.

We do not provide material benefits to you (such as renewal rights or the right to open additional The Glass Guru Businesses) based on whether you purchase through the sources we designate or approve; however, purchases of unapproved products or purchases from unapproved suppliers in violation of the Franchise Agreement will entitle us, among other things, to terminate the Franchise Agreement.

ITEM 9. FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the Franchise Agreement and other agreements. It will help you find more detailed information about your obligations in these agreements and in other ITEMS of this Disclosure Document.

	Obligation	Section in the Franchise Agreement	Disclosure Document Item
a.	Site selection and acquisition/lease	Section 5	ITEMS 11 and 12
b.	Pre-opening purchases/leases	Sections 5, 12 and 15	ITEMS 7 and 8
C.	Site development and other pre- opening requirements	Sections 5 and 8	ITEMS 7, 8 and 11
d.	Initial and ongoing training	Section 8	ITEMS 6, 7 and 11
e.	Opening	Sections 5 and 8	ITEM 11
f.	Fees	Sections 3, 8, 10, 11, 13, 15, 18, and 21.	ITEMS 5, 6 and 7
g.	Compliance with standards and policies/Operating Manual	Sections 6, 7, 9, 10, and 13	ITEMS 8 and 16
h.	Trademarks and proprietary information	Sections 6, 7, and 9	ITEMS 13 and 14
i.	Restrictions on products/services offered	Sections 5, 6, and 13	ITEMS 8 and 16
j.	Warranty and customer service requirements	Section 13	Not applicable
k.	Territorial development and sales quotas	Section 5	ITEM 12
l.	Ongoing product/service purchases	Section 13	ITEMS 8 and 11
m.	Maintenance, appearance, and remodeling requirements	Sections 5, 10, and 13	ITEM 6 and 11
n.	Insurance	Section 15	ITEMS 6, 7 and 8
0.	Advertising	Section 11	ITEMS 6 and 11
p.	Indemnification	Section 21	ITEM 13
q.	Owner's participation/management/ staffing	Section 13	ITEM 15

	Obligation	Section in the Franchise Agreement	Disclosure Document Item
r.	Records and reports	Section 12	Not applicable
S.	Inspections and audits	Sections 6, 7 12,14 and 18	ITEMS 6, 11 and 13
t.	Transfer	Section 18 and Exhibit 1 and 5	ITEM 17
u.	Renewal	Section 4 and Exhibit 1 and 5	ITEM 17
V.	Post-termination obligations	Section 17 and Exhibit 2 and 5	ITEM 17
W.	Non-competition covenants	Sections 7 and 17 and Exhibit 2 and 5	ITEM 17
X.	Dispute resolution	Section 23 and Exhibit 5	ITEM 17
у.	Other	Not applicable	Not applicable

ITEM 10. FINANCING

We do not guarantee your lease or other obligations.

If you meet Franchisor's credit standards, we may finance a portion of the Initial Franchise Fee. The portion financed must <u>not</u> exceed 50% of the total non-refundable Initial Franchise Fee. The availability of this installment payment option will be subject to the sole discretion of the Franchisor. If Franchisor offers financing it will be over a maximum term of 60 months at an APR of 6% - 9%, based on Franchisee credit score, in a loan (the "Loan") using a standard form Promissory Note (the "Note") and Security Agreement (the "Security Agreement"), copies of which are attached to the Franchise Disclosure Document and marked as *Exhibit "I*". The Note may be prepaid without penalty during the 60-month term. Security for the Note will be any and all assets of Franchisee's business and must be personally guaranteed by you and your spouse and/or all shareholders of your corporation. If you do not pay on time, we can call the loan and demand immediate payment of the full outstanding balance of the loan and obtain all costs and fees incurred in obtaining payment, if a collection action is necessary.

We also have the right to terminate your Franchise Agreement if you do not make your payments on time during the term of the Note (see Franchise Agreement, section 16.2.2.1). We may assign the Note to a third-party lender who may be immune under the law to any defenses to payment you may have against Us (see Promissory Note). Should you default on the Note, we may take possession and sell your assets to satisfy all sums due, under the terms of the Security Agreement.

Any financing undertaken by the Franchisor will comply with the required state law.

SUMMARY OF FINANCING

Item Financed	Amount Financed	Down	Term	APR	Monthly	Prepay	Security	Liability	Loss of Rights on Default
Initial Fee	50% or less	50% or more	5 Years	6% Credit Score >700	approximately\$290.00 to \$483.00 depending on amount financed	None	Assets & Personal Guarantee	Loss of Franchise and payment of loan balance	Collection of your accounts receivable. You must pay collection costs and attorneys' fees
Initial Fee	50% or less	or more	5 Years	7% Credit Score 650- 699	approximately\$297.00 to \$495.00 depending on amount financed	None	Assets & Personal Guarantee	Loss of Franchise and payment of loan balance	Collection of your accounts receivable. You must pay collection costs and attorneys' fees
Initial Fee	50% or less	or more	5 Years	8% Credit Score 600- 649	approximately\$304.00 to \$507.00 depending on amount financed	None	Assets & Personal Guarantee	Loss of Franchise and payment of loan balance	Collection of your accounts receivable. You must pay collection costs and attorneys' fees
Initial Fee	50% or less	50% or more	5 Years	9% Credit Score <600	approximately\$311.00 to \$519.00 depending on amount financed	None	Assets & Personal Guarantee	Loss of Franchise and payment of loan balance	Collection of your accounts receivable. You must pay collection costs and attorneys' fees

[BALANCE OF THIS PAGE INTENTIONALLY LEFT BLANK]

SUPPLEMENTAL INFORMATION FOR FINANCING

- 1. The financing provided by Franchisor includes only the non-refundable Initial Franchise Fee as described in this ITEM 10. We do not provide financing or guarantee your lease or any other obligations.
- 2. We currently do not plan to assign the Notes to third party lenders but reserve the right to do so.
- 3. We plan to offer installment payment terms for up to 50% of the non-refundable Initial Franchise Fee, which in view of the stated maximum amount of the non-refundable Initial Franchise Fee is not likely to exceed \$25,000.
- 4. APR. We will offer fixed rate financing at 6% 9% annual percentage rate or "APR" at a time when the prime lending rate is approximately 8.50% APR as of the date of this document. Franchisor reserves the right to adjust the interest rate in the event the prime lending rate increases above 9.50%.
- 5. The term for repayment of the Note will not exceed 60 months and payment must be made monthly unless otherwise agreed.
- 6. The security interest for the Note will be in the assets of the franchise business, the Franchise Agreement. In addition, all owners and their spouses must personally guaranty the Note.
- 7. There will be no pre-payment penalty in connection with the Note.
- 8. Upon default, you will be liable for the balance due on the Note, court costs, attorney's fees and other costs of collection, as well as loss of franchise rights and all assets of the Franchise.
- 9. Among other things, the Note and Security Agreement permit us to exercise all rights and remedies permitted under the Uniform Commercial Code, including without limitation, collection of payments due to you from your customers, recovery of legal costs and expenses including attorneys' fees. If we exercise the right to assign the Loan, you may also lose your right to assert certain defenses against our assignee. You should read all applicable documents carefully and consult your own legal counsel for a detailed explanation of your rights and your obligations.

ITEM 11. FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

- 1. Before you open your Franchise Business, we will:
 - A. Designate the Area of Primary Responsibility (territory/APR) within which you will operate the Franchise Business. (Section 2.3 and Section B of Exhibit 1) Additionally,
 - (1) We do not own and will not lease any real property to you.
 - (2) We will agree on your Area of Primary Responsibility before you sign the Franchise Agreement. You must run your Franchise Business from an office located inside of your Area of Primary Responsibility unless otherwise approved by us. (See Section 5 of the Franchise Agreement for a detailed discussion of various requirements for establishing a Franchise Business).
 - (3) You must select an approved location for the Franchise Business within 60 days of signing the

Franchise Agreement. If you do not select a site within 60 days, we may terminate the Franchise Agreement. Additionally, if we are unable to reach an agreement for an approved site, we may terminate the Franchise Agreement. (Sections 5.1 and 5.2)

- (4) The main factors the Franchisor considers in approving sites that are suitable for a glass shop are: size, cost, proximity, and visibility. Your office must be located inside of your Area of Primary Responsibility unless otherwise approved by us. (Section 5.1)
- B. Furnish you with an initial partial inventory of The Glass Guru Restoration Tools and an initial inventory of branded products upon receipt of payment. Further information about The Glass Guru Restoration Tools we supply to you and other franchisees is included in ITEMS 5, 7 and 8 of this Disclosure Document. (Sections 5.4.1.3, 11.1,13.2)
- C. Provide you with training and guidance on establishing prices, such as setting minimum or maximum prices at which you should sell products and services.
- D. Provide you with our criteria for vehicles and other equipment necessary for the operation of the Franchise Business. (Section 5.4)
- E. Provide an initial training program for your Designated Manager, one office assistant and up to 2 additional field or office assistants with your request and with our prior written approval. This training is described in detail later in this ITEM. (Section 8.1)
- F. Provide to you the option, at your expense, of training and guidance to assist you with the opening of the Franchise Business. (Section 8.3)
- G. Provide you with access to an electronic copy of the Manual. The Table of Contents of the Operations Manual, along with number of pages devoted to each section, is included as Exhibit D to this Disclosure Document. (Section 9.3)

2. After the opening of the Franchise Business, we will:

- A. Periodically, advise and offer general guidance to you by telephone, e-mail, webinars, conference call, intranet, newsletters, and other methods; however, you will use your own judgment to make all business decisions and should not rely solely upon any advice given or statements made by us. Our guidance is based on our and our franchisees' experience in operating The Glass Guru Business. Such advice and guidance may consist of knowledge and experience relating to the authorized services or products, as well as operational methods, accounting procedures, and marketing and sales strategies. (Section 14.1)
- B. At our discretion, make periodic visits to the Franchise Business to give you assistance with or guidance on various aspects of the operation and management of a The Glass Guru Business. We may prepare written reports outlining any suggested changes or improvements in the operations of the Franchise Business and detail any deficiencies that become evident as a result of any such visit. If we prepare a report, you may request a copy from us. (Section 14.2)
- C. Make available to you operations assistance and ongoing training and hold Conventions for franchisees as we deem necessary. (Sections 8.1, 8.6 and 8.7)
- D. Provide training to your employees who do not attend and successfully complete the initial training program in accordance with Section 8.1. (Section 8.2)

- E. Make available to you changes and additions to the System as generally made available to all franchisees. (Section 14.3)
- F. Provide advertising and promotional templates including ad-slicks, brochures, fliers and other materials for your use. (Section 14.4)
- G. Franchisees may utilize their own ad materials, but those items must be approved by the Franchisor. We will approve forms of advertising materials you will use for Local Advertising, Brand Identity Package Advertising and Cooperative Advertising. Cooperative advertising is not required of franchisees. (Sections 11.1, 11.2, 11.4)
- H. We will conduct advertising for the franchise system and individual locations through the utilization of the required Marketing Fee. The media selected is at the sole discretion of the Franchisor. The programs may be local, regional, or national. The advertising may be executed by Franchisor's marketing staff or through partnerships with appropriate third parties or agencies. We do not warrant that any franchisee will benefit directly or pro rata from any advertising for the franchise system.
- I. Provide you with modifications to the Manual and intranet knowledge base of resources as they are made available to franchisees. (Section 9.4)

3. Advertising and Promotion:

- A. Prior to beginning operation of the Franchise Business, you must spend the published rate in the Manual; currently \$5,000 (plus applicable shipping and sales taxes) on the Brand Identity Package. (Section 11.1)
- B. Each month, you must spend the greater of the minimum monthly local advertising expenditure of \$1,250 or 5% of the prior year's Gross Sales divided by the months of operation in the prior year (Section 11.2).
- C. You and all franchisees are required to pay a Marketing Fee on a monthly basis to us equal to 3% of the corresponding monthly Gross Sales. Marketing Fees must be made at the same time and in the same manner as Royalty Fees. All Affiliate-Owned outlets contribute at the same rate and frequency as required of all other franchisees. Franchisor is not required to spend any designated amount of Ad Funds in any franchisee's territory.

We agree to oversee all marketing programs, with sole discretion over the creative concepts, materials and media used in such programs, and the placement and allocation of that marketing. These items may be delivered by internal staff or external third-party providers at our discretion. We do not warrant that any franchisee will benefit directly or pro rata from any expenditures by the Marketing Fee. The programs may be local, regional or system-wide. The monies deposited into the Marketing Fee will be used to meet the costs of, or to reimburse us, for our costs of producing, maintaining, administering and directing consumer advertising, including the cost of preparing and conducting television, radio, Internet, magazine and newspaper advertising campaigns and other public relations activities, developing or hosting an internet web page or site and similar activities, employing advertising agencies to assist, and providing promotional brochures and other marketing materials to franchisees. No percentage of advertising fees are used to solicit new franchise sales. You may contact and request from the Franchisor, via email to compliance@theglassguru.com, a report obtaining an accounting of the Marketing Fee. Presently, there is not an Advertising Council of franchisees in existence to advise franchisor on its ad policies.

The Marketing Fee is audited annually as part of the corporate audit of Franchisor. The fee is subject to additional audit by request of franchisees. Any fees gathered in a year that are not expended are rolled into the next year. A period accounting of how the fees are spent are provided in updates to the franchisees via electronic communication, live updates, conference updates, and annual FDD disclosures that are available to all franchisees. In the fiscal year ended December 31, 2023, the contributions collected by the Marketing Fee were <u>spent</u> as follows:

Type of Expense	% of Expenditure
Creative Design & Production	11.92%
Corp Sponsored Media Placements	32.12%
Website Design/Maintenance/Hosting	6.09%
SEO/Digital Marketing Management Platforms	1.97%
Call Tracking	7.66%
Software & Subscriptions (Creative, Business Intelligence, etc.)	14.09%
Outside Agencies and Contractors	4.18%
Other - Administrative	0.05%
Marketing Specific Salaries (Management and Administration)	21.92%
Totals	100.00%

- D. All advertising and marketing expenses must be recorded in the required QuickBooks accounting software, and such expense reports made available to Franchisor upon request. Unless authorized in advance by Franchisor in writing, in the event that you don't spend the amount required on monthly local advertising in any given month, you will be required to pay franchisor the difference between what was required versus what was actually spent. Franchisor agrees to utilize these funds for general marketing purposes. Payment will be collected along with Royalties and Marketing Fees in the next royalty period or as published in the Manual, if assessed at a lesser interval.
- E. Under the Franchise Agreement, you are restricted from establishing a presence on, or marketing using, the Internet in connection with the Franchise Business without our prior written consent. We have established and maintain an Internet website at the uniform resource locator www.theglassguru.com that provides information about the System and about The Glass Guru Business. We may (but we are not required to) include at The Glass Guru website an interior page containing information about your Franchise Business and a dedicated website for your Franchise Business. If we include such information on The Glass Guru website, or your Franchise Business website, we have the right to require you to prepare all or a portion of the page(s), at your expense, using a template that we provide. All such information is subject to our approval prior to posting. We retain the sole right to market on the Internet, including the use of websites, domain names, uniform resource locators, linking, social media pages, search engines (and search engine optimization techniques), banner ads, meta-tags, marketing, auction sites, e-commerce, and co-branding arrangements. You may be requested to provide content for our Internet marketing, and you must follow our intranet and Internet usage rules, policies, and requirements. We retain the sole right to use the Marks on the Internet, including on websites, as domain names, directory addresses, search terms and meta-tags, and in connection with linking, marketing, co-branding, and other arrangements. We retain the sole right to approve any linking to, or other use of, The Glass Guru website. (Section 11.5)
- F. Franchisees are not currently required to be a part of a local or regional advertising cooperative.

4. Computer/Point-of Sale System

We have the right, under the Franchise Agreement, to require you to purchase and use, maintain, (and replace or upgrade as necessary) the computer hardware and software programs and point of sale equipment that we designate, as outlined in the Manual. (Section 12.5) Presently, we require you to have at least one computer workstation loaded with Microsoft Office Suite and QuickBooks accounting software and GPS Insight business and customer manager software or an equivalent software approved by us and other software as described below. As previously outlined, QuickBooks is an additional third party expense you will incur apart from your technology fee. The cost of GPS Insight software is paid through the required technology fee and is referenced in the table within Item 7 as Software Systems and additionally outlined in Note 12 of Item 7.

You are required to enter customer contact information - full name, address, phone number, and e-mail address (whenever available) within 48 hours of initial customer contact. Failure to enter customer information within 48 hours of initial customer contact will result first in a written warning. Each subsequent monthly written notice of failure to comply will also result in a fine to be paid to us as outlined in the Manual; currently \$175. Should you fail to enter customer information one or more times during a calendar month, in three (3) or more calendar months during a calendar year, you may be required to obtain, and present to us, on or before 90 days post yearend, year-end audited financial statements, prepared by and independent accounting firm, at your expense.

You may use currently owned equipment and software, provided it is suitable for the requirements described in this section. However, if you have no computer equipment or software, the estimated cost to purchase the required computer system is approximately \$1,250 for used equipment and the required software, up to \$4,800 for new equipment. Those estimates will vary depending on geography, computer experience, and other factors. You should consult an independent technical advisor of your choosing for a complete and accurate assessment of your needs and your costs. Contractually, Franchisor will not be provided access to your owned equipment or direct access to your QuickBooks software. Franchisor will have access to all other third-party technology provided or required by franchisor through online access.

You are not required to enter into any ongoing maintenance or support agreements, but you may find it advantageous to do so. At your own expense, you must update or upgrade computer hardware and software as we deem necessary but contractually, not more than once per year. (Section 10.2) There are no contractual limits on the annual cost for maintenance, updates, upgrades, and support, but it is estimated to be \$1,500. Contractually, the Franchisor has no obligation for maintenance, repairs, updates, or upgrades to franchisee's computerized system.

You will be required to pay us a technology fee that covers the direct costs of approved software, software licenses, subscriptions, development, maintenance, and hosting for all systems substantially benefiting franchise locations including but not limited to feature sets like; field service management (FSM), business productivity software, Intranet, learning management (LMS) system, and business intelligence systems (BI), and any additional systems/software otherwise authorized and/or required in the Operations Manual. The monthly fee as published in the manual is currently, the greater of \$360 or 1% of Gross Sales, with a maximum fee of \$760 being assessed. You are also required to use Intuit QuickBooks Online for your accounting software. The QuickBooks license is not included with the technology fee, and you must acquire and license the software directly with Intuit or an approved reseller at a cost of \$75 to \$150 per month.

We have the right to independently access all information you collect or compile at any time without first notifying you. (Section 12.6)

5. Typical Length of Time Before Operation

We estimate that the typical length of time between the signing of the Franchise Agreement and the opening of The

Glass Guru Business is 60 days. Factors that may affect your beginning operations include ability to secure permits, zoning and local ordinances, weather conditions and delays in installation of equipment and fixtures. You are required to open your Franchise Business and be operational within 90 days after signing the Franchise Agreement.

6. Training

We will conduct an initial training program that the owner and any corporate approved Designated Manager must attend and complete to our satisfaction. Training is offered at various times during the year, depending on the number of new franchisees entering our franchise system. Although initial training is mandatory for the Designated Manager, it is also available, by request for franchisee's employees, specifically one office assistant and up to two additional field or office assistants with your request and our prior approval and additional costs as previously outlined. Training will take place at our headquarters or at another Qualified Training Location we designate on an as needed basis as we may determine. The initial training program covers all material aspects of the operation of a The Glass Guru Business, including such topics as technical procedures, maintenance of quality standards; customer service techniques; sales and marketing methods; use of CRM (FSM/POS) programs; record keeping and reporting procedures; other operational issues and on-the-job training. All franchisees and Designated Managers must complete initial training to our satisfaction. We expect franchisees will advance through the training program at different rates depending on a variety of factors such as background and experience. Accordingly, the time frames provided in the following chart are an estimate of the time it will take to complete training. Initial training must be completed within 90 days of signing your Franchise Agreement. If you replace yourself with a new corporate approved Designated Manager, your new Designated Manager must attend our training program. We do not charge for initial training; however, you must pay for all travel costs and living expenses for yourself and/or your Designated Manager and any other of your attendees that require training. You will be charged fees for additional training of a new Designated Manager or any additional staff that require training after the initial training program has been conducted.

You are responsible for training your own employees, but any employee may come for additional training at any Qualified Training Location upon request and with our prior approval. You will be charged fees for additional training. Our standard fee is \$500 per person per day. The initial training must be completed within 90 days of signing the Franchise Agreement.

TRAINING PROGRAM

Subject	Hours of Classroom Training	Hours of On- The-Job Training	Location
Pre-Launch Checklist	0	40	Home
LMS Video Training	10	0	Home
Orientation, Products and Services	4	4	Training Center
Restoration Demos and Glazing Basics	8	4	Training Center
Pricing, Estimating Training	4	4	Training Center
Customer Service in Office	3	1	Training Center
Admin Training - CRM Workflow & Recordkeeping	2	0	Training Center
Supplies, Vendors and Goal Setting	3	0	Training Center
Marketing	2	0	Training Center
	36	53	-

Training may be conducted at the Corporate Headquarters, the Affiliated-Owned location, or any Qualified Training Location deemed appropriate by us.

The instructional materials used in the initial training program include the Manuals, LMS courses, guides, forms, and QuickBooks manuals.

Currently the following individuals facilitate and conduct the initial training program.

VP Franchise Operations | Dave Hull

Mr. Hull joined our predecessor in January 2011 and has served as VP of Franchise Development, Chief Development Officer, and most recently VP Franchise Operations. He was General Manager of our Affiliate, The Glass Guru of Roseville, CA from October 2009 to March 2011 and has served as the managing owner from 2018 to present. Mr. Hull has 13 years of experience with our Affiliate and/or us, and 30 years of combined glass and glazing industry/trade experience.

VP Franchisee Business Development | Rob Lopez

Mr. Lopez joined our predecessor as Director of Franchise Development in November 2015 and was promoted to VP of Franchise Business Development in 2020. Mr. Lopez has 8 years of experience with us, and 17 years of combined franchise industry experience.

If circumstances require, a substitute trainer with a similar work history and experience for the subject being trained may provide training. We also reserve the right to name additional trainers periodically. There are no limits on our right to assign a substitute to provide training provided the substitute is approved by us to teach the topic, has similar work history and experience.

Periodically, we may require that previously trained and experienced franchisees or Designated Managers attend refresher-training programs to be conducted at our headquarters or another location. We may charge a fee for these training programs. Travel, living costs, and employees' salaries incurred for attending these programs will be at your sole expense. We will not require you to attend more than 2 of these programs in any calendar year and these programs will not collectively exceed 7 days during any calendar year. (Section 8.6 and Section 8.8)

We may hold a Convention of franchisees, or Regional Workshops or other educational events, and if we do, you and/or your Designated Manager must attend any Convention. We charge a registration fee for attending these events at the rates published in the Manual. If you (or your Designated Manager) fail to attend, you must still pay the registration fee. You are responsible for all travel costs, room and board and employees' salaries incurred by you and your Designated Manager in attending the convention. (Section 8.7)

ITEM 12. TERRITORY

You will receive an exclusive territory which we refer to as your Area of Primary Responsibility (APR). The "Area of Primary Responsibility" is an area surrounding the Franchise Business location and is defined using US zip codes. This area is considered the geographic market boundary for the Franchise Business. The population of an Area of Primary Responsibility for a Franchise Business will typically range from 150,000 to 800,000 but there is no minimum and no maximum. Population should be determined by referencing the most recent figures available to us from the United States Census Bureau at the time the Area of Primary Responsibility is designated. The Area of Primary Responsibility is defined in Section B of Exhibit 1 of the Franchise Agreement. You may solicit and accept work outside of your APR; however, that work must not be part of another franchise owner's APR. You will incur an additional 2% Non-APR Fee for doing business in the Non-APR zip code not owned by another franchisee. Service of immediate family members is permitted in a Non-APR fee. Providing services or selling product to non-immediate family members in the APR of another franchise owner is otherwise prohibited..

Should any zip code be purchased by another franchisee in the future, all customer data including those that you may have previously served will belong to the new franchise owner of that APR/zip code.

As long as the Franchise Agreement is in full force and effect, we will not establish or operate or grant others the right to establish and operate a The Glass Guru Business at any location within the Area of Primary Responsibility since this is considered an exclusive territory.

We may establish ourselves or through an affiliate or grant the right to others to establish and operate a The Glass Guru Business at any location outside of the Area of Primary Responsibility. Franchisor and its affiliates will not conduct business within your Area of Primary Responsibility with the direct exception for immediate family identified in the following paragraph. Services provided to immediate family members only of franchisor or its affiliates will not result in compensation to the franchisee whose Area of Primary Responsibility is affected. All other advertising or services within your Area of Primary Responsibility are prohibited by all parties. We retain all rights and control with respect to the Marks and System, outside the Area of Primary Responsibility to: (a) establish or operate, and to grant others the right to establish or operate, other businesses offering the same or similar products or services as the Franchise Business utilizing other trade names, trademarks and service marks within or outside of the Area of Primary Responsibility; (b) establish or operate, and to grant others the right to establish or operate businesses offering products and services different than those offered by the Franchise Business utilizing the Marks or other trade name, trademarks and service marks within or outside of the Area of Primary Responsibility; (c) to acquire or be acquired by a company establishing businesses identical or similar to the Franchise Business, even if the other business operates franchises and/or licenses competitive businesses anywhere, including your Area of Primary Responsibility; (d) sell or otherwise distribute products and services similar to those offered through the Franchise Business through any alternate channel of distribution, including through retail outlets at a fixed location by direct mail and Internet sales; and (e) engage in any activities not expressly forbidden by the Franchise Agreement.

You may use any approved channels of advertising to make sales, but you may not provide services or sell products in the Area of Primary Responsibility of another The Glass Guru Franchise Business Location; provided, however, you do have the right to provide services and sell products to immediate family located inside the Area of Primary Responsibility of others. For this purpose, immediate family is defined as children, grandchildren, mother, father, stepparents, grandparents, sisters and brothers.

You may only have one Approved Location within the Area of Primary Responsibility granted to you. However, you may request our consent to you establishing any additional locations approved within the Area of Primary Responsibility. If we consent to such request, we may require you to execute an additional Franchise Agreement for the additional location and your original Area of Primary Responsibility will be split between the original Franchise Agreement and the new Franchise Agreement in the manner agreed upon by the parties. Upon the execution of the second Franchise Agreement, your original Agreement will be amended in writing to revise the Area of Primary Responsibility.

You may not relocate the Approved Location(s) without our prior written consent. If the lease for the Approved Location expires or is terminated due to no fault of yours or if the premises of the Approved Location are destroyed, condemned, or become otherwise unusable, we may allow you to relocate the Approved Location. Any such relocation will be at your sole expense and must comply with the requirements listed in Sections 5.1 through 5.8 of the Franchise Agreement. We have the right to charge you for any costs we incur in providing you assistance, including legal and accounting fees, however, we have no obligation to provide relocation assistance. If a substitute site is not mutually agreed upon within 90 days after the lease expires or is terminated or the Approved Location becomes unusable, the Franchise Agreement will terminate as provided in Section 16.2.1.1 of the Franchise Agreement.

You maintain rights to your Area of Primary Responsibility if you meet the Minimum Performance Requirements specified in the Franchise Agreement, Section 5.11 – For the 1st through 4th years of

operation of a Unit in the Area of Primary Responsibility, Franchisee must achieve annual Gross Sales of at least the following amounts: (i) One Hundred Fifty Thousand Dollars (\$150,000) for the first (1st) year of operation; (ii) Two Hundred Thousand Dollars (\$200,000) for the second 2nd year of operation; (iii) Two Hundred Fifty Thousand Dollars (\$250,000) for the third (3rd) year of operation; and (iv) Three Hundred Thousand Dollars (\$300,000) for the fourth (4th) year of operation. After a Franchise Business has been operating in the Area of Primary Responsibility for four (4) years, for each subsequent year of operation Franchisee must: (A) achieve Gross Sales of at least Three Hundred Fifty Thousand Dollars (\$350,000) and (B) perform in the top ninety percent (90%) of all Franchise Business operating in the country of origin in the annual growth percentage of Gross Sales as determined by Franchisor. If Franchisee business fails to achieve the Minimum Performance Requirements or is not in compliance with the Minimum Performance Requirements at the time of renewal, Franchise Agreement. Additionally, failure to meet the minimum performance requirements may result in termination of the Franchise Agreement.

You maintain rights to your Area of Primary Responsibility with no price increase during the term of your Franchise Agreement, even if the population increases over time. Apart from the Minimum Performance Requirements detailed above, your rights to the Area of Primary Responsibility are not dependent upon you achieving a certain sales volume, market penetration or other continency. As long as you are in compliance with the Franchise Agreement and the Franchise Agreement is in effect, we will not modify or alter your Area of Primary Responsibility. Unless otherwise agreed in writing, you do not acquire any additional options, rights of first refusal, or similar rights to acquire additional franchises wherever located.

The franchisor currently does not operate and has no plans to operate, or franchise any similar or related business under a different trademark.

Furthermore, you must not provide services or sell products in areas inside the Area of Primary Responsibility of another franchisee; provided, however, Franchisee has the right to provide services and sell products to immediate family located inside another franchisee's Area of Primary Responsibility. If we determine that you provide services to or sell products in to customers in the Area of Primary Responsibility of another The Glass Guru franchisee, other than to immediate family, prior to obtaining the express written consent of Franchisor, you must pay the franchisee that has the right to the infringed Area of Primary Responsibility an amount equal to the total sums you received for the Services and Products, less the actual cost you paid for the Products sold and any third party hired labor costs. All sums must be paid within 10 days after receiving notice from Franchisor. A second violation could result in the termination of the Franchise Agreement.

ITEM 13. TRADEMARKS

We grant our franchisees the right to operate The Glass Guru Business under the name "The Glass Guru," which is the principal Mark used to identify our System. You may also use any other current or future Mark to operate your Franchise Business that we designate in writing, including the logo on the front of this Disclosure Document and the trademark listed below. By "Mark," we mean any trade name, trademark, service mark or logo used to identify The Glass Guru Business. The following Mark(s) has been registered on the Principal Register of the U.S. Patent and Trademark Office ("USPTO") on the dates shown below:

Mark	Registration Number	Registration Date	
Dead - THE GLASS GURU	3226008	April 3, 2007	
Live -THE GLASS GURU	5628020	December 11, 2018	
Live -THE GLASS GURU logo	5726951	April 16, 2019	

We intend to file all required affidavits to maintain and renew the registration when due.

We know of no currently effective material determinations of the USPTO, United States Copyright Office, or trademark trial and appeal board, the trademark administrator of any state or a Court regarding the patent or; copyright; pending infringement, opposition, or cancellation; or pending material litigation involving the Marks.

There are no agreements currently in effect that limit our rights to use or license the use of the Marks in any manner material to the franchise. We are not aware of any infringing uses of our Marks that could materially affect your use of the Marks.

You will not receive any rights to the Marks other than the nonexclusive right to use them during the operation of your Franchise Business. You may only use the Marks in accordance with our standards, operating procedures, and specifications. Any unauthorized use of the Marks by you is a breach of the Franchise Agreement and an infringement of our rights in the Marks. You may not contest the validity or ownership of the Marks, including any Marks that we license to you after you sign the Franchise Agreement. You may not assist any other person in contesting the validity or ownership of the Marks.

You must immediately notify us of any apparent infringement of, or challenge to your use of, any Mark, or any claim by any person of any rights in any Marks, and you may not communicate with any person other than us and our counsel regarding any infringements, challenges or claims unless you are legally required to do so; however, you may communicate with your own counsel at your own expense. We may take whatever action we deem appropriate in these situations or may take no action; we have exclusive control over any settlement or proceeding concerning any Mark. You must take any actions that, in the opinion of our counsel, may be advisable to protect and maintain our interests in any proceeding or to otherwise protect and maintain our interests in the Marks.

We may but are not contractually obligated to reimburse you for all expenses reasonably incurred by you in any trademark or similar proceeding disputing your authorized use of any Mark, provided that you have timely notified us of the proceeding and have complied with the Franchise Agreement and our directions in responding to such proceeding. Contractually, we do not have an obligation or requirement to protect franchisees against claims of infringement or unfair competition arising out of the franchisee's use of the trademark but may elect to do so. Contractually we are not required to participate in franchisee's defense and/or indemnify the franchisee for expense or damages if the franchisee is a party to an administrative or judicial proceeding involving a trademark licensed to be used by the franchisee, or if a proceeding is resolved unfavorably to the franchisee. At our option, we or our designee may defend and control the defense of any proceeding arising directly from your use of any Mark. This indemnification does not include your expenses in removing signage or discontinuance of the use of the Marks, does not apply to litigation between us and you wherein your use of the Marks is disputed or challenged by us, and does not apply to any separate legal fees or costs incurred by you in seeking independent counsel separate from legal counsel representing our and your use of the Marks.

We can require you to modify or discontinue the use of any Mark and to use other trademarks or service marks. If we adopt and use new or modified Marks, you may be required to add or replace equipment, signs, and fixtures, and you may have to make other modifications as necessary to maintain uniformity with our current standards and specifications; however, you will not be required to spend an amount unreasonably disproportionate to your original investment during the initial term of the Franchise Agreement. We will not be required to reimburse you

for modifying or discontinuing the use of a Mark or for substituting another trademark or service mark for a discontinued Mark. We are not obligated to reimburse you for any loss of goodwill associated with a modified or discontinued Mark.

We may but are not contractually required to reimburse you for all of your expenses reasonably incurred in any legal proceeding disputing your authorized use of any Mark, but only if you notify us of the proceeding in a timely manner and you have complied with our directions with regard to such proceeding. We have the right to control the defense and settlement of any such proceeding. Our reimbursement does not include your expenses for removing signage or discontinuing your use of any Mark. Our reimbursement also does not apply to any disputes where we challenge your use of a Mark. Our reimbursement does not apply to legal fees you incur in seeking separate, independent legal counsel.

You must use the Marks as the sole trade identification of the Franchise Business, but you may not use any Mark or part of any Mark as part of your corporate or other business entity name in any modified form unless authorized. You may not use any Mark in connection with the sale of any unauthorized products or services, or in any other manner that we do not authorize in writing. You must obtain a fictitious or assumed name registration if required or permitted by your state or local law.

You must notify us if you apply for your own trademark or service mark registrations that does not include or is not similar to our Marks. You cannot operate the Franchise Business under any other marks you may register. The Franchise Business can only be operated under our Marks. You must not register or seek to register as a trademark or service mark, either with the USPTO or any state or foreign country, (i) any of our Marks or a trademark or service mark that is confusingly similar to any of our Marks, and (ii) any trademark or service mark for the glass, door, or window industry.

You may not establish, create, or operate an Internet site or website using any domain name containing the words "THE GLASS GURU" or any variation of those words without our prior written consent. You may not advertise on the Internet without our consent.

ITEM 14. PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

As of the date of this Disclosure Document, we do not hold any applicable patents, nor do we have any patents pending. We own copyrights in the Manual, marketing materials and other copyrightable items that are part of the System. While we claim copyrights in these and similar items, we have not registered these copyrights with the United States Registrar of Copyrights and need not do so to protect them. You may use these items only as we specify while operating the Franchise Business and you must stop using them if we direct you to do so.

We know of no currently effective determinations of the United States Patent and Trademark Office, the United States Copyright Office or a Court regarding the patent or copyright materials. Our right to use or license copyrighted items is not materially limited by any agreement or known infringing use. We know of no agreements in effect that limit the use of any patent, patent application, or copyright. Further we know of no current patent or copyright infringement that could materially affect any franchisee.

You must immediately notify us of any apparent infringement of, or challenge to your use of, any patents, copyrights, or proprietary information. You may not communicate with any person other than us and our counsel regarding any infringements, challenges or claims unless you are legally required to do so; however, you may communicate with your own counsel at your own expense. Although not contractually required to take action, we may take whatever action we deem appropriate in these situations or may take no action; we have exclusive control over any settlement or proceeding concerning any patents, copyrights, or proprietary information. You must take any actions that, in the opinion of our counsel, may be advisable to protect and maintain our interests in any proceeding or to otherwise protect and maintain our interests in the patents, copyrights, or proprietary information.

We may but are not contractually obligated to reimburse you for all expenses reasonably incurred by you in any proceeding disputing your authorized use of any patents, copyrights, or proprietary information, provided that you have timely notified us of the proceeding and have complied with the Franchise Agreement and our directions in responding to such proceeding. Contractually, we do not have an obligation or requirement to protect franchisees against claims of infringement or unfair competition arising out of the franchisee's use of the patents, copyrights, or proprietary information but may elect to do so. Contractually we are not required to participate in franchisee's defense and/or indemnify the franchisee for expense or damages if the franchisee is a party to an administrative or judicial proceeding involving any patents, copyrights, or proprietary information licensed or authorized to be used by the franchisee, or if a proceeding is resolved unfavorably to the franchisee. At our option, we or our designee may defend and control the defense of any proceeding arising directly from your use of any patents, copyrights, or proprietary information. This indemnification does not include your expenses in removing signage or discontinuance of the use of the patents, copyrights, or proprietary information, does not apply to litigation between us and you wherein your use of the patents, copyrights, or proprietary information is disputed or challenged by us, and does not apply to any separate legal fees or costs incurred by you in seeking independent counsel separate from legal counsel representing our and your use of the patents, copyrights, or proprietary information.

We can require you to modify or discontinue the use of any patents, copyrights, or proprietary information. If we adopt and use new or modified Marks, you may be required to add or replace equipment, signs, and fixtures, and you may have to make other modifications as necessary to maintain uniformity with our current standards and specifications; however, you will not be required to spend an amount unreasonably disproportionate to your original investment during the initial term of the Franchise Agreement. We will not be required to reimburse you for modifying or discontinuing the use of a patents, copyrights, or proprietary information. We are not obligated to reimburse you for any loss of goodwill associated with any modified or discontinued patents, copyrights, or proprietary information.

We may but are not contractually obligated to reimburse you for all of your expenses reasonably incurred in any legal proceeding disputing your authorized use of any patents, copyrights, or proprietary information, but only if you notify us of the proceeding in a timely manner and you have complied with our directions with regard to such proceeding. We have the right to control the defense and settlement of any such proceeding. Our reimbursement does not include your expenses for removing signage or discontinuing your use of any patents, copyrights, or proprietary information. Our reimbursement also does not apply to any disputes where we challenge your use of any patents, copyrights, or proprietary information. Our reimbursement does not apply to legal fees you incur in seeking separate, independent legal counsel.

We have developed certain Trade Secrets and other Confidential Information, including methods of business management, sales and promotion techniques, and know-how, knowledge of, and experience in, operating a The Glass Guru Business and consider this proprietary information. We will provide our Trade Secrets and other Confidential Information to you during training, in the Manual and as a result of the assistance we furnish you during the term of the franchise. You may only use the Trade Secrets and other Confidential Information for the purpose of operating your Franchise Business. You may only divulge Trade Secrets and other Confidential Information to employees who must have access to it in order to perform their job duties within the Franchise Business. You are responsible for enforcing the confidentiality provisions as toyour employees.

Certain individuals with access to Trade Secrets or other Confidential Information, including your owners, officers, directors, executives, managers, members of your professional staff, all employees and other individuals having access to Trade Secrets or other Confidential Information are required to sign nondisclosure and non-competition agreements in a form the same as or similar to the Nondisclosure and Non-Competition Agreement attached to the Franchise Agreement, where permitted by state law. We will be a third-party beneficiary of these agreements and we have the independent right to enforce their restrictive covenants.

All ideas, concepts, techniques, or materials concerning the Franchise Business, whether or not protectable intellectual property and whether created by or for you or your owners or employees, must be promptly disclosed to us and will be deemed our sole and exclusive property and a part of the System that we may choose to adopt and/or disclose to other franchisees. Likewise, we will disclose to you concepts and developments of other franchisees that we make part of the System. You must also assist us in obtaining intellectual property rights in any concept or development if requested.

Your use of the Manual, Trade Secrets or other Confidential Information in an unauthorized manner is a default of the Franchise Agreement that may result in automatic termination of the Franchise Agreement. Further information about termination of the Franchise Agreement following a default is included in ITEM 17 of this Disclosure Document.

ITEM 15. OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

The Franchise Business must always be under the direct full-time supervision of a corporate approved Designated Manager. The Designated Manager must attend and satisfactorily complete our initial training program before opening the Franchise Business. You must keep us informed of the identity of your current Designated Manager. If you replace your Designated Manager, he or she must be approved by us and attend the initial training program within 60 days. The Designated Manager need not be one of your owners.

As described in ITEM 14, your owners, officers, directors, executives, managers, members of your professional staff, all employees and other individuals having access to Trade Secrets or other Confidential Information are required to sign nondisclosure and non-competition agreements in a form the same as or similar to the Nondisclosure and Non-Competition Agreement attached to the Franchise Agreement. We will be a third-party beneficiary with the right to enforce the agreements.

Anyone who owns a 10% or greater interest in the entity, Franchisee's officers, directors, managers, and representatives, and any affiliate of Franchisee (including spouses) involved with or providing assets or services to the Franchised Business, must personally guarantee the performance of all of your obligations under the Franchise Agreement and agree to be personally liable for your breach of the Franchise Agreement by signing the Guaranty and Assumption of Obligations attached to the Franchise Agreement.

ITEM 16. RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

You must operate the Franchise Business in strict conformity with the methods, standards, and specifications in the Manual and as we may require otherwise in writing. You may not deviate from these standards, specifications, and procedures without our written consent.

You must offer the services and products we specify in strict accordance with our standards and specifications. You may not sell any services or products that we have not authorized, and you must discontinue offering any services or products that we may, in our sole discretion, disapprove in writing at any time.

We may change the types of authorized services and products that you may offer. There are no limits on our right to make these changes.

Periodically, we may allow certain services or products that are not otherwise authorized for general use as a part of the System to be offered locally or regionally based upon such factors as we determine, including test marketing, your qualifications, and regional or local differences. We will agree on the royalty and other fees applicable to these products and services in advance.

ITEM 17. RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists important provisions of the Franchise and related Agreements. You should read these provisions in the agreements attached to this disclosure document.

Provision	Section in the Franchise Agreement	Summary
a. Length of franchise term	Section 4.1	The initial term is 5 years.
b. Renewal or extension of the term	Section 4.2	You may renew for 3 additional terms of 5 years each, subject to (c) below. If you fail to meet any one of these conditions, we may refuse to renew or extend the terms of your Franchise Agreement.
c. Requirements for you to renew or extend	Section 4.2	You may renew the Franchise Agreement if you: have substantially complied with the provisions of the Franchise Agreement; have made capital expenditures as necessary to maintain uniformity with the System; have satisfied all monetary obligations owed to us; are not in default of any provision of the Franchise Agreement or any other agreement between you and us; have given timely written notice of your intent to renew; sign a current Franchise Agreement which may contain materially different terms and conditions; comply with current qualifications and training requirements; and sign a general release in a form the same as or similar to the General Release attached to the Franchise Agreement. *Upon renewal, franchisees may be asked to sign a Franchise Agreement with materially different terms and conditions than their original contract/Franchise Agreement.
d. Termination by you	Section 16.1	You may terminate the Franchise Agreement if you are in compliance with it and we materially breach it and we fail to begin to cure our breach within 30 days of receiving your written notice.
e. Termination by us without cause	Not Applicable	
f. Termination by us with cause	Section 16.2	We may terminate the Franchise Agreement only if you default. If we terminate the Franchise Agreement following a default, your interest in the franchise will terminate.

Provision	Section in the Franchise Agreement	Summary
g. "Cause" defined-defaults that can be cured	Section 16.2	If a default arises from your failure to comply with a mandatory specification in the Franchise Agreement or Manual, you can avoid termination of the Franchise Agreement if you cure the default within 30 days of receiving our notice of default. If a default arises from your failure to maintain insurance, you can avoid termination of the Franchise Agreement if you cure the default within 10 days of receiving our notice of your failure to maintain insurance. If a default arises from your failure to make payments due to us, you can avoid termination of the Franchise Agreement if you cure the default within 10 days of receiving our notice of default. If we terminate the Franchise Agreement following a default, your interest in the franchise will terminate.
h. "Cause" defined-defaults that cannot be cured	Section 16.2	We have the right to terminate the Franchise Agreement without giving you an opportunity to cure if you: fail to timely select an approved site for or establish, equip and begin operations of the Franchised Business; fail to have your Designated Manager satisfactorily complete training; made a material misrepresentation or omission in the application for the franchise; are convicted of or plead no contest to a felony or other crime or offense likely to affect the reputation of either party or the Franchised Business; use the Manual, Trade Secrets or other Confidential Information in an unauthorized manner; if required, fail to have your owners, officers, directors, managers, executives, employees and professional staff, and other individuals having access to Trade Secrets or other Confidential Information sign nondisclosure and non-competition agreements; abandon the Franchised Business for 5 or more consecutive days; surrender or transfer of control of the

Provision	Section in the Franchise Agreement	Summary
		Franchised Business in an unauthorized manner; fail to maintain the Franchised Business under the supervision of a Designated Manager following the death or incapacity of one of your owners; are adjudicated bankrupt, insolvent or make a general assignment for the benefit of creditors; misuse or make unauthorized use of the Marks; fail on 3 or more occasions within any 12 months to submit reports or records or to pay any fees due us or any Affiliate; violate any health, safety or other laws or conducts the Franchised Business in a manner creating a health or safety hazard to customers, employees or the public; engage in any activity reserved to us; fail to comply with laws or regulations after notice of non-compliance; repeatedly breach the Franchise Agreement or repeatedly fail to comply with requirements, specifications, standards or procedures; or default under any other agreement between you and us (or our Affiliate) such that we (or our Affiliate) have the right to terminate the agreement.
i. Your obligations on termination/non-renewal	Section 17.1	If the Franchise Agreement is terminated or not renewed, you must: stop operating the Franchised Business; stop using any Trade Secret or other Confidential Information, the System and the Marks; if the Franchised Business Office is not within Franchisee's principal residence and if requested, assign your interest in the Approved Location to us; cancel or assign to us any assumed names; return any leased, loaned or other third-party equipment used in the operation of the Franchised Business; pay all sums owed to us (or our Affiliates) including damages and costs incurred in enforcing the Franchise Agreement; return the Manual, Trade Secrets and all other Confidential Information; assign your telephone and facsimile numbers to us; comply with the covenants not to compete and any other surviving provisions of the Franchise Agreement.
j. Assignment of contract by us	Section 18.1	There are no restrictions on our right to assign our interest in the Franchise Agreement.

Provision	Section in the Franchise Agreement	Summary
k. "Transfer" by you-definition	Section 18.2	"Transfer" includes transfer of an interest in the franchise, the Franchise Agreement, the Franchised Business's assets or the franchisee entity.
I. Our approval of transfer by you	Section 18.2	You may not transfer your interest in any of the items listed in (k) above without our prior written consent.
m. Conditions for our approval of transfer	Section 18.2	We will consent to a transfer if: we have not exercised our right of first refusal; all obligations owed to us are paid; you and the transferee have signed a general release in a form the same as or similar to the General Release attached to the Franchise Agreement; the prospective transferee meets our business and financial standards; the transferee and all persons owning any interest in the transferee sign the then current Franchise Agreement; you provide us with a copy of all contracts and agreements related to the transfer; you or the transferee pay a transfer fee of \$7,500; the transferee or the owners of transferee have agreed to be personally bound by all provisions of the Franchise Agreement; you have agreed to guarantee performance by the transferee, if requested by us; the transferee has obtained all necessary consents and approvals of third parties; you or all of your equity owners have signed a non-competition agreement in a form the same as or similar to the Nondisclosure and Non-Competition attached to the Franchise Agreement; and the transferee has agreed that its Designated Manager will complete the initial training program before assuming management of the Franchised Business.
n. Our right of first refusal to acquire your Franchised Business	Section 19	We may match an offer for your Franchised Business or an ownership interest you propose to sell.
o. Our option to purchase your Franchised Business	Section 17.4	Except as described in (n) above, we do not have the right to purchase your Franchised Business; however, during the 30-day period after the termination or expiration of the Franchise Agreement, we have the right to purchase any assets of the Franchised Business for fair market value.

Provision	Section in the Franchise Agreement	Summary
p. Death or disability	Section 18.6	After a death or incapacity of one of your owners, his or her representative must transfer, subject to the terms of the Franchise Agreement, the owner's interest in the Franchised Business within 180 days of death or incapacity or we may terminate the Franchise Agreement
q. Non-competition covenants during the term of the franchise	Exhibit 2	You, your owners and your officers, directors, executives, managers, professional staff and employees are prohibited from: attempting to divert any business or customer of the Franchised Business to a Competitive Business or causing injury or prejudice to the Marks or the System; owning or working for a Competitive Business. Non-competition provisions are subject to state law.
r. Non-competition covenants after the franchise is terminated or expires	Sections 17.2	For 3 years after the termination or expiration of the Franchise Agreement, you, your owners and your officers, directors, executives, managers or professional staff are prohibited from: owning or working for a Competitive Business operating within 50 miles of the Approved Location (or within the Area of Primary Responsibility, if greater), or within 50 miles of any other The Glass Guru Businesses; or soliciting or influencing any of our customers, employees or business associates to compete with, or terminate their relationship with us or any other The Glass Guru Business. Non-competition provisions are subject to state law.
s. Modification of the agreement	Sections 9.2, 22.7, and 22.8	The Franchise Agreement can be modified only by written agreement between you and us. We may modify the Manual without your consent if the modification does not materially alter your fundamental rights under the Franchise Agreement.

Provision	Section in the Franchise Agreement	Summary
t. Integration/merger clause	Section 22.7	Only the terms of this Franchise Disclosure Document and the Franchise Agreement are binding. Unless described in this Disclosure Document Any other promises may not be enforceable. Nothing in the agreement or in any related agreement is intended to disclaim the representations made in the franchise disclosure document.
u. Dispute resolution by arbitration or mediation	Section 23.7	Except for claims relating to the Marks, Trade Secrets or other Confidential Information and covenants not to compete, and subject to state law, all disputes must be arbitrated in Collin County, Texas.
v. Choice of forum	Section 23.2	Subject to state law, any litigation must be pursued in courts located in Collin County, Texas.
w. Choice of law	Section 23.1	Subject to state law, Texas law applies, except that disputes over the Marks will be governed by the United States Trademark Act of 1946 (Lanham Act, 15 U.S.C. Sec. 1051 et seq.).

You should refer to any state-specific addenda attached to this Disclosure Document for exceptions to this ITEM 17.

ITEM 18. PUBLIC FIGURES

We do not presently use any public figures to promote our franchise.

ITEM 19. FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a Franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a Franchisor provides the actual records of an existing outlet you are considering buying; or (2) a Franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

As of December 31, 2023, The Franchisor had 73 The Glass Guru franchised outlets. Of these, 71 outlets were in business for 12 months or more, and which outlets also had 12 full months of reported Sales data for calendar year 2023. Because the number of franchised outlets fluctuated during the year, we have not included any data for franchised outlets that did not have 12 full months of reported Sales data for 2023. Excluded locations include franchised outlets that started operations in 2023, or ceased operations in 2023, and did not have 12 full periods of reporting submitted to the Franchisor. These specific franchised outlets would likely have Sales figures that differ from those depicted in this report. Franchised outlets defined as "Affiliates" have also been excluded from this data.

For this Item 19, "Sales" means total payments received from whatever source (whether in the form of cash, credit, agreements to pay, barter, trade credits, good will or other consideration) that arise, directly or indirectly, from or in connection with the operation of your franchise, including but not limited to:

- a. the sale of goods and/or services offered by or through the franchise.
- b. the sale of goods and/or services by you or a third-party selling products and/or services on your behalf that are sold or that are required to be sold under the terms of the Franchise Agreement, no matter from what location or business the income is generated;
- c. the proceeds from any business interruption insurance and/or damages or settlement amounts received to compensate you for lost revenue of the franchise; and
- d. any revenue generated from commissions, rebates, or affiliated programs.

All of the franchised outlets for which Sales are reported are operated by franchisees. Franchisor did not operate any of the businesses. All of the locations reflected in the report are comparable to franchised outlets offered in this disclosure document and offered substantially the same services to the public. No adjustments have been made to reflect geographic location or applicable exchange rate. In addition, all of the information in this report was derived from franchise reported data for a full FIFTY-TWO (52) week period covering the 2023 calendar year.

CHART 1 ANNUAL SALES LEVELS

The chart below shows Annual **Sales** figures derived from payments received reported by all qualifying franchised outlets during 2023 and 2022, respectively. These figures are separated into groups that include Average (mean) Sales, Middle (median) Sales, Highest Sales, Lowest Sales, and percentiles that include the 90th Percentile, 75th Percentile, 50th Percentile, 25th Percentile, and 10th Percentile. The chart also reflects the number of franchised outlets that exceeded the Annual Sales value for each group, as well as the Annual Sales variance between 2023 and 2022 for each group.

Measure	2023 Annual Sales	Franchised Outlets Exceeding Value	2022 Annual Sales	Franchised Outlets Exceeding Value	2023 vs. 2022	Explanation
Sales Average Mean	\$700,345	31	\$683,530	26	+ \$16,816	Shows total of reported annual sales \$ (received payments) for all qualifying franchise outlets, divided by the total number (count) of all qualifying franchise outlets. A total of 31 franchised outlets exceeded this percentile value in 2023 and 26 in 2022 respectively.
Sales Middle Median	\$638,114	36	\$554,956	36	+ \$83,158	Shows mid-point of reported annual sales \$ (received payments) by all qualifying franchise outlets. A total of 36 franchised outlets exceeded this percentile value in 2023 and 36 in 2022 respectively.
Sales Highest Value	\$2,122,434	1	\$2,357,330	1	- \$234,895	Shows highest reported annual sales \$ (received payments) for all qualifying franchise outlets.
Sales Lowest Value	\$37,643	71	\$61,228	72	- \$23,585	Shows lowest reported annual sales \$ (received payments) for all qualifying franchise outlets.
Sales 90th Percentile Top 10%	\$1,910,192	2	\$2,121,598	1	- \$211,406	Shows 90th percentile (top 10%) of reported annual sales \$ (received payments) by all qualifying franchise outlets; only 10% of franchise outlets reported higher sales than this. A total of 2 franchised outlets exceeded this value in 2023 and 1 in 2022 respectively.
Sales 75th Percentile Top 25%	\$1,591,826	5	\$1,767,998	4	- \$176,172	Shows 75th percentile (top 25%) of reported annual sales \$ (received payments) by all qualifying franchise outlets; only 25% of franchise outlets reported higher sales than this. A total of 5 franchised outlets exceeded this value in 2023 and 4 in 2022 respectively.
Sales 50th Percentile Top 50%	\$1,061,218	12	\$1,178,665	11	- \$117,448	Shows 50th percentile (top 50%) of reported annual sales \$ (received payments) by all qualifying franchise

Measure	2023 Annual Sales	Franchised Outlets Exceeding Value	2022 Annual Sales	Franchised Outlets Exceeding Value	2023 vs. 2022	Explanation
						outlets; only 50% of franchise outlets reported higher sales than this. A total of 12 franchised outlets exceeded this percentile value in 2023 and 11 in 2022 respectively.
Sales 25th Percentile Bottom 25%	\$530,609	40	\$589,333	34	- \$58,724	Shows 25th percentile (bottom 25%) of reported annual sales \$ (received payments) by all qualifying franchise outlets; only 75% of franchise outlets reported higher sales than this. A total of 40franchised outlets exceeded this value in 2023 and 34 in 2022 respectively.
Sales 10th Percentile Bottom 10%	\$212,244	65	\$235,733	64	- \$23,490	Shows 10th percentile (bottom 10%) of reported annual sales \$ (received payments) by all qualifying franchise outlets; only 90% of franchise outlets reported higher sales than this. A total of 65 franchised outlets exceeded this percentile value in 2023 and 64 in 2022 respectively.

CHART 2 ANNUAL SALES & POPULATION BY LOCATION

The Chart below reflects the Annual Sales reported and the corresponding territory population for each of the 71 Franchised Outlets operating for 12 months or more, and which outlets also had 12 full months of reported Gross Sales for calendar year 2023. Data is ranked by the highest to lowest Annual Sales as reported to us in information completed by our Franchisees. Markers Average (mean) Sales, Middle (median) Sales, Highest Sales, Lowest Sales, as well as percentiles including the 90th Percentile, 75th Percentile, 50th Percentile, 25th Percentile, and 10th Percentile are also included. The populations used in this chart were generated using 2023 HERE USA map data and include only those zip/postal codes in the Area of Primary Responsibility as defined in the Franchise Agreement.

Percentile / Marker	2023 Annual Sales	2023 Population
	\$2,122,434	1,357,689
Sales Highest Payments	\$2,122,434	
	\$2,042,440	1,015,148
Sales 90th Percentile Payments	\$1,910,192	
	\$1,648,612	739,661
	\$1,633,714	687,653
	\$1,595,527	176,756
Sales 75th Percentile Payments	\$1,591,826	
	\$1,535,485	817,700
	\$1,484,526	698,130
	\$1,446,831	947,009
	\$1,326,508	270,823
	\$1,276,367	550,192
	\$1,220,902	501,037
	\$1,168,848	525,771
Sales 50th Percentile Payments	\$1,061,218	
	\$1,007,394	585,982
	\$1,001,100	569,940
	\$990,069	578,779
	\$946,636	387,864
	\$894,550	654,482
	\$884,298	233,301
	\$864,345	565,355
	\$840,683	188,271
	\$821,044	579,173

	\$813,956	502,198
	\$797,980	359,920
	\$787,249	316,234
	\$783,307	445,914
	\$769,273	684,750
	\$760,542	974,602
	\$739,455	501,080
	\$713,647	392,770
	\$711,898	513,338
	\$705,063	456,157
Sales Average / Mean Payments	\$700,345	
	\$695,743	645,681
	\$695,205	831,300
	\$688,186	874,988
	\$683,386	892,760
	\$638,114	288,568
Sales Middle / Median Payments	\$638,114	
	\$629,847	416,266
	\$606,108	844,892
	\$571,838	673,900
	\$542,353	401,577
Sales 25th Percentile Payments	\$530,609	
	\$529,706	460,058
	\$498,503	483,541
	\$496,577	200,002
	\$477,336	514,938
	\$465,825	309,636
	\$464,600	281,715
	\$415,733	381,097
	\$401,466	494,310
	\$395,576	528,555
	\$384,487	631,078
	\$358,417	826,449
	\$346,740	817,141
	\$332,294	628,533

\$328,910	374,998
\$325,562	163,419
\$325,198	288,296
\$320,125	554,757
\$317,145	419,636
\$293,027	670,839
\$290,699	511,437
\$281,651	247,433
\$265,843	544,265
\$258,721	710,594
\$258,298	478,935
\$230,874	256,871
\$212,244	
\$162,265	423,991
\$117,445	180,107
\$116,842	147,361
\$73,590	309,931
\$67,955	223,995
\$37,643	
\$37,643	110,052
	\$325,562 \$325,198 \$320,125 \$317,145 \$293,027 \$290,699 \$281,651 \$265,843 \$258,721 \$258,298 \$230,874 \$212,244 \$162,265 \$117,445 \$116,842 \$73,590 \$67,955 \$37,643

CHART 3 AVERAGE JOB SIZE (INVOICED)

The chart below shows Average Job Size (AJS) based on <u>pre-tax invoice amounts</u> derived from invoices reported by all qualifying franchised outlets during 2023 and 2022, respectively. These figures are separated into groups that include Average (mean) Sales, Middle (median) Sales, Highest AJS, Lowest AJS, and percentiles that include the 90th Percentile, 75th Percentile, 50th Percentile, 25th Percentile, and 10th Percentile. The chart also reflects the number of franchised outlets that exceeded the Average Job Size value for each group, as well as the Average Job Size variance between 2023 and 2022 for each group.

Measure	2023 Average Job Size (AJS)	Franchised Outlets Exceeding Value	2022 Average Job Size (AJS)	Franchised Outlets Exceeding Value	2023 vs. 2022	Explanation
AJS Average Mean	\$1,108	30	\$1,104	29	+ \$04	Shows reported Average Job Size (AJS) for all qualifying franchised outlets. The value is derived from the total invoiced sales, divided by the total number (count) of all invoices. A total of 30 franchised outlets exceeded this value in 2023 and 29 in 2022 respectively.
AJS Middle Median	\$1,034	35	\$801	33	+ \$233	Shows mid-point of reported Average Job Size (AJS) by all qualifying franchised outlets. A total of 35 franchised outlets exceeded this value in 2023 and 33 in 2022 respectively.
AJS Highest Value	\$2,684	1	\$2,236	1	+ \$448	Shows highest reported Average Job Size (AJS) for all qualifying franchised outlets.
AJS Lowest Value	\$400	69	\$531	65	- \$131	Shows lowest reported Average Job Size (AJS) for all qualifying franchised outlets.
AJS 90th Percentile Top 10%	\$2,417	2	\$2,013	1	+ \$404	Shows 90th percentile (top 10%) of reported Average Job Size (AJS) by all qualifying franchised outlets; only 10% of franchised outlets reported higher sales than this. A total of 2 franchised outlets exceeded this value in 2023 and 1 in 2022 respectively.
AJS 75th Percentile Top 25%	\$2,014	3	\$1,678	3	+ \$336	Shows 75th percentile (top 25%) of reported Average Job Size (AJS) by all qualifying Franchise Businesses; only 25% of franchise businesses reported higher sales than this. A total of 3 units exceeded this value in 2023 and 3 in 2022 respectively.
AJS 50th Percentile Top 50%	\$1,343	14	\$1,119	11	+ \$224	Shows 50th percentile (top 50%) of reported Average Job Size (AJS) by all qualifying Franchise Businesses; only 50% of franchise businesses reported higher sales than this. A total of 14 units exceeded this value in 2023 and 11 in 2022 respectively.
AJS 25th Percentile Bottom 25%	\$671	63	\$559	59	+ \$112	Shows 25th percentile (bottom 25%) of reported Average Job Size (AJS) by all qualifying franchised outlets; only 75% of franchised outlets reported higher sales than this. A total of 63 franchised outlets exceeded this value in 2023 and 59 in 2022 respectively.

AJS 10th Percentile Bottom 10%	\$269	69	\$224	65	+ \$45	Shows 10th percentile (bottom 10%) of reported Average Job Size (AJS) by all qualifying franchised outlets; only 90% of franchised outlets reported higher sales than this. A total of 69 franchised outlets exceeded this value in 2023 and 65 in 2022 respectively.
----------------------------------------	-------	----	-------	----	--------	-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------

The historical information listed in this item includes the Sales and Average Job Size derived from historical operating results of the Franchise Businesses indicated for the period of time reflected in this report. Sales is defined as any and all monies received & reported by Franchise Businesses for goods and services sold under the franchise brand during the reporting period specified. Average Job Size is derived from the pre-tax invoice amounts for Invoices generated and reported by all Franchised Businesses by the reporting period specified.

Franchisor obtained these figures from information provided by Franchisees for the period of time January 1, 2022, through December 31, 2023.

Neither we, or our independent Certified Public Accountants, have audited or verified any of the figures reported to us. We make no representations as to the accuracy of gross revenues reported by our franchisees or the extent that these figures were derived using generally accepted accounting principles (GAAP). These are based upon reasonable assumption of accuracy due to daily submission and monitoring of financial performance data from each franchisee.

Some outlets have sold this amount. Your individual results may differ. There is no assurance that you will sell as much.

Written substantiation for these financial performance representations will be made available for a prospective franchisee upon request.

Other than the preceding financial performance representation, The Glass Guru Enterprises, Inc. does not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projection of your future income, you should report it to the franchisor's management by contacting Mr. Daniel Frey, President, 5550 Granite Parkway, Suite 280, Plano, TX 75024, (877) 654-8507, the Federal Trade Commission, and the appropriate state regulatory agencies.

ITEM 20. OUTLETS AND FRANCHISEE INFORMATION

Item 20 : Table No. 1 System-Wide Outlet Summary for Years 2021-2023

	ı		1	
Outlet Type	Year	Outlets at START of year	Outlets at END of Year	Net Change
	2021	71	77	+6
USA Franchised	2022	77	77	0
	2023	77	72	-5
	2021	1	1	0
Affiliate-Owned	2022	1	1	0
	2023	1	1	0
	2021	72	78	+6
Total Outlets	2022	78	78	0
	2023	78	73	-5

Item 20: Table No. 2 Transfers of Outlets to New Owners (other than the Franchisor) for Years 2021-2023

State	Year	Number of Transfers
	2021	0
California	2022	0
	2023	1
	2021	1
Colorado	2022	1
	2023	1
	2021	1
Totals	2022	1
	2023	2

Item 20 : Table No. 3 Status of Outlets Franchised for Years 2021-2023

State	Year	Outlets Year Start	Outlets Opened	Terminated	Non- Renewals	Re-acquired by Franchisor	Ceased Operation Other Reasons	Outlets at End of the Year
	2021	2	0	0	0	0	0	2
Alabama	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	2021	11	1	0	1	0	0	11
California	2022	11	0	0	0	0	1	10
	2023	10	1	1	0	0	0	10
	2021	5	2	1	0	0	0	6
Colorado	2022	6	3	1	0	0	1	7
	2023	7	1	3	0	0	0	5
	2021	5	1	0	0	0	0	6
Florida	2022	6	0	1	0	0	1	4
	2023	4	1	0	0	0	1	4
Georgia	2021	3	0	0	0	0	0	3
	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	1	2
Hawaii	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	1	0	0	0	0	0	1
Illinois	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	2	0	0	0	0	0	2
Indiana	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	2021	2	0	0	0	0	0	2
Kansas	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	2021	1	0	0	0	0	0	1
Louisiana	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	1	0	0	0	0	1	0
Maryland	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0

	2021	3	1	1	0	0	0	3
Michigan	2022	3	0	1	0	0	0	2
	2023	2	0	0	0	0	0	2
	2021	2	1	0	0	0	0	3
Minnesota	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	0	3
	2021	2	0	0	0	0	0	2
Missouri	2022	2	0	0	0	0	0	2
	2023	2	0	0	0	0	0	2
	2021	1	0	0	0	0	0	1
New Jersey	2022	1	0	0	0	0	0	1
,	2023	1	0	0	0	0	0	1
	2021	1	0	0	0	0	0	1
New York	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	1	0	0	0	0	0	1
North	2022	1	0	0	0	0	1	0
Carolina	2023	0	0	0	0	0	0	0
Ohio	2021	6	0	0	0	0	0	6
	2022	6	0	0	0	0	0	6
	2023	6	0	0	0	0	0	6
Oklahoma	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	0	0	0	0	0	0	0
Oregon	2022	0	1	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	1	0	0	0	0	0	1
Pennsylvania	2022	1	0	0	0	0	0	1
	2023	1	1	1	0	0	0	1
	2021	1	1	0	0	0	0	2
	2022	2	0	0	0	0	0	2
South Carolina	2023	2	0	0	0	0	0	2
	2021	5	0	0	0	0	0	5
Tennessee	2022	5	0	0	0	0	0	5
	2023	5	0	0	0	0	1	4
	2021	10	3	0	0	0	1	12
Texas	2022	12	3	0	0	0	0	15
	2023	15	0	0	0	0	1	14
	2021	2	1	0	0	0	0	3
Washington	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	0	3

	2021	1	0	0	0	0	0	1
Wisconsin	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	71	11	2	1	0	2	77
USA	2022	77	7	3	0	0	4	77
	2023	77	4	5	0	0	4	72
	2021	5	0	0	0	0	0	5
Canada	2022	5	0	0	0	0	0	5
	2023	5	0	0	0	0	0	5
	2021	76	11	2	1	0	2	82
Totals	2022	82	7	3	0	0	4	82
	2023	82	4	5	0	0	4	77

Item 20 : Table No. 4 Status of Affiliate-Owned Outlets for Years 2021-2023

State	Year	Outlets Year Start	Outlets Opened	Terminated	Non- Renewals	Re-acquired by Franchisor	Ceased Operation Other Reasons	Outlets at End of the Year
	2021	1	0	0	0	0	0	1
California	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2021	1	0	0	0	0	0	1
Totals	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1

Item 20 : Table No. 5 Projected Openings as of 12/31/24

State	Franchise Agreements Signed But Outlet Not Opened	Projected New Franchised Outlet In The Next Fiscal Year	Projected New Company Owned Outlet In the Next Fiscal Year
Arizona	0	1	0
California	0	2	0
Florida	0	2	0
Georgia	0	1	0
Illinois	0	1	0
Missouri	0	1	0
North Carolina	0	1	0
Ohio	0	1	0
Texas	0	2	0
TOTALS	0	12	0

ITEM 21. FINANCIAL STATEMENTS

The audited financial statements of The Glass Guru Enterprises, Inc. for the years ended December 31, 2021, December 31, 2022, and December 31, 2023, are included in Exhibit E to this Disclosure Document.

ITEM 22. CONTRACTS

The Glass Guru Enterprises, Inc. Franchise Agreement (with exhibits) is attached to this Disclosure Document as Exhibit A.

The Glass Guru Enterprises, Inc. Specifics is attached to the Franchise Agreement as Exhibit 1.

The Glass Guru Enterprises, Inc. General Release is attached to the Franchise Agreement as Exhibit 2.

The Glass Guru Enterprises, Inc. Nondisclosure and Non-Competition Agreement is attached to the Franchise Agreement as Exhibit 3.

The Glass Guru Enterprises, Inc. Guaranty and Assumption of Obligations is attached to the Franchise Agreement as Exhibit 4.

The Glass Guru Enterprises, Inc. Holder of Legal or Beneficial Interest in Franchisee; Officers; Directors is attached to the Franchise Agreement as Exhibit 5.

The Glass Guru Enterprises, Inc. Assignment of Telephone Numbers and Internet Tools is attached to the Franchise Agreement as Exhibit 6.

The Glass Guru Enterprises, Inc. Lease Addendum and Collateral Assignment is attached to the Franchise Agreement as Exhibit 7.

The Glass Guru Enterprises, Inc. Promissory Note and Security Agreement that you must sign if we finance a portion of the initial franchise fee is attached to this Disclosure Document as Exhibit I.

We provide no other contracts or agreements for your signature.

ITEM 23. RECEIPTS

Our copy and your copy of the Franchise Disclosure Document Receipt are located on the last 2 pages of this Disclosure Document.

EXHIBIT A

TO THE DISCLOSURE DOCUMENT THE GLASS GURU ENTERPRISES, INC. FRANCHISE AGREEMENT



FRANCHISE AGREEMENT

THE GLASS GURU ENTERPRISES, INC EXHIBIT A TO THE DISCLOSURE DOCUMENT

CONTENTS

1	DEFINITIONS	9
	GRANT OF FRANCHISE	
2.1	Grant	
2.2	Approved Location	
2.3	Area of Primary Responsibility	
2.4	Nonexclusive License	
2.5	Marketing, Solicitation and Service Restrictions for Areas of Primary Responsibility	
2.6	Sub-franchising/Agents	
2.7	Notice to Franchisor for Certain Events	
2.8	Right to Obtain Franchisee's Credit Report/Background Checks	
2.9	Supplemental Exhibits and Agreements	16
	FEES, OTHER PAYMENTS AND PROCEDURES	
3.1	Franchise Fee	
3.2	Monthly Royalty Fee	
3.3	Marketing Fee	
3.4	Non-APR Service Fee	
3.5	Technology Fee	19
3.6	Taxes	19
3.7	Payment of Fees	19
3.8	Late Fees	20
3.9	NSF Fees and Interest	20
3.10	Application of Payments	20
	TERM AND RENEWAL	
4.1	Initial Term	
4.2	Successor Terms	
4.3	Good Cause for Withholding Approval of Renewal	
4.4	Right to Change Area of Primary Responsibility on Renewal of Franchise	23
4.5	Continued Operation Following Expiration	23
	ESTABLISHING THE FRANCHISE BUSINESS	
5.1	Selection of Site	
5.2	Failure to Select Site	24

5.3	Lease of Approved Location	24
5.4	Development of Franchise Business and Vehicles	25
5.5	Franchisee Entity Requirements.	25
5.6	Opening	26
5.7	Failure to Open	27
5.8	Use of Approved Location	27
5.9	Relocation of the Franchise Business	27
5.10	Employee Non-disclosure and Confidentiality Agreements	28
5.11	Minimum Performance Requirements	28
5.12	Franchisor's Remedies Relating to Minimum Performance Requirements	29
	PROPRIETARY MARKS	
6.1	Ownership	
6.2	Limitations on Use	
6.3	Manner of Using Marks	30
6.4	Goodwill	
6.5	Permitted Business Name	
6.6	Notification of Infringements and Claims	31
6.7	Indemnification for Use of Marks	32
6.8	Discontinuance of Use	32
6.9	Right to Inspect	32
6.10	Franchisor's Sole Right to Domain Name	33
	CONFIDENTIAL INFORMATION	
	Confidentiality	33
7.2	Additional Developments	
7.3	Exclusive Relationship	
7.4	Nondisclosure and Non-Competition Agreements with Certain Individuals	
7.5	Reasonableness of Restrictions	35
	FRAINING AND ASSISTANCE	
8.1	Initial Training	
8.2	Technician Training & Certification Programs	
8.3	Opening Assistance	
8.4	Failure to Complete Initial Training Program	
8.5	New Designated Manager	
8.6	Ongoing Training	3 /

8.7	Annual Franchisee Convention	37
8.8	Training and Performance Improvement Requirements	37
9 N	/ANUALS	38
9.1	Definition of Manuals	
9.2	Obligation to Comply with All Company Policies and Procedures; Confidentiality of	38
9.3	Revisions	39
9.4	Confidentiality	39
10 F	RANCHISE SYSTEM	39
10.1	Uniformity	39
10.2	Modification of the System	39
10.3	Variance	40
11 A	DVERTISING AND PROMOTIONAL ACTIVITIES	40
11.1	Brand Identity Package	40
11.2	Local Advertising	40
11.3	Internet Advertising	41
11.4	Telemarketing	42
11.5	Required Advertising on Glass Trucks and Other Business Vehicles	42
11.6	General Limitations on Advertising and Marketing	42
11.7	Present and Future Advertising Policy May Limit Franchisee's Right to Advertise	42
11.8	Agreement to Comply with Franchisor Policy	43
11.9	Agreement to Comply with Modifications and Changes to Franchisor Policy	43
11.10	No False Advertising	43
11.11	Advertisements Produced by 3 rd Parties	43
11.12	Franchisor Remedies for Displaying Unapproved or Unauthorized Advertising	43
12 A	CCOUNTING RECORDS AND REPORTING OBLIGATIONS	44
12.1	Records	
12.2	Monthly Reports	44
12.3	Financial Statements	45
12.4	Other Reports	46
12.5	Computer, Point-of-Sale, and Accounting Systems	46
12.6	Right to Inspect	47
12.7	Release of Records	47
12.8	Chart of Accounts; Central Accounting Office	47

13 S	TANDARDS OF OPERATION	.47
13.1	Authorized Products, Services and Suppliers	. 47
13.2	The Glass Guru Restoration Tools and Products	. 49
13.3	Appearance and Condition of the Franchise Business	. 49
13.4	Ownership and Management	. 50
13.5	Obligation to Personally Supervise and Manage Day-to-Day Operations	. 50
13.6	Days of Operation	. 50
13.7	Franchise Business Staff	. 50
13.8	Only Franchisee Has the Right to Control Employees	. 50
13.9	Contributions and Donations	. 51
13.10	Licenses and Permits	. 51
13.11	Notification of Proceedings	. 51
13.12	Compliance with Good Business Practices	. 52
13.13 Laws a	Obligation to be Governed by Highest Ethical Standards and to Comply with and Regulations	
13.14	Disputes Arising with Third Parties	. 52
13.15	Uniforms	. 53
13.16	Credit Cards	. 53
13.17	E-Mail	. 54
13.18	Participation in Call Center	. 54
13.19	Best Efforts	. 54
13.20	Automation Systems	. 54
13.21	Proprietary Computer Systems	. 55
13.22	Franchisor Access to and Use of Information	. 55
13.23	Certain Telephone Technology Prohibited	. 56
14 F	RANCHISOR'S ADDITIONAL OPERATIONS ASSISTANCE	.56
14.1	General Advice and Guidance	
14.2	Periodic Visits	
14.3	System Improvements	
14.4	Marketing and Promotional Materials	. 57
15 II	NSURANCE	
15.1	Types and Amounts of Coverage	
15.2	Future Increases	
15.3	Carrier Standards	
15.4	Evidence of Coverage	. 58

15.5	Failure to Maintain Coverage	58
16 [DEFAULT AND TERMINATION	59
16.1	Termination by Franchisee	59
16.2	Termination by Franchisor	59
16.3	Reinstatement and Extension	
16.4	Right of Franchisor to Discontinue Services to Franchisee	62
16.5	Right of Franchisor to Operate Franchise Business	62
17 F	RIGHTS AND DUTIES UPON TERMINATION	63
17.1	Actions to be Taken	63
17.2	Post-Termination Covenant Not to Compete	
17.3	Unfair Competition	
17.4	Franchisor's Option to Purchase Certain Business Assets	
17.5	Survival of Certain Provisions	67
18 1	TRANSFERABILITY OF INTEREST	
18.1	Transfer by Franchisor	67
18.2	Transfer by Franchisee to a Third Party	67
18.3	Franchisor's Disclosure to Transferee	69
18.4	Franchisee Disclosure to Transferee	69
18.5	For-Sale Advertising	69
18.6	Transfer by Death or Incapacity	70
19 F	RIGHT OF FIRST REFUSAL	70
19.1	Submission of Offer	70
19.2	Franchisor's Right to Purchase	
19.3	Non-Exercise of Right of First Refusal	71
20 E	BENEFICIAL OWNERS OF FRANCHISE	71
21 F	RELATIONSHIP AND INDEMNIFICATION	72
21.1	Relationship	72
21.2	Standard of Care	72
21.3	No Employment Relationship	72
21.4	Indemnification	
21.5	Right to Retain Counsel	73
22 (GENERAL CONDITIONS AND PROVISIONS	74
22.1	No Waiver	74
22.2	Injunctive Relief	74

22.3	Notices	74
22.4	Cost of Enforcement or Defense	75
22.5	Guaranty and Assumption of Obligations	75
22.6	Approvals	75
22.7	Entire Agreement	75
22.8	Severability and Modification	75
22.9	Construction	76
22.10	Force Majeure	76
22.11	Timing	76
22.12	Withholding Payments	76
22.13	Further Assurances	77
22.14	Third-Party Beneficiaries	77
22.15	Multiple Originals; Electronic Signature	77
23 C	DISPUTE RESOLUTION	77
23.1	Choice of Law	77
23.2	Consent to Jurisdiction	77
23.3	Cumulative Rights and Remedies	78
23.4	Limitations of Claims	78
23.5	Limitation of Damages	78
23.6	Waiver of Jury Trial	78
23.7	Arbitration	78
24 A	AFFIRMATIONS	79
24.1	Receipt of this Agreement and the Franchise Disclosure Document	79
24.2	True and Accurate Information	79
24.3	No Violation of Other Agreements	79

THE GLASS GURU ENTERPRISES, INC.

FRANCHISE AGREEMENT

This Franchise Agreement, made	_, is by and between The Glass
Guru Enterprises, Inc., a Texas corporation, having its principal pla	ace of business at 5550 Granite
Parkway., Suite 280, Plano TX 75024 ("Franchisor"), and the indi	vidual or entity ("Franchisee"),
with the names and addresses for each party listed on Exhibit 1.	

WITNESSETH:

WHEREAS, Franchisor and its Affiliate* have developed, and are continuing to develop, a System identified by the service mark "THE GLASS GURU" relating to the establishment and operation of a franchise business to install, repair and replace residential and commercial glass, windows and doors and provide other related services and sell other related products to residential and commercial customers, which is referred to in this Agreement as "The Glass Guru Business;" and

WHEREAS, Franchisor and its Affiliate have custom designed, developed and/or acquired, and are in theprocess of continuing to design and develop, certain window restoration tools and equipment, and

products ("The Glass Guru Restoration Tools") especially suited for use in the operation of The GlassGuru Business, which are or will be supplied by Franchisor to franchisees on a for-profit basis; and

WHEREAS, in addition to The Glass Guru Restoration Tools and the service mark "THE GLASS GURU" and other Marks, the distinguishing characteristics of the System include, among other things, uniform standards, specifications, methods, and procedures for business operations; procedures and strategies for marketing, advertising and promotion of The Glass Guru Business; customer service and development techniques; other strategies, techniques, Trade Secrets and Confidential Information, and the Manuals (the System)

WHEREAS, Franchisor intends to grant Franchisee the right to own and operate a Franchise Businessusing the System, the Marks and other resources described in this Agreement; and

WHEREAS, Franchisee desires to operate The Glass Guru Business, has applied to Franchisor for the rightto operate a Franchise and such application has been approved by Franchisor in reliance upon all the representations of Franchisee made in the Agreement, including exhibits; and,

WHEREAS, Franchisee understands and acknowledges the importance of Franchisor's standards of highand uniform quality, operations and service and the necessity of operating the Franchise Business in strict conformity with Franchisor's System.

NOW, THEREFORE, Franchisor and Franchisee, intending to be legally bound, agree as follows:

^{*}Capitalized terms not otherwise defined are defined in Section 1.

1 DEFINITIONS

Whenever used in this Agreement, the following words and terms have the following meanings:

"**Action**" includes complaints, cross claims, counterclaims, and third-party complaints in a judicial action or proceeding, and their equivalents in an administrative action or arbitration;

"Affiliate" when used in reference to Franchisee includes, but is not limited to, all persons who are related to the Franchisee (including but not limited to Franchisee's spouse), principals, owners, officers, directors, shareholders, members of limited liability companies, partners, employees, guarantors, parent companies, sister companies and subsidiaries. An entity is affiliated with Franchisee when the entity controls, are controlled by or under common control with Franchisee or any other affiliate as

defined in this Section. The term "affiliate" or "affiliates" when used in reference to Franchisor means any entity that controls, is controlled by or under common control with Franchisor;

"**Agreement**" means this agreement entitled "The Glass Guru Enterprises, Inc. Franchise Agreement" and all instruments supplemental to this Agreement or in amendment or confirmation of this Agreement;

"**Approved Location**" means the site selected by Franchisee and approved in writing by Franchisor from which Franchisee manages and administers the Franchise Business and maintains the books and records of the Franchise Business;

"Approved Supplier" has the meaning given to such term in Section 13.1;

"Area of Primary Responsibility" has the meaning given to such term means a geographic territoryfor which Franchisor Grants to Franchisee a license to operate under the terms and conditions set forth in this Agreement in Section 2.3 and Section B of Exhibit 1;

"Competing Business" means any business, whether direct or indirect, that offers (or grants franchises or licenses to others to operate a business that offers) products and services the same as or similar to those provided by Franchise Business or in which Trade Secrets or other Confidential Information could be used to the disadvantage of Franchisor, any Affiliate or its other franchisees; provided, however, that the term "Competitive Business" must not apply to (a) any business operated by Franchisee under a Franchise Agreement with Franchisor, or (b) any business operated by a publicly-held entity in which Franchisee owns less than a five percent (5%) legal or beneficial interest;

"Confidential Information" means technical and non-technical information used in or related to TheGlass Guru Business and not commonly known by or available to the public, including, without limitation, Trade Secrets and any other information identified as confidential when delivered by Franchisor. Confidential Information must not include, however, any information that: (a) is now or subsequently becomes generally available to the public through no fault of Franchisee; (b) Franchiseecan demonstrate was rightfully in its possession, without obligation of nondisclosure, prior to disclosure under this Agreement; (c) is independently developed without the use of any Confidential Information; or (d) is rightfully obtained from a third party who has the right, without obligation of nondisclosure, to transfer or disclose such information;

"Converted Business" means a Competitive Business that Franchisor, at its sole discretion, grants an Area of Primary Responsibility in which to begin the Franchise Business. In these circumstances, Franchisee is required to merge the Competitive Business with <u>and</u> into the Franchise Business. All subsequent operations and activities of the Franchise Business will be conducted by the Franchisee under the Mark(s) of The Glass Guru, utilizing appropriate copyrighted materials, manuals, systems, etc. subject to the terms and conditions of this Agreement. While Franchisor will discuss a business conversion with any Competitive Business, a Competitive Business is only eligible receive "Incentives for Converted Businesses" when the Competitive Business converting to a Franchise Business can demonstrate Gross Sales equal to, or greater than \$500,000 for the most current 12-month operatingperiod.

"Designated Manager" means the person designated by Franchisee who has primary responsibility for managing the day-to-day affairs of the Franchise Business, and if Franchisee is an individual and not a business entity, the Designated Manager may be Franchisee;

"**Effective Date**" means the date on which Franchisee and Franchisor fully execute this Agreement, commencing its effectiveness and term;

"Electronic Transfer Account" means an account established at a national banking institution approved by Franchisor and providing Franchisor with access to electronically withdraw or deposit anyfunds due to or from Franchisor;

"Financial Performance Representation" means any representation, including any oral, written or visual representation, to a prospective franchisee, including a representation in the general media, that states, expressly or by implication, a specific level or range of actual or potential sales, income, gross profits, or net profits. The term includes a chart, table, or mathematical calculation that shows possible results based on a combination of variables;

"**Franchise**" means the right granted to Franchisee by Franchisor to use the System and the Marks:

"Franchise Fee" means the amount specified in Section D of Exhibit 1

"Franchise Business" means The Glass Guru business established and operated by Franchisee under

this Agreement, utilizing our System, in one or more stores within an Area of Primary Responsibility.

"Franchisee" means the individual or entity specified as "Franchisee" in in Exhibit 1 "Franchisor" means The Glass Guru Enterprises, Inc.;

"Franchisor Indemnitees" has the meaning given to such term in Section 21.3; "Brand Identity Package" has the meaning given to such term in Section 11.1;

"Gross Sales" means the total gross revenues collected or received from whatever source (whether in the form of cash, credit, agreements to pay, barter, trade credits, good will, or other consideration) that arise, directly or indirectly, from or in connection with the operation of your

franchise, including but not limited to:

- A the sale of goods and/or services offered by or through the franchise;
- B. the sale of goods and/or services by you or a third-party selling products and/or services on your behalf that are sold or that are required to be sold under the terms of the Franchise Agreement, no matter from what location or business the income isgenerated;
- C the proceeds from any business interruption insurance and/or damages or settlement amounts received to compensate you for lost revenue of the franchise;
- D. any revenue generated from commissions, rebates, or Affiliate programs; and,
- E except as provided below for charity activities, the value of any goods and/orservices provided without compensation to you.

You are not required to pay royalty on the value of any portion of goods and/or services provided without compensation to you if those goods and/or services are provided for a charitable organizationapproved by us ("Guru Charity"). Currently, all charitable organizations listed by the Internal Revenue Service ("IRS") as exempt under Internal Revenue Code §501(c)(3) ("501(c)(3) organizations") are automatically approved by us. Go to IRS.gov to search for a list of 501(c)(3) organizations. If you provide goods and/or services without compensation that are not for an organization that is listed by the IRS as a 501(c)(3) organization or a charitable organization otherwise approved by us, the value of those goods and/or services must be included in your Gross Sales for purposes of paying royalty to us. The value of the goods and/or services provided without compensation will be equal to the amount that you would normally charge for the goods and/or services.

"Gross Sales Reports" has the meaning given to such term in Section 12.2;

"**Incapacity**" means the inability of Franchisee to operate or oversee the operation of the FranchiseBusiness on a regular basis by reason of any continuing physical, mental or emotional condition, chemical dependency or other limitation;

"Immediate Family Members" means a person's parents, spouse, children (by blood, adoption, or marriage), and siblings; and, will also include the parent's spouse and any grandchildren.

"Internet" means any one (1) or more local or global interactive communications media that is now available, or that may become available, including sites and domain names on the World Wide Web;

"Local Advertising" has the meaning given to such term in Section 11.2;

"Manual" means The Glass Guru Operations Manual and Safety Manual, whether in paper or electronic form, and any other items as may be provided, added to, changed, or otherwise revised by Franchisor from time to time that contain or describe the standards, methods, procedures and specifications of the System, including other operations, administration and managers' manuals

and all books, computer programs, password-protected portions of an Internet site, pamphlets, and other publications prepared by, or on behalf of, Franchisor.

"Marketing Fee" has the meaning given to such term in Section 3.4;

"Marks" means the service mark "THE GLASS GURU" and such other trade names, trademarks, servicemarks, trade dress, designs, graphics, logos, emblems, insignia, fascia, slogans, drawings and other commercial symbols as Franchisor may designate to be used in connection with The Glass Guru Business:

"Non-APR Service Fee" has the meaning given to such term in Section 3.4;

"Parent" means an entity that controls another entity directly, or indirectly through one or more subsidiaries;

"**Person**" means any individual, group, association, limited or general partnership, corporation, or any other entity;

"Plain English" means the organization of information and language usage understandable by a person unfamiliar with the Franchise Business. It incorporates short sentences; definite, concrete, everyday language; active voice; and tabular presentation of information, where possible. It avoidslegal jargon, highly technical business terms, and multiple negatives;

"**Predecessor**" means a person from whom the franchisor acquired, directly or indirectly, the majorportion of the franchisor's assets;

"Qualified Training Location" means A Qualified Training Location (QTL) is any location approved forthe initial or ongoing training provided to franchise owners or franchise staff. A QTL is typically a Franchisor facility, one or more individual franchise locations, or may be any facility or location definedby the Franchisor;

"Required Payment" means all consideration that the franchisee must pay to the franchisor or an Affiliate, either by contract or by practical necessity, as a condition of obtaining or commencing operation of the franchise. A required payment does not include payments for the purchase of reasonable amounts of inventory at bona fide wholesale prices for resale or lease;

"Royalty Fee" has the meaning given to such term in Section 3.3;

"**System**" means the uniform standards, methods, procedures and specifications developed by Franchisor and as may be added to, changed, modified, withdrawn or otherwise revised by Franchisor for the operation of The Glass Guru Business;

"**The Glass Guru Restoration Tools**" has the meaning given to such term in the Recitals precedingthis Section;

"The Glass Guru Business" refers to the Franchise Business operating under the System, including rights, privileges, and obligations set forth in this Agreement for the establishment and operation of

installing, repairing and replacing residential and commercial glass, windows and doors; and, providing other related services and selling other related products to residential and commercial customers.

"Trade Secrets" means information in any form (including, but not limited to, technical or non-technical data, formulas, patterns, compilations, programs, devices, methods, techniques, drawings, processes, financial data, financial plans, product plans, passwords, lists of actual or potential customers or suppliers) related to or used in The Glass Guru Business that is not commonly known byor available to the public and that information: (a) derives economic value, actual or potential, from not being generally known to, and not being readily ascertained by proper means by, other persons who can obtain economic value from its disclosure or use; and (b) is the subject of efforts that are reasonable under the circumstances to maintain its secrecy;

"Trademark" includes trademarks, service marks, names, logos, and other commercial symbols.

2 GRANT OF FRANCHISE

2.1 Grant

Franchisor grants to Franchisee, and Franchisee undertakes and accepts, upon the termsand conditions of this Agreement, a revocable, limited license to operate a Franchise Business at one (1) location approved by Franchisor using the System and Marks. Franchisee shall be prohibited from operating more than one (1) location under this Agreement, but may request Franchisor to grant an additional franchise to Franchisee foran additional location in the Area of Primary Responsibility as described in Section 2.3.

2.2 Approved Location

The street address (or detailed description of the premises) of the Location approved in writing by Franchisor for the Franchise Business as described in Section A of Exhibit 1.

2.3 Area of Primary Responsibility

The "Area of Primary Responsibility" is an area surrounding the Franchise Business location and is defined using US zip codes. This area is considered the geographic market boundary for the Franchise Business. As long as this Agreement is in full force and effect and as long as you are in compliance with the Minimum Performance Requirements described in Section 5.11, Franchisor will not establish or operate or grant others the right to establish and operate a The Glass Guru Business at any location within the Area of Primary Responsibility. The population of an Area of Primary Responsibility for a Franchise Business will typically range from 150,000 to 800,000; however, the said range is not intended to establish a minimum population or a maximum population for an Area of Primary Responsibility. Population will be determined by referencing the most recent figures available from the United States Census Bureau at the time the Area of Primary Responsibility is designated. **The Area of Primary Responsibility is described in Section B of Exhibit 1.**

2.3.1 Unless as described in Section 5.12, the Area of Primary Responsibility shall not be subject to change duringthe Term unless Franchisee requests and Franchisor consents to Franchisee establishing an additional location approved by Franchisor with the Area of Primary Responsibility. If Franchisor consents to such request, an additional franchise agreement will be granted to Franchisee and Franchisee's existing Area of Primary Responsibility will be split between the two (2) Franchise Agreements in the manner agreed upon bythe parties. Upon the execution of the second Franchise Agreement, this Agreement will be amended in writing to revise the Area of Primary Responsibility.

2.4 <u>Nonexclusive License</u>

The Franchise granted under this Agreement is nonexclusive and Franchisor retains all rights and discretion with respect to the Marks and System, including the right to:

- 2.4.1 establish or operate, and to grant others the right to establish or operate, aThe Glass Guru Business at any location as Franchisor deems appropriate; provided however such established businesses and operations are conducted outside the Area of Primary Responsibility specified by the Franchisor for the Franchise Business:
- 2.4.2 establish or operate, and to grant others the right to establish or operate, other businesses offering the same or similar products or services as the Franchise Business utilizing other trade names, trademarks and service marks within or outside of the Area of Primary Responsibility;
- establish or operate, and to grant others the right to establish or operate businesses offering products and services different than those offered by the Franchise Business utilizing the Marks or other trade name, trademarks and service marks within or outside of the Area of Primary Responsibility;
- 2.4.4 to acquire or be acquired by a company establishing businesses identical or similar to the Franchise Business, even if the other business operates franchises and/or licenses Competing Businesses anywhere, including your Area of Primary Responsibility;
- 2.4.5 sell or otherwise distribute products and services similar to those offered through the Franchise Business through any alternate channel of distribution, including through retail outlets at a fixed location and by direct mail and Internet sales, under terms and conditions that Franchisordeems appropriate; and
- 2.4.6 engage in any activities not expressly forbidden by this Agreement.

2.5 Marketing, Solicitation and Service Restrictions for Areas of Primary Responsibility

2.5.1 Other than for Immediate Family Members, Franchisee must not directly

provide services or sell product in areas inside the Area of Primary Responsibility of another The Glass Guru franchisee;

- 2.5.1.1 If we determine that you, prior to obtaining the express written consent of Franchisor, provide Services or sell Products in the Area of Primary Responsibility of another The Glass Guru franchisee, you must pay the franchisee that has the right to the infringed Area of Primary Responsibility an amount equal to the total sums you received for the Services and Products, less the actual cost you paid for the Products sold and any third party hired labor costs. All sums must be paid within ten (10) days after receiving notice from Franchisor. A second violation could result in the termination of this Agreement. Services to Immediate Family Members are exempt from this Section.
- 2.5.1.2 Further, Franchisee must not advertise in any form of media whose primary location, broadcast or circulation is outside of their Area of Primary Responsibility without the prior written approval of Franchisor. Franchisor shall make reasonable efforts to enforce these restrictions with regard to Franchisee and any other The Glass Guru Business, but under no circumstances must Franchisor be required to engage in litigation or similar actions with regard to these restrictions.
- 2.5.2 Marketing, Solicitation and Service Restrictions for areas not granted.

Other than for Immediate Family Members, if you wish to use any part of the System (including, but not limited to, the providing of any Services) and/or the Marks outside your Area of Primary Responsibility in an area not yet franchised by us, we may (but arenot obligated to) allow you to do so on a temporary basis under such terms and conditions as we deem fit, including (but not limited to) our granting such rights on a non-exclusive basis. Any work completed outside your Area of Primary Responsibility is subject to our right to charge a fee (a Non-APR Service Fee') per customer for authorizing such

Also, upon receipt of notice from us, you must immediately stop all marketing and salesactivities in any area outside of your Area of Primary Responsibility and submit a full accounting of all current activity to us including all leads, proposals, and Sales. Franchisor has the sole discretion to determine the process for completing all activities for customers and the process for making the transition to any new or current franchisee. You do NOT retain rights to ANY customers outside of your Area of Primary Responsibility.

2.6 <u>Sub-franchising/Agents</u>

Franchisee must not sublicense the use of the System or Marks to any person or entity to perform any part of Franchisee's rights or obligations licensed hereunder, or to grant any person or entity the right to act as Franchisee's agent to perform any part of Franchisee's rights or obligations hereunder.

2.7 Notice to Franchisor for Certain Events

Franchisee must notify Franchisor in writing, within five (5) days of the event, of any of the following events: (1) the commencement of any civil or criminal action, suit, or proceeding by Franchisee or by any person or government agency against Franchisee; (2)Franchisee receives a notice of noncompliance with any law, rule, or regulation; (3) theissuance of any order, suit, or proceeding of any court, agency, or other governmental body that may adversely affect the operation or financial condition of the Franchise Business; (4) any complaints, inspections, reports, warnings, certificates, or ratings of Franchisee or the Franchise Business, communicated, issued, performed, or scheduled by any governmental agency; (5) the scheduling or conducting of an audit of Franchisee by the Internal Revenue Service or any other federal, state, or local governmental authority; and (6) any unionization effort, collective bargaining agreement, labor strike, dispute, slowdown, work stoppage, or lockout. Franchisee must notify Franchisor within one hour of the incident, using the emergency contact procedure specified by Franchisor, of any emergency situation relating to the Franchise Business, including but not limited to: (a) afatality; (b) an accident; (c) an injury requiring medical attention; (d) media attention; (e) significant property damage; or (f) any incident involving a law enforcement agency. Franchisee must provide Franchisor with any additional information Franchisor requests, within five days of request, about the status, progress, or outcome of any of the events listed in this Section.

2.8 Right to Obtain Franchisee's Credit Report/Background Checks.

Franchisee consents to permitting Franchisor to obtain Franchisee's credit report and/orto perform background checks, including criminal, at any time to confirm that Franchisee is paying third- party creditors and otherwise meeting the standards of Franchisor, and Franchisee agrees to cooperate and to sign any authorizations necessary to enable Franchisor to

acquire Franchisee's credit report and/or to perform background checks.

2.9 <u>Supplemental Exhibits and Agreements</u>

Franchisee is required to sign supplemental agreements simultaneous with the execution of this Agreement, including the following:

- A. Exhibit 1, <u>Specifics</u>. This document describes the Area of Primary Responsibility, the Franchise Fee, and provides other information.
- B. Exhibit 3, <u>Non-Disclosure and Non-Competition Agreement</u>. Simultaneously with the signing of this Agreement, Franchisee will sign the Non-Disclosure and Non-Competition Agreement in the form of Exhibit 3. This Agreement provides for the protection of the System and the Marks.
- C. Exhibit 4, <u>Guaranty and Assumptions of Obligations</u>. The shareholders, members, officers, directors, and representatives of and the Affiliates of

Franchisee involved with or that provide assets or services to the Franchise Business must sign this document to agree to be liable to Franchisor for the obligations of Franchisee.

- D. Exhibit 5, <u>Holders of Beneficial Interest in Franchisee</u>; <u>Officers, Directors</u> and Managers. Franchisee must provide Franchisor with the names and contact information for the owners of Franchisee and other persons beneficially interested in Franchisee, officers, directors and managers of the Franchisee and to provide updates to this information as it occurs.
- E. Exhibit 6, <u>Assignment of Telephone Numbers and Internet Tools</u>. Simultaneously with the signing of this Agreement and any time thereafter, as Franchisor requests, Franchisee will sign an assignment of the telephone numbers, Internet domain names, e-mail addresses, websites, social media (such as Facebook, LinkedIn, Twitter, YouTube), blogs, vlogs (social videos), online social networks, wikis, forums, content sharing communities, and other internet tools ("Telephone Numbers and Internet Tools") used by Franchisee in the Franchise Business in the form of Exhibit 6 attached to this Agreement. The assignment provides that Franchisee will have a limited license to use the Telephone Numbers and Internet Tools during the term of this Agreement and as long as Franchisee complies with the policies and procedures specified by Franchisor.
- F. Exhibit 7, <u>Lease Addendum</u>. Franchisee shall be required to execute this Addendum along with the landlord for the lease for the Approved Locationas a condition of approval of the lease.

3 FEES, OTHER PAYMENTS AND PROCEDURES

Franchisee must pay to Franchisor the following fees and other items listed below in the amounts, at the intervals and in accordance with the procedures described in this Agreement. Each such payment by Franchisee must be payable separately and is in addition to all other payments required in this Agreement.

3.1 Franchise Fee

In consideration of the rights and license granted by Franchisor, Franchisee agrees to pay, at the time of signing of this Agreement, a non-refundable initial franchise fee (the "Franchise Fee") to Franchisor in the amount stated in Section C of Exhibit 1 The Franchise Fee is payment, in part, for expenses incurred by Franchisor in furnishing assistance and services to Franchisee as described in this Agreement and for costs incurred by Franchisor, including general sales

and marketing expenses, training, legal, accounting, and other professional fees. The fee reflected in Section C of Exhibit 1 reflects any applicable incentive(s) that may have been applied toward the Franchise Fee as determined by Franchisor at the time the Agreement is executed.

3.2 <u>Monthly Royalty Fee</u>

On the twentieth (20th) day of each month, Franchisee must pay to Franchisor without offset, credit or deduction of any nature, so long as this Agreement is in effect, a non-refundable royalty fee ("Royalty Fee") equal to five percent (5%) of Gross Sales for the previous calendar month.

In the case of a Converted Business, Franchisee must pay to Franchisor Royalty Fee reflecting a mutually agreed upon discount schedule based on one of two optionsreflected below.

- 1. For a Franchisee electing to receive a fifty percent (50%) reduction in the Franchise Fee, Franchisee will receive a fifty percent (50%) reduction in the monthly Royalty Fee (as defined above) for the mutually agreed upon Gross Sales from most current twelve (12) month period prior to converting. Any Gross Sales exceeding this amount derived during the first twelve (12) months of operations will be subject to the full monthly Royalty Fee amount defined above. This reduction in the monthly Royalty Fee will terminate following the Franchise Business's second (2nd) full year (12-months) of operation.
- 2. For a Franchisee electing to pay full Franchise Fee, for a period of twenty-four (24) months, Franchisee will receive a fifty percent (50%) reduction in the monthly Royalty Fee (as defined above) for the mutually agreed upon Gross Sales from most current twelve (12) month period prior to converting. Any gross sales exceeding this amount derived during this twelve (12) month period will be subject to the full Monthly Royalty Fee amount as defined in section 3.2 of the Franchise agreement. This reduction in monthly Royalty Fees willreset and renew for a subsequent twelve (12) month period during the franchise Business 2nd full year of operation. This reduction in the monthly Royalty Fee will terminate following the Franchise Business's second (2nd) full year (24-months) of operation.

Any discount schedule applying to Royalty Fee must be reflected in an amendment to the Agreement executed by all parties at the time the Agreement is executed.

You are not required to pay the Royalty Fee on the value of any portion of goods and/or services provided without compensation to you if those goods and/or services are provided for a charitable organization approved by us ("Guru Charity"). Currently, all charitable organizations listed by the Internal Revenue Service ("IRS") as exempt under Internal Revenue Code §501(c)(3) ("501(c)(3) organizations") are automatically approved by us. Goto IRS.gov to search for a list of 501(c)(3) organizations. If you provide goods and/or services without

compensation that are not for an organization that is listed by the IRS as a 501(c)(3) organization or a charitable organization otherwise approved by us, the value of those goods and/or services must be included in your Gross Sales for purposes of paying royalty to us. The value of the goods and/or services provided without compensation will be equal to the amount that you would normally charge for the goodsand/or services Franchisor requires Franchisee to pay Royalty Fees through electronic transfer as described in Section 3.6.

3.3 <u>Marketing Fee</u>

Franchisee must pay to Franchisor without offset, credit or deduction of any nature, so long as this Agreement is in effect, a monthly Marketing Fee ("Marketing Fee") equal tothree percent (3%) of Gross Sales for the previous calendar month. Marketing Fee Contributions must be made at the same time and in the same manner as Royalty Fees. Franchisor shall spend Marketing Fees collected from franchisees as it deems advisable in its sole discretion.

3.4 Non-APR Service Fee

On the twentieth (20th) day of each month, Franchisee must pay to Franchisor without offset, credit or deduction of any nature, so long as the Franchise Agreement is in effect, a fee ("Non-APR Service Fee") equal to two percent (2%) of Gross Sales for the previouscalendar month for any and all services provided and products sold outside of Franchisees Area of Primary Responsibility.

3.5 <u>Technology Fee</u>

Franchisee must pay to Franchisor a monthly technology fee in an amount determined by Franchisor ("Technology Fee"). We have the right to determine how and for what purposes the technology fee will be used, which may include covering Franchisor's costs or paying fees to third party providers for technology research and development, maintenance and usage for the franchise system, and subscription and license fees paid by us in order for franchisees to have access to and use certain technology tools. The Technology fee must be made at the same time and in the same manner as Royalty Fees. Franchisor will give Franchisee at least sixty (60) days prior notice before changing the amount of the Technology fee charged.

3.6 Taxes

Franchisee must pay to Franchisor an amount equal to all sales taxes, use taxes and similar taxes imposed on the fees payable by Franchisee to Franchisor hereunder and onservices or goods furnished to Franchisee by Franchisor at the same time as Franchisee remits such fees to Franchisor, whether such services or goods are furnished by sale, lease or otherwise, unless the tax is an income tax assessed on Franchisor for doing business in the state where the Franchise Business is located.

3.7 Payment of Fees

Franchisee must pay the full amount of the Royalty Fee, Marketing Fee, Technology Fee and Non-APR Service Fee as provided above and any and all other fees due under this Agreement,

and no offset or claim may be made against those fees unless otherwise stated in this Agreement. All payments for fees and charges must be made to Franchisor in the manner and at the times specified by Franchisor. Currently Franchisor specifies that: (a) the Royalty Fee, Marketing Fee, Technology Fee, Non-APR Service Fee, improper reporting fees, and late aka non-filing fees must be paid to Franchisor on the twentieth (20th) day of the month following the end of the calendar month in which the revenue was generated; (b) miscellaneous fees and charges, including but not limited to fees for goods or services, must be paid to Franchisor at the time they are incurred. All Payments must be paid via Automated Clearing House("ACH") transfers or other form of electronic payment transfer as Franchisor specifies. In somecircumstances, it may be necessary for Franchisor to ACH transfer funds for miscellaneous fees other on the twentieth (20th) day of the month. In that case, the amount and date of deduction will be communicated to Franchisee in advance. Franchisee must not close the Account associated with Payment of Fees without Franchisor's written consent.

Franchisee must promptly execute and deliver to Franchisor any electronic transfer authorization form required for making ACH transfers upon delivery of such from to Franchisee by Franchisor. Franchisee shall not make any change in the Account used for making payment of the fees due hereunder, including nay change in the account number, closing of any account, issuing stop payment orders, or changing of banks without the prior written approval of Franchisor and execution of new electronic transfer authorization forms.

3.8 <u>Late Fees</u>

All Royalty Fees, Marketing Fees, Technology Fees, Non-APR Service Fees, amounts due for purchases by Franchisee from Franchisor and other amounts that are not received by Franchisor within five (5) days after the due date will incur late fees at the rate of one and one-half percent (1.5%) per month or the maximum rate of interest allowed by the law ofthe state where the Franchisee is located, (whichever is greater) from the date payment isdue to the date payment is received by Franchisor. Franchisee must pay Franchisor for allcosts incurred by Franchisor in the collection of any unpaid and past due Royalty Fees, Marketing Fees, Technology Fee, Non-APR Service Fees or any other amounts due Franchisor, including reasonable accounting and legal fees. This Section must not constitute an agreement by Franchisor to accept any payments after the due date or a commitment by Franchisor to extend credit to or otherwise finance Franchisee.

3.9 NSF Fees and Interest

Franchisee must pay on demand a fee equal to any charges Franchisor may incur as a result of ACH or other transfers returned to Franchisor for non-sufficient funds or other reasons, but not less than Seventy-five (\$75.00) for each item returned ("NSF fees"). Also, Franchisee must pay to Franchisor, on demand, interest on all overdue payments from the date the payment was due until paid equal to the greater of (i) one and one-half percent (1.5%) per month or (ii) the maximum rate of interest permitted by law. The assessment of NSF fees and interest will not be the sole remedies of Franchisor in suchcircumstances. All amounts paid will be first applied to NSF fees and interest and the balance to principal.

3.10 Application of Payments

Notwithstanding any designation by Franchisee, Franchisor has the right to apply all orany portion of any payments by Franchisee to any past due indebtedness of Franchiseefor Royalty

Fees, Marketing Fees, Technology Fees, Non-APR Fees, purchases from Franchisor or any other amount owed to Franchisor in any proportion or priority.

4 TERM AND RENEWAL

4.1 <u>Initial Term</u>

This Agreement shall be effective and binding for an initial term of five (5) years from the Effective Date, unless sooner terminated under Section 16.

4.2 Successor Terms

Subject to the conditions below and Section 4.3, Franchisee has the right to obtain a successor franchiseat the expiration of the term of this Agreement by entering into a new franchise agreement with Franchisor. Franchisee's right to a successor franchise is limited to three (3) successive terms of five (5) years each, such that the total term of the Franchise may not exceed twenty (20) years. Following the expiration date of three (3) successive terms Franchisor has the right to charge a franchise fee commensurate with the current and prevailing Franchise Agreement as of that date. To qualify for a successor franchise, each of the following conditions must have been fulfilled and remain true as of the last day of the term of this Agreement:

- 4.2.1 Franchisee has, during the entire term, substantially complied with all material provisions of this Agreement;
- 4.2.2 Franchisee has at its expense, made such capital expenditures as were necessary to maintain uniformity with any Franchisor-required System modifications such that the Franchise Business reflects Franchisor's then-current standards and specifications;
- 4.2.3 Franchisee has satisfied all monetary obligations owed by Franchisee to Franchisor (or any Affiliate), and has timely met these obligations throughout the term of this Agreement;
- 4.2.4 Franchisee is not in default of any provision of this Agreement or any other agreement between Franchisee and Franchisor;
- 4.2.5 Franchisee has given written notice of its intent to operate a successor franchise to Franchisor not less than six (6) months nor more than twelve (12) months prior to the end of the term of this Agreement
- 4.2.6 Franchisee has executed at Franchisor's option either (i) Franchisor's then-The Glass Guru Enterprises, Inc. 2024 Franchise Agreement

21

current form of the franchise agreement (or has executed other documents at Franchisor's election thatmodify this Agreement to reflect the fact that the Franchise Agreement relates to the grant of a successor franchise), which franchise agreement must supersede this Agreement in all respects, and the terms of which may differ from the terms of this Agreement by requiring, among other things, a different percentage Royalty Fee or Marketing Fee; and a different or reduced Area of Primary Responsibility; provided, however, that Franchisee will not be required to pay the then-current Franchise Fee; or (ii) an extension of this agreement.

- 4.2.7 Franchisee has complied with Franchisor's then-current qualifications for anew franchisee and has agreed to comply with any training requirements;
- 4.2.8 Franchisee and it's owners have executed a general release, in a form the same as or similarto the General Release attached as Exhibit 2, releasing any and all claims against Franchisor, any Affiliate and their officers, directors, shareholders, managers, members, partners, owners and employees.
- 4.2.9 Unless otherwise noted, all conditions to renewal must be met before the expiration of the initial term of this Agreement. If all of the other applicable conditions to renewal have been met, Franchisor will not withhold approval of renewal without good cause. Notwithstanding the foregoing, Franchisor may withhold approval of renewal if Franchisor has withdrawn from the market in which Franchisee's Area of Primary Responsibility is located and no longer operates or grants others the right to operate a The Glass Guru business in such market.

4.3 Good Cause for Withholding Approval of Renewal

Good cause for withholding approval of renewal may include, but is not limited to: Franchisee has failed to achieve minimum scores in Franchisor's Achievements in Excellence rankings or a similar Franchisor ranking system (the Achievements in Excellence rankings is a balanced scorecard approach used by Franchisor to measure customer service, operational excellence, learning and growth, community involvement, and system citizenship); Franchisee has failed to achieve minimum reply card scores; Franchisee has failed to achieve acceptable status on other compliance requirements; Franchisee's lack of commitment to Franchisor's core values; Franchisee has failed to use and cooperate in the use of operating systems and tools provided by Franchisor to improve the Franchise Business and/or the franchise system as a whole; Franchisee has failed to regularly attend and actively participate in conference calls, meetings, conventions, and other events sponsored or suggested by Franchisor to increase the chance of success and/or maximize the performance of the Franchise Business and/or the franchise system as a whole; Franchisee has failed to embrace new programs and business building initiatives developed for the enhancement of the performance of the Franchise Business; Franchisee has regularly acted in a combative or confrontational manner with Franchisor, vendors, customers, or other franchisees; Franchisee has had an excessive number of customer complaints and/or has not acted reasonably and in the best interests of the franchise system in resolving customer complaints; or Franchisor hasreasonable concerns

about the financial condition or creditworthiness of Franchisee.

If Franchisee does not meet all of the conditions for renewal as described above by the date of expiration of the Term and Franchisee has made good faith efforts to do so, Franchisor may, in its sole discretion, extend the Term of this Agreement for a reasonable period of time to permit Franchisee to take all necessary actions to meet the conditions for renewal of this Agreement. Any such term shall be in writing signed by both parties. However, nothing herein shall obligate Franchisor to grant such additional time to Franchisee.

4.4 Right to Change Area of Primary Responsibility on Renewal of Franchise

If Franchisee's franchise is renewed, Franchisor may change the Area of Primary Responsibility granted in the new franchise agreement signed in connection with the renewal or transfer in order to comply with Franchisor's then current manner of designating Area of Primary Responsibility, to make minor changes in the Area of Primary Responsibility to correct overlap or other issues, and for other valid businessreasons.

4.5 <u>Continued Operation Following Expiration</u>

If Franchisor permits Franchisee to continue to operate the Franchise Business after the expiration of the term of this Agreement, but before the execution of a renewal Franchise Agreement as required by this Section 4, then the temporary continuation of the Franchise Business will be on a month-to-month basis, and this Agreement will be terminable at Franchisor's will by giving Franchisee written notice of termination at least thirty (30) days before the termination is effective. If applicable law requires a longer notice period, the thirty-day period will be deemed modified to be the shortest notice period required by the laws of such jurisdiction.

5 ESTABLISHING THE FRANCHISE BUSINESS

5.1. <u>Selection of Site</u>

Franchisee must promptly select a site for the Franchise Business within the Area of Primary Responsibility and must notify Franchisor of such selection. Franchisor shall evaluate the site and notify Franchisee of its approval or disapproval of the site within a reasonable time of receiving notice of thesite from Franchisor. If Franchisor has not disapproved of the initial location or any substitute location proposed by Franchisee within fifteen (15) days of receiving written notice of the location, then Franchisor will be deemed to have approved it. Upon approval of such selection, the site must be designated as the Approved Location. If Franchisor does not approve of such selection, Franchisee must select and notify Franchisor of new sites until Franchisor approves a site for the Franchise Business, whichwill not be unreasonably withheld. Franchisor shall provide Franchisee with general guidelines to assist Franchisee in selecting a site suitable for the Approved Location.

Franchisor has the right to approve or disapprove a proposed location based on such factors as it deems appropriate, including the size and condition of the premises, demographics of the surrounding area, proximity to other The Glass Guru Business, lease requirements (if

applicable) visibility and overall suitability. Franchisee must not locate the Franchise Business on a selected site without the prior written approval of Franchisor.

Franchisee must not locate the Franchise Business on a new selected site without the prior written approval of Franchisor. The essence of Franchisor's core values, care, and integrity requires that each franchisee in the franchise system respect all other franchisees. When Franchisee selects a location for its Franchise Business, Franchisee must consider each of the factors described in this paragraph, while honoring Franchisor's core values. Franchisor does not represent that it or its Affiliate, owners or employees have special expertise in selecting sites. Neither Franchisor's assistance nor approval is intended to indicate or indicates that the Franchise Business will be profitableor successful at the Approved Location. Franchisee is solely responsible for identifying the Approved Location.

5.2. Failure to Select Site

Should Franchisee fail to select a site for the Franchise Business, which meets with Franchisor's approval within sixty (60) days after the Effective Date, Franchisor has the right to terminate this Agreement. If this Agreement is terminated under this Section, Franchisor will **not** return to Franchisee the Franchise Fee paid by Franchisee. Franchiseewill be required to present Franchisor a general release, the same as or similar to the General Release attached as Exhibit 2, releasing any and all claims against Franchisor, anyAffiliate and their officers, directors, shareholders, managers, members, partners, owners and employees. Franchisee is not entitled to a refund. The non-refundable Franchise Fee is compensation to Franchisor for its efforts in offering and selling a franchise to Franchisee, Franchisor's marketing and sales activities to promote the sale of a franchiseto qualified franchisees, Franchisor's participation in the franchise sale, Franchisor's legalcompliance with franchise laws and regulations, site selection assistance and guidelines and the development and hosting of initial training programs and participation in terminating the franchise.

5.3. Lease of Approved Location

If Franchisee is to execute a lease for, or a binding agreement to purchase for, the Approved Location, Franchisee must obtain Franchisor's approval of the terms. Franchisor shall not unreasonably withhold its approval. Franchisor's review of a lease orany advice or recommendation offered by Franchisor does not constitute a representation or guarantee that Franchisee will succeed at the Approved Location nor constitute an expression of Franchisor's opinion regarding the terms of such lease or purchase agreement. Franchisor shall be entitled to require that nothing in the lease is contradictory to, or likely to interfere with, Franchisor's rights or Franchisee's duties under this Agreement. The franchisee must take all actions necessary to maintain the lease of the Approved Location while this Agreement is in effect. Any default for which the lease may be terminated may also be deemed a default hereunder and the time to cure the same should expire when the lease is terminated.

As a condition of obtaining Franchisor's consent to any lease for the Approved Location, Franchisor may require that Franchisee and the landlord execute the Addendum to Leaseand Collateral Assignment of Lease attached to this Agreement as Exhibit 7, or alternatively, providing similar language in the form of lease to be executed by Franchisee and the landlord.

5.4. Development of Franchise Business and Vehicles

- 5.4.1. Franchisor shall make available to Franchisee, at no charge to Franchisee, copies of specifications for an office, vehicle, and other equipment necessary for the operation of a The Glass Guru Business. Franchisee mustcause the Franchise Business to be developed and equipped in accordance with such requirements and specifications prior to, but no later than the Franchisee's initial training date, or ninety (90) days after the execution date of the Agreement. In connection with the development of the Franchise Business, Franchisee must:
 - 5.4.1.1. obtain all permits and licenses required for operation of the Franchise Business, and certify in writing that all such permits and certifications have been obtained;
 - 5.4.1.2. purchase or lease a vehicle meeting Franchisor's specification and all other required signs and equipment and install such signs and equipment in either the vehicle or Franchisee's office, as appropriate; The use of any vehicle other than those specified must be approved in writing by Franchisor.
 - 5.4.1.3. purchase The Glass Guru Restoration Tools as described in Section 13.2, the Brand Identity Package as described in Section 11.1 and any other any supplies or inventory necessary for the operation of the Franchise Business, as specified in the Manual;
 - 5.4.1.4. establish broadband or high-speed Internet access and obtain at least one business telephone number solely dedicated to the Franchise Business;
 - 5.4.1.5. purchase required number of mobile devices including mobile phones and tablets and establish wireless service connection(s) to the device(s) or mobile hotspot(s); and,
 - 5.4.1.6. purchase and install all signs, furniture and office equipment and office supplies, including any required computer hardware and software, required for the management and administration of the Franchise Business and the maintenance and storage of the books and records of the Franchise Business.

5.5. <u>Franchisee Entity Requirements</u>

During the Term of this Agreement, the entity that is the Franchisee hereunder must be in compliance with the following:

- 5.5.1. Franchisee's organizational documents provide that its activities are confined exclusively to the operation of The Glass Guru franchise businesses unless Franchisee obtains Franchisor's prior written consent for other business activities to be conducted under the entity that is the Franchisee. If Franchisor provides consent for Franchisee to conduct multiple business operations under the entity, Franchisee must use the dba with The Glass Guru designation exclusively for the business activities conducted under this Agreement, must maintain separate books and records, and financial statements for the Franchise Business for reporting to Franchisor, must maintain a separate insurance policy for the Franchise Business operations, and meet any other conditions required by Franchisor. However, in no event may Franchisee operate any competing business under the entity that is the Franchisee.
- 5.5.2. copies of the Franchisee's articles of incorporation or organization, bylaws, operating agreement, federal tax identification number and other governing regulations or documents, including resolutions of the board of directors or limited liability company managers authorizing entry into this Agreement, have been promptly furnished to Franchisor. Any amendment to any such documents must also be furnished to Franchisor immediately upon adoption.
- 5.5.3. each stock certificate or other ownership interest certificate of the Franchiseehas conspicuously endorsed upon the face of that certificate the statement "The transfer of the [shares][ownership interest] represented by this certificate is subject to the terms and conditions of a certain written franchise agreement entered into with The Glass Guru Enterprises, Inc.", ora statement in a form satisfactory to Franchisor that it is held subject to, and thatany assignment or transfer of ownership interest and/or the ownership certificate is subject to, all restrictions imposed upon transfers and assignmentsby this Agreement; and
- 5.5.4. Franchisee shall remain in good standing with the state in which it was incorporated or organized;
- 5.5.5. Franchisee shall promptly advise Franchisor of any changes in the information contained in Exhibit 5 attached hereto on owners, officers, directors and managers of Franchisee.

5.6. Opening

- 5.6.1. Before opening the Franchise Business and commencing business, Franchisee must:
 - 5.6.1.1 fulfill all of the obligations of Franchisee under the other provisions of this Section 5;
 - 5.6.1.2 furnish Franchisor with copies of all insurance policies required by this

Agreement, or by any lease (if applicable), or such other evidence of insurancecoverage and payment of premiums as Franchisor may request;

- 5.6.1.3 ensure that the Franchisor approved Designated Manager has completed initial training to the satisfaction of Franchisor;
- 5.6.1.4 hire any other personnel necessary or required for the operation of the Franchise Business;
- 5.6.1.5 obtain all necessary permits and licenses;
- 5.6.1.6 pay in full all amounts due to Franchisor.
- 5.6.2 Franchisee must comply with these conditions and must open and continuously operate the Franchise Business within ninety (90) days afterthe Effective Date. Time is of the essence.

5.7. Failure to Open

Should Franchisee fail to commence operations of the Franchise Business within ninety (90) days after the Effective Date, Franchisor has the right to terminate this Agreement. If this Agreement is terminated under this Section, Franchisor will retain the entire Franchise Fee paid by Franchisee. The Franchise Fee retained is specifically understood and agreed by the parties to be in consideration of the services provided, time expended, work performed, and other efforts of Franchisor up to the date of Franchisee's failure to timely commence operations of the Franchise Business and must not be construed nor considered to be a penalty.

5.8. Use of Approved Location

Franchisee must not use the Approved Location for any purpose other than for the operation of Franchise Business in full compliance with this Agreement and the Manual, unless approved in writing by Franchisor.

5.9. Relocation of the Franchise Business

Franchisee must not relocate the Franchise Business without the prior written consent of Franchisor, which consent will not be unreasonably withheld or delayed. If the lease for the Approved Location expires or is terminated without the fault of Franchisee or if the Franchise Business's premises is destroyed, condemned or otherwise rendered unusable, or as otherwise may be agreed upon in writing by Franchisor and Franchisee, Franchisor may, in its reasonable discretion, allow Franchisee to relocate the Franchise Business. Anysuch relocation will be at Franchisee's sole expense and will proceed in accordance with the requirements described in Sections 5.1 through 5.8. Franchisor has the right to charge Franchisee for any costs incurred by Franchisor in providing assistance to Franchisee, including legal and

accounting fees. Notwithstanding the foregoing, Franchisor has no obligation to provide relocation assistance. If Franchisor and Franchisee do not agree upon a substitute site within ninety (90) days after the lease expires or is terminated or the Approved Location is rendered unusable, or if Franchisor grants approval of a substitute site but Franchisee fails to promptly relocate the Franchise Business, Franchisor may terminate this Agreement as provided in Section 16.2.1.1.

5.10. Employee Non-disclosure and Confidentiality Agreements

Subject to applicable law, Franchisee must require its employees and agents to sign such agreements and documents as Franchisor in its sole discretion deems necessary to maintain the confidentiality and proprietary nature of Franchisor's materials and documents.

5.11. <u>Minimum Performance Requirements</u>

The Franchise Business must achieve the minimum performance requirements specified in this Section ("Minimum Performance Requirements").

- 5.11.1 Minimum Performance Requirements. For the 1st through 4th years of operation of a Unit in the Area of Primary Responsibility, Franchisee must achieve annual Gross Sales of at least the following amounts: (i) One Hundred Fifty Thousand Dollars (\$150,000) for the first (1st) year of operation; (ii) Two Hundred Thousand Dollars (\$200,000) for the second 2nd year of operation; (iii) Two Hundred Fifty Thousand Dollars (\$250,000) for the third (3rd) year of operation; and (iv) Three Hundred Thousand Dollars (\$300,000) for the fourth (4th) year of operation. After a Franchise Business has been operating in the Area of Primary Responsibility for four (4) years, for each subsequent year of operation Franchisee must: (A) achieve Gross Sales of at least Three Hundred Fifty Thousand Dollars (\$350,000); and (B) perform in the top ninety percent (90%) of all Franchise Business operating in the country of origin in the annual growth percentage of Gross Sales as determined by Franchisor. For purposes of this Section, a year of operation is the twelve (12) month period beginning on the first date of operation of a Franchise Business in the Area of Primary Responsibility and each anniversary of that date. However, if the first date of operation of a Franchise Business in the Area of Primary Responsibility isnot the first day of the month, a year of operation will be the twelve (12) month period beginning on the first (1st) day of the calendar month after the first (1st) day operation and each anniversary of that date. Except as provided in subsection 5.11.2, the time periods specified in this subsection begin on the date that a Franchise Business was first operated in the Area of Primary Responsibility, whether or not operated by Franchisee.
- 5.11.2 Notwithstanding the prior section, if Franchisee acquires a Franchise Business that includes an Area of Primary Responsibility (and/or portion thereof) of an existing or former Franchise Business (i.e. the Area of Primary Responsibility is "transferred" to Franchisee by another franchisee or the Area of Primary

Responsibility granted to Franchisee had previouslybeen the Area of Primary Responsibility or part of the Area of Primary Responsibility of a former Franchise Business), but there was no Franchise Business operating in that Area of Primary Responsibility for a period of sixmonths or more at the time that

Franchisee acquired the Franchise Business, then the time periods specified in the prior section will begin on the date that Franchisee begins operation of the Franchise Business.

5.12. <u>Franchisor's Remedies Relating to Minimum Performance Requirements</u>

If Franchisee fails to achieve the Minimum Performance Requirements for a year of operation, Franchisor may notify Franchisee of the failure. If Franchisee fails to achievethe Minimum Performance Requirements again the following year of operation or if Franchisee is not in compliance with the Minimum Performance Requirements at the time of renewal, then Franchisor may, by written notice to Franchisee, elect to:

- A. Require Franchisee to enter into a performance improvement plan, which may include retraining;
- B. Reduce the Area of Primary Responsibility (the reduced Area of Primary Responsibility will include the Approved Location, but will otherwise be determined by Franchisor in its sole discretion);
- Offer to renew this Agreement at the end of its term based on a reduced-insize Area of Primary Responsibility as determined by Franchisor in its sole discretion;
- D. Terminate this Agreement; and/or
- E. Refuse to renew this Agreement at the end of its term.

The remedies in this Section are in addition to any other remedies of Franchisor under this Agreement.

6 PROPRIETARY MARKS

6.1 <u>Ownership</u>

Franchisee's right to use the Marks is derived solely from this Agreement, is non-exclusive and is limited to the conduct of business by Franchisee under, and in compliance with, this Agreement and all applicable standards, specifications and operating procedures prescribed from time to time by Franchisor. Any unauthorized useof the Marks by Franchisee is a breach of this Agreement and an infringement of the rights of Franchisor in and to the Marks. Franchisee's use of the Marks, and any goodwill created, must inure to the benefit of Franchisor. Franchisee must not at any time acquirean ownership interest in the Marks by

virtue of any use it may make of the Marks. This Agreement does not confer any goodwill, title or interest in the Marks to Franchisee. Franchisee must not, at any time during the term of this Agreement or after its termination or expiration, contest the validity or ownership of any of the Marks or assistany other person in contesting the validity or ownership of any of the Marks.

6.1.1 Promise Not to Contest Validity or Ownership of Marks

Franchisee expressly promises that during the term of this Agreement and after the termination, expiration without renewal, or transfer of this Agreement, Franchisee will not, directly or indirectly, contest or aid in contesting the validity or ownership of the Marks. Immediately upon termination, expiration without renewal, or transfer of this Agreement, Franchisee will cease and desist from using the Marks and will return or destroy all documents, instructions, displays, paper products, and other materials and advertising items and the like bearing any of the Marks. Franchisee agrees not to interfere with, in any manner, or attempt to prohibit the use of the Marks by any other existing or future franchisee or other licensee of Franchisor. Whenever Franchisor requests, Franchisee agrees to sign any and all other papers, documents, and/or assurances to effectuate this purpose and agrees to fully cooperate with Franchisor and/or any other franchisee to secure the necessary and required consents of any governmental agency.

6.2 Limitations on Use

Franchisee agrees to use the Marks as its sole identification for the Franchise Business, except that Franchisee agrees to identify itself as an independent owner in the manner Franchisor approves. Except as otherwise explicitly authorized by this license or as Franchisor may otherwise authorize in writing, Franchisee may not use any Mark: (i) as part of any entity or legal business name, (ii) with any prefix, suffix, or other modifying words, terms, designs or symbols, (iii) in selling any unauthorized services or products, (iv) as a part of or in connection with any Internet domain names, email addresses, websites, social media (such as Facebook, LinkedIn, Twitter, YouTube), blogs,vlogs (social videos), online social networks, wikis, forums, content sharing, or other Internet tools, or (e) in any other manner that Franchisor has not expressly authorized inwriting. Franchisee may not use any Mark in advertising the transfer, sale, or other disposition of the Franchise Business or an ownership interest in the Franchisee without Franchisor's prior written consent, which Franchisor will not unreasonably withhold.

Franchisee agrees to display the Marks prominently as Franchisor prescribes at its Franchise Business and on vehicles, forms, advertising, supplies, and any other materials Franchisor designates. Franchisee agrees to give the notices of trade and service mark registrations that Franchisor specifies and to obtain any fictitious or assumed name registrations required under applicable law. Franchisee must not register or seek to register as a trademark or service mark, either with the United States Patent and Trademark Office or any state or foreign country, any of the Marks or a trademark orservice mark that is confusingly similar to any Mark licensed to Franchisee. Franchisee should include on its letterhead, forms, cards and other such identification, and must display at any office location, a prominent notice stating that the Franchise Business is "Independently Owned and Operated".

6.3 Manner of Using Marks

Franchisee must operate the Franchise Business under the Marks and under no other name or mark. Franchisee agrees to operate the Franchise Business using the Marks in accordance with the terms of this Agreement and the Manuals, as amended from time to time. Franchisee and Franchisee's employees must use the Marks only in the manner authorized by Franchisor in the Manuals or otherwise in writing. The Marks mustonly be used with the letters "SM" or "TM" or ®, as appropriate, wherever the Marks are used. Franchisee will not use its name or any other name that Franchisor has not previously approved in writing in connection with any of the Marks. This means, among other things, that Franchisee will not operate, be employed by, or otherwise be affiliated with another business at or adjacent to the Franchise Business, unless Franchisor, in its sole discretion, authorizes such operation, employment, or affiliation in writing. Franchisee understands that commingling the Marks with the names or Marks of otherswill injure the Marks and System and is grounds for termination of this Agreement.

6.4 Goodwill

Franchisee acknowledges that valuable goodwill is attached to the Marks and that Franchisee will use the goodwill solely as Franchisor authorizes. Franchisee expressly acknowledges that any and all goodwill associated with the Marks, including any goodwill that might be deemed to have accrued through Franchisee's activities, inures directly and exclusively to Franchisor's benefit, except as otherwise provided in this Agreement or by law. Franchisee acknowledges and agrees that its use of the Marks andany goodwill established by that use does not confer any goodwill or other interests in the Marks upon Franchisee (other than the rights expressly conferred by this Agreement). All provisions of this Agreement relating to the Marks apply to any additional Marks Franchisor authorizes Franchisee to use.

6.5 <u>Permitted Business Name</u>

Franchisee and its affiliates must not use any part of any of the Marks or any words similar to any of the Marks in its entity name. Franchisee shall conduct business under the terms of this Agreement using the assumed or fictious name described in Section C of Exhibit 1 where required or permitted by applicable law. Franchisee may register the assumed or fictitious name assigned to Franchisee in Section C of Exhibit 1. Franchisee and its affiliates must not use any part of the Marks or words similar to the Marks as a business name, except as Franchisorauthorizes by written agreement.

6.6 <u>Notification of Infringements and Claims</u>

If Franchisee receives notice, or is informed, of any claim, demand, or suit against Franchisee on account of any alleged infringement, unfair competition, or similar matter relating to Franchisee's use of the Marks, Franchisee agrees to promptly notify Franchisorin writing of any such claim, demand, or suit. Franchisor will then take such action as Franchisor deems necessary and appropriate to protect and defend Franchisee against such claim by any third party. If Franchisee receives notice or is informed or learns that any third party, who Franchisee believes is not authorized to use the Marks, is using the Marks or any name or

mark confusingly similar to the Marks, Franchisee must promptly notify Franchisor of the facts relating to such alleged infringing use. Franchisor will, in its sole discretion, determine whether or not it wishes to take any action against such third party on account of such alleged infringement of the Marks. If Franchisor undertakes the defense or prosecution of any litigation pertaining to any of the Marks, Franchisee agrees to sign any and all documents and do such acts and things as may, in the opinionof Franchisor's counsel, be necessary to carry out such defense or prosecution.

Franchisee does not have any right to, and must not itself, defend or prosecute the Marks.

6.7 Indemnification for Use of Marks

Franchisor may reimburse Franchisee for all expenses reasonably incurred by Franchisee in any trademark or similar proceeding disputing Franchisee's authorized use of any Mark, provided that Franchisee has timely notified Franchisor of such proceeding andhas complied with this Agreement and Franchisor's directions in responding to such proceeding. At Franchisor's option, Franchisor or its designee may defend and control the defense of any proceeding arising directly from Franchisee's use of any Mark. This indemnification shall not include the expense to Franchisee of removing signage or discontinuance of the use of the Marks. This indemnification shall not apply to litigationbetween Franchisor and Franchisee where Franchisee's use of the Marks is disputed or challenged by Franchisor. This indemnification shall not apply to any separate legal feesor costs incurred by Franchisee in seeking independent counsel separate from the counsel representing Franchisor and Franchisee in the event of litigation disputing

Franchisor and Franchisee's use of the Marks.

6.8 <u>Discontinuance of Use</u>

If Franchisor deems it necessary for Franchisee to modify or discontinue use of any of the Marks, and/or use one (1) or more additional or substitute trade names, trademarks, service marks or other commercial symbols, Franchisee must comply with Franchisor's directions within a reasonable time after notice to Franchisee by Franchisor and subject to the limitations in Section 10.2. Franchisor shall not be required to reimburse Franchisee for its expenses in modifying or discontinuing the use of a Mark or any loss ofgoodwill associated with any modified or discontinued Mark or for any expenditures made by Franchisee to promote a modified or substitute Mark.

6.9 Right to Inspect

To preserve the validity and integrity of the Marks and any copyrighted materials licensed hereunder, and to ensure that Franchisee is properly employing the Marks in the operation of the Franchise Business, Franchisor and its designees have the right to enter and inspect the Franchise Business at all reasonable times and, additionally, have the right to observe the manner in which Franchisee renders services and conducts activities and operations, and to inspect facilities, vehicles, equipment, accessories, products, supplies, reports, forms and

documents and related data to ensure that Franchisee is operating the Franchise Business in accordance with the quality control provisions and performance standards established by Franchisor.

Franchisor and its agents must have the right, at any reasonable time, to remove sufficient quantities of products, supplies or other inventory items offered for retail sale, or used in rendering services, to test whether such products or items meet Franchisor's then-current standards. Franchisor or its designee has the right to observe Franchisee and its employees during the operation of the Franchise Business and to interview and survey (whether in person or by mail) customers and employees. Franchisee and its employees must fully cooperate with Franchisor's representatives in conducting any such inspections and must provide Franchisor's representatives with any and all requested information, data, reports and documents requested by Franchisor's representatives and must provide Franchisor's representatives with copies of any requested documents or materials.

6.10 Franchisor's Sole Right to Domain Name

Franchisee must not advertise on the Internet using, and must not establish, create or operate an Internet site or website using a domain name or uniform resource locator containing, the Marks or the words "THE GLASS GURU", or any variation. Franchisor is the sole owner of all right, title and interest in and to such domain names as Franchisormay designate in the Manual.

7 CONFIDENTIAL INFORMATION

7.1 <u>Confidentiality</u>

Franchisee acknowledges that Franchisor will disclose Trade Secrets and other Confidential Information to Franchisee during the training program, through the Manuals, and as a result of guidance furnished to Franchisee during the term of this Agreement. Franchisee does not acquire any interest in the Trade Secrets or other Confidential Information, other than the right to use it in the development and operation of the Franchise Business and in performing its duties during the term of this Agreement. Franchisee acknowledges that the use or duplication of the Trade Secrets or other Confidential Information in any other business venture would constitute an unfair method of competition. Franchisee acknowledges that the Trade Secrets and other Confidential Information are proprietary and are disclosed to Franchisee solely on the condition that Franchisee (and all holders of a legal or beneficial interest in Franchisee and all officers, directors, executives, managers and members of the professional staff of Franchisee): (a) must not use the Trade Secrets or other Confidential Information in any other business or capacity; (b) must maintain the absolute confidentiality of the Trade Secrets and other Confidential Information during and after the term of this Agreement; (c) must not make any unauthorized copies of any portion of the Trade Secrets or otherConfidential Information disclosed in written or other tangible form; and (d) shall adopt and implement all reasonable procedures prescribed from time to time by Franchisor to prevent unauthorized use or disclosure of the Trade Secrets and other Confidential Information. Franchisee must enforce this Section as to its employees, agents and representatives and must be liable to Franchisor for any unauthorized disclosure or use of Trade Secrets or other Confidential Information by any of them.

7.2 <u>Additional Developments</u>

All ideas, concepts, techniques or materials concerning the System or developed, in whole or in part, using Trade Secrets or other Confidential Information, whether or not protectable intellectual property and whether created by or for Franchisee or its owners or employees, must be promptly disclosed to Franchisor and must be deemed the sole and exclusive property of Franchisor and works made-for-hire for Franchisor, and no compensation must be due to Franchisee or its owners or employees therefor. Franchisor has the right to incorporate such items into the System. To the extent any item does not qualify as a "work made-for-hire" for Franchisor, Franchisee shall assign, and by this Agreement, does assign, ownership of that item, and all related rights to that item, to Franchisor and shall sign any assignment or other document as Franchisor requests toassist Franchisor in obtaining or preserving intellectual property rights in the item. Franchisor shall disclose to Franchisee concepts and developments of other franchisees that are made part of the System. As Franchisor may reasonably request, Franchisee shalltake all actions to assist Franchisor's efforts to obtain or maintain intellectual property rights in any item or process related to the System, whether developed by Franchisee or not.

7.3 <u>Exclusive Relationship</u>

Franchisee acknowledges that Franchisor would be unable to protect the Trade Secrets and other Confidential Information against unauthorized use or disclosure and would beunable to encourage a free exchange of ideas and information among The Glass Guru franchisees if owners of The Glass Guru Business and members of their immediate families or households were permitted to hold an interest in or perform services for any Competitive Business.

Therefore, during the term of this Agreement, neither Franchisee nor any holder of a legal or beneficial interest in Franchisee (or any member of their immediate families or households), nor any officer, director, executive, manager or member of the professional staff of Franchisee, either directly or indirectly, for themselves, or through, on behalf of or in conjunction with any person, partnership, corporation, limited liability company or other business entity, unless otherwise authorized in writing by Franchisor, shall:

- 7.3.1 Divert or attempt to divert any business or customer of the Franchise Business or any other Franchise Business to any Competitive Business, by direct or indirect inducement or otherwise, or do or perform, directly or indirectly, any other act injurious or prejudicial to the goodwill associated with the Marks or the System; or
- 7.3.2 Own an interest in, manage, operate, or perform services for anyCompetitive Business wherever located.

7.4 Nondisclosure and Non-Competition Agreements with Certain Individuals

Subject to appliable law, Franchisor has the right to require Franchisee and any holder of a legal or beneficial interest in Franchisee, and any officer, director, executive, manager or member of the professional staff and all employees of Franchisee to execute a nondisclosure and non-competition agreement, in a form the same as or similar to the Nondisclosure and

Non-Competition Agreement attached as Exhibit 3, upon the signing of this Agreement or prior to each such person's affiliation with Franchisee. Upon Franchisor's request, Franchisee must provide Franchisorwith copies of all nondisclosure and non-competition agreements signed under this Section. Such agreements must remain on file at the offices of Franchisee and are subject to audit or review as otherwise described in this Agreement. Franchisor shall be a third- party beneficiary with the right to enforce covenants contained in such agreements.

7.5 Reasonableness of Restrictions

Franchisee acknowledges that the restrictive covenants contained in Section 7 are essential elements of this Agreement and that without their inclusion, Franchisor would not have entered into this Agreement. Franchisee acknowledges that each of the termsdescribed here, including the restrictive covenants, is fair and reasonable and is reasonably required for the protection of Franchisor, the System and the Marks and Franchisee waives any right to challenge these restrictions as being overly broad, unreasonable or otherwise unenforceable.

8 TRAINING AND ASSISTANCE

8.1 <u>Initial Training</u>

Franchisor will make an initial training program available to the Owners and Designated Manager and one (1) office assistant. Up to two (2) additional field or office assistants will also be allowed to attend the initial training program with Franchisor's prior authorization. Prior to the opening of the Franchise Business and within ninety (90) days from the date of this Agreement, the Owner and the Designated Manager must attend and successfully complete, to Franchisor's satisfaction, an initial training program pertaining to operation of the Franchise Business including, but not limited to technical procedures, maintenance of quality standards; customer service techniques; sales and marketing methods; financial controls; record keeping and reporting procedures, and other operational items. Franchisor will conduct the initial training program at its headquartersor at another location designated by Franchisor. Franchisor will not charge tuition or similar fees for initial training; however, all expenses incurred by Franchisee in attending such program including, but not limited to, travel costs, room and board expenses and employees' salaries, will be the sole responsibility of Franchisee.

8.2 <u>Technician Training & Certification Programs</u>

After Franchisee's Owners and Designated Manager attend and successfully complete the initial training program in accordance with Section 8.1, should Franchisee hire additional window restoration technicians, each such technician must be properly trained to provide window restoration services for the Franchise Business. Such technician training may be conducted by the Franchisee should be conducted at either Franchisor's headquarters, Franchisee's location or at another location designated by Franchisor. For training provided by Franchisor at the request of the Franchisee, Franchisee must pay Franchisor's then-current standard rates for technician training programs. In addition, all expenses incurred by Franchisee's technicians in attending such programs including, but not limited to, travel costs,

room and board expenses and employees' salaries, will be the sole responsibility of Franchisee.

8.3 Opening Assistance

In conjunction with the beginning of operation of the Franchise Business, Franchisor willmake available to Franchisee, a dedicated Franchisor Liaison, experienced in the System, for the purpose of providing general assistance and guidance in connection with the opening of the Franchise Business. If Franchisee requests any additional assistance with respect to the opening or continued operation of the Franchise Business such as on- location support, and should Franchisor deem it necessary and appropriate to comply with such request, Franchisee must pay Franchisor's then-current standard rates, plus expenses, for such additional assistance.

8.4 <u>Failure to Complete Initial Training Program</u>

If Franchisor determines that the Designated Manager is unable to satisfactorily complete the training program described above, Franchisee may be permitted to select asubstitute Designated Manager and such substitute Designated Manager must complete the initial training to Franchisor's reasonable satisfaction within (30) days after the Franchisee's initial training date, and no more than ninety (90) days after the execution date of the Agreement. Franchisor may require Franchisee to pay Franchisor's then current rates for additional training for providing initial training to the substitute manager. Franchisee must be responsible for all travel costs, room and board and employees' salaries incurred in connection with the new Designated Manager's attendance at such training.

If the Franchisee, Designated Manager, or designated substitute manager fails to satisfactorily complete the training program described above prior to ninety (90) days after the execution date of the Agreement, Franchisor has the right to terminate this Agreement. If this Agreement is terminated under this Section, Franchisor will **not** return to Franchisee the Franchise Fee paid. Further, Franchisee is required to present Franchisor a general release, the same as or like the General Release attached as Exhibit 2, releasing any and all claims against Franchisor, any Affiliate and their officers, directors, shareholders, managers, members, partners, owners and employees. Franchisee is not entitled to a refund. The Franchise Fee is reimbursement to Franchisor of the expense of its efforts in offering and selling a Franchise to Franchisee, Franchisor's marketing and sales activities to promote the sale of a franchise to qualified franchisees, Franchisor's participation in the Franchise sale, Franchisor's legal compliance with franchise laws and regulations, site selection assistance and guidelines and the development and hosting ofinitial training programs and participation in terminating the Franchise.

8.5 New Designated Manager

After beginning operations, should Franchisee wish to request approval of a new Designated Manager, Franchisee must notify Franchisor of the identity of the new Designated Manager and the new Designated Manager must complete the initial training program to Franchisor's satisfaction within sixty (60) days of being named. Franchisor may require Franchisee to pay Franchisor's then-current rates for additional training for providing initial training to the

substitute manager. Franchisee is responsible for all travel costs, room and board and employees' salaries incurred in connection with the new Designated Manager'sattendance at such training.

8.6 Ongoing Training

From time to time, Franchisor may provide ongoing training programs or seminars during the term of this Agreement, and if it does, Franchisor has the right to require that the Franchisee and Designated Manager attend. Franchisor may charge a fee for any mandatory ongoing training. Franchisor will not require the Designated Manager to attend more than two (2) sessions in any calendar year and collectively not more than seven (7) days in any calendar year, including attendance at the Annual Convention. Franchisee will be responsible for all travel costs, room and board and employees' salaries incurred in connection with the Designated Manager, or any other employees of the Franchise Business attendance at such training.

8.7 <u>Annual Franchisee Convention</u>

During the term of this Agreement, the Franchisee and Designated Manager must attend any Convention or conference for franchisees if Franchisor holds such a convention, conference or meeting of franchisees. Franchisor may charge a fee for attendance at the Convention. The fee will be set each year by Franchisor based on the cost to hold the conference. The Convention registration fee must be paid to Franchisor by Franchisee whether Franchisee or a Designated Manager attends the convention, conference or meeting or not. This provision does not obligate Franchisor to hold an Convention of franchisees each year. If no Convention is held, Franchisee will not be obligated to pay the Annual Convention registration fee. Franchisee must be responsible for all travel costs, room and board and employees' salaries incurred in connection with Franchisee or its managers' and employees' attendance at the Convention.

8.8 <u>Training and Performance Improvement Requirements</u>

If the Franchise Business is performing unsatisfactorily (as determined by Franchisor in itssole discretion) in any of the areas listed below, or in another area that Franchisor determines to be material, Franchisor may require Franchisee, at Franchisee's expense, to: (a) attend training session specified by Franchisor; (b) visit another Franchise Business for a Performance Improvement Visit; (c) make a Performance Improvement Visit to Franchisor; (d) receive a Performance Improvement Visit from a Franchisor staff member; (e) participate in an ongoing performance improvement program; or (f) receive a Performance Improvement Visit from a third-party auditor or consultant. The following are examples of matters that are of material concern to Franchisor and/or a material breach of this Agreement, and is not an exhaustive list:

- Failing to meet the Minimum Performance Requirements;
- Suffering a below franchise system average growth rate;
- Failing to comply with paperwork procedures;
- Failing to adhere to OSHA Safety and Regulatory compliance requirements and other legal requirements relating to the Franchise Business;

- Failing to properly control risk or meet loss, safety, and other risk benchmarks specified by Franchisor;
- Failing to meet operations, signage, or truck requirements;
- Failing to meet systems benchmarks;
- Failing to maintain financials according to generally accepted accounting principles (GAAP);
- Customer satisfaction and referral rates that are below the franchisesystem averages, or other customer service issues:
- Behavior damaging to The Glass Guru brand;
- Failing to adhere to The Glass Guru core values;
- Failing to maintain accurate and timely operating system data;
- Failing to follow the marketing best practices as outlined in the guides developed by Franchisor;
- Failing to timely report financial information as required in the Manuals;
- Failing to follow sales standard operating procedures as outlined by Franchisor;
- Failing to follow the training procedures as outlined by Franchisor;
- Failing to follow customer service procedures as outlined by Franchisor; and
- Performance in multiple areas that are below expectations, none of which on its own is unsatisfactory, but the combination of which results in a determination by Franchisor that the Franchise Business is performing unsatisfactorily.

If Franchisor requires Franchisee to attend a Franchisor approved training session, Franchisee must pay all travel and living expenses, as well as a fee to cover the cost of the training session. Franchisee must attend the training session within three months of receiving notice that Franchisor requires Franchisee to attend the training session. If Franchisor requires Franchisee to undertake one of the Performance Improvement Visit options, Franchisee will be responsible for all costs and expenses associated with the visit. Franchisee must complete the Performance Improvement Visit requirement in the time period determined by Franchisor. If the Performance Improvement Visit results in an action plan to improve performance in one or more areas, Franchisee must diligently implement the action plan by the dates specified in the plan.

Nothing herein shall be construed as a waiver of Franchisor's rights to terminate this Agreement for any one or more failures listed above that we deem to be a material breach of the terms of this Agreement or the requirements of the Manuals.

9 MANUAL

9.1 Definition of Manuals

For purposes of this Agreement, The Glass Guru Manuals ("Manuals") include, but are not limited to, the manuals entitled Operational Manual & Safety Manual, and all otherwritten, electronic, video, and audio recorded policies, procedures, techniques, memos, bulletins, newsletter, forms, guides, guidelines, and other materials prepared by Franchisor in connection with the System or to assist Franchisee in the operation of the Franchise Business.

9.2 <u>Obligation to Comply with All Company Policies and Procedures; Confidentiality of Manuals</u>

To preserve and enhance the reputation and the goodwill associated with the System and

Marks and to maintain uniform standards of operations throughout the entire franchise system, Franchisee must comply with all lawful standards, policies and procedures in the Manuals that Franchisor specifies from time to time as mandatory in connection with the operation of the Franchise Business. Franchisee will be given access to the currently existing Manuals after execution of this Agreement via Franchisor's Automation Systems and/or in another manner specified by Franchisor. The Manuals remain Franchisor's confidential property must not be duplicated by Franchisee and may not be used by Franchisee other than in the operation of the Franchise Business under this Agreement.

9.3 Revisions

Franchisor has the right to add to or otherwise modify the Manual from time to time to reflect changes in the specifications, standards, operating procedures and rules prescribed by Franchisor; provided, however, that no such addition or modification materially alter Franchisee's fundamental status and rights under this Agreement. Franchisor may make such additions or modifications without prior notice to Franchisee. Franchisee must immediately, upon notice, adopt any such changes and must ensure that its copy of the Manuals is up to date at all times. If a dispute as to the contents of the Manuals arises, the terms of the master copy of the Manuals maintained by Franchisor at Franchisor's headquarters must be controlling.

9.4 <u>Confidentiality</u>

The Manuals contain Trade Secrets and other Confidential Information of Franchisor and its contents must be kept confidential by Franchisee both during the term of the Franchise and subsequent to the expiration and non-renewal or termination of this Agreement. Franchisee must always ensure that its copy of the Manuals is in a current and up-to-date manner. If the Manuals are in paper form or stored on computer-readable media, Franchisee must maintain the Manuals in a secure manner in Franchisee's office; if the Manuals are in electronic form, Franchisee must maintain the Manuals in a password-protected file. Franchisee must only grant authorized personnel, as defined in the Manuals, access to the Manuals or any key, combination or passwords needed for access to the Manuals.

Franchisee must not disclose, duplicate, or otherwise use any portion of the Manuals in an unauthorized manner.

10 FRANCHISE SYSTEM

10.1 <u>Uniformity</u>

Franchisee must comply and must cause the Franchise Business to comply, with all requirements, specifications, standards, operating procedures and rules described in this Agreement, the Manuals or other communications supplied to Franchisee by Franchisor in order to maintain uniformity within the franchise system and for the protection of Franchisor's brand and the Marks.

10.2 <u>Modification of the System</u>

Franchisee recognizes that from time to time, Franchisor may introduce, as part of the System, other methods or technologies which require certain System modifications including, without limitation, the adoption and use of modified or substitute Marks, new computer hardware and software, vehicles, equipment, or signs, the introduction of new products and services, and new specifications, standards, operating procedures relating to the promotion and marketing of the Franchise Business. Franchisee agrees to make all required upgrades and modifications at its expense as may be reasonably required by Franchisor. If such additional investment is required to be made in the last year of any term, Franchisee may avoid making the investment by providing notice of intent not to renew the Franchise unless the investment is in connection with a modification to the System required by law or court order. Franchisee acknowledges that any required expenditures for changes or upgrades to the System must be in addition to expenditures for repairs and maintenance as required in Section 13.3 of this Agreement.

10.3 Variance

Franchisor has the right to vary standards or specifications for any franchisee based uponthat franchisee's qualifications, the peculiarities of the site or circumstances, the demographics of the trade area, business potential, existing business practices or any other condition which Franchisor deems to be of importance to the successful operation of any The Glass Guru Business. Franchisor is not required to disclose or grant to Franchisee a like or similar variance under this Agreement.

11 ADVERTISING AND PROMOTIONAL ACTIVITIES

11.1 <u>Brand Identity Package</u>

Prior to beginning operation of the Franchise Business, Franchisee must purchase from Franchisor a set of brand identity items selected by Franchisor at a non-refundable feeof FIVE THOUSAND DOLLARS (\$5,000.00), to promote The Glass Guru brand and the services offered by The Glass Guru Business ("Brand Identity Package"). These items include, for example, store posters, displays, banners, trade show items, rack and other cards, and uniforms.

Brand Identity Package expenditures will be in addition to any Local Advertising expenditures and Marketing Fee.

11.2 Local Advertising

11.2.1 In addition to Brand Identity Package, (see Section 11.1) Franchisee must promote the Franchise Business. Each month, Franchisee must spend the greater of five percent (5%) of the prior year's Gross Sales divided by the months of operation in the prior year, or minimum monthly local advertising expenditure of \$1,250 ("Minimum Local Advertising Expenditure") on advertising, promotions, and public relations within the agreed upon Area of Primary Responsibility surrounding the Franchise Business ("Local

Advertising"). Franchisee's LocalAdvertising expenditures will be in addition to any Marketing Fee paid to Franchisor. Local Advertising expenditures must be made directly by Franchisee to providers of advertising services, subject to the approval and direction of Franchisor. Franchisor will provide general guidelines to Franchisee for conducting Local Advertising. Franchisee must maintain accurate records and furnish to Franchisor upon request an accurate accounting of the expenditures on Local Advertising for any prior period. Any marketing activities conducted by Franchisee outside the Area of Primary Responsibility must be approved in writing by the Franchisor, subject to all other terms and conditions of this Agreement.

- At the conclusion of each calendar month, if it is determined that a Franchise Business spent less than the Minimum Local Advertising Expenditure on Local Advertising approved by Franchisor, the Franchise Business will incur a fee ("Local Advertising Non- Compliance Fee") payable to the Franchisor. The Local Advertising Non-Compliance Fee shall be equal to the difference between the Minimum Local Advertising Expenditure required and actual approved Local Advertising expenses incurred by the Franchise Business during the prior calendar month. Payment of any Local Advertising Non-Compliance Fee will be collected along with Royalties and Marketing Fees in the next royalty period following the assessment of this fee.
- 11.2.2 Each month, upon the submission date of the prior period's Gross Sales and approved Local Advertising expenses to Franchisor, Franchisee shall provide Franchisor with a report showing any surplus or shortfall applying to Local Advertising expenses incurred by the Franchise Business. Any surplus/shortfalls will be displayed month-by-month. Franchisee should be aware of any shortfalls in Local Advertising expenses and can take the opportunity to ensure that Local Advertising expenditures are made before the end of the current month to avoid any subsequent Local Advertising Non-Compliance Fees.
- 11.2.3 Exclusive of approved material offered through our (internal) Marketing Store, Franchisee shall submit to Franchisor, for its prior approval, all advertising and promotional materials to be used by Franchisee including, but not limited to, ad copy, coupons, flyers, direct mail or digital ads. Franchisor must use reasonable efforts to provide notice of approval or disapproval within ten (10) business days from the date all requested material is received by Franchisor. If Franchisor does not approve submitted materials by the end of such ten (10) business day period, such materials must be deemed to have not received the required approval. Franchisee must not use any marketing or promotional material prior to approval by Franchisor. The submission of advertising materials to Franchisor for approval must not affect Franchisee's right to determine the prices at which Franchisee sells products or provides services.

11.3 <u>Internet Advertising</u>

Franchisee may not establish a presence on, or market using, the Internet in connection with the Franchise Business without Franchisor's prior written consent. Franchisor has established and maintains an Internet website at the uniform resource locator www.theglassguru.com that provides information about the System and the services that Franchisor and its franchisees provide. Franchisor may (but is not required to) include an independent stand-alone website containing information about the Franchise Business linked from The Glass Guru website. If Franchisor includes such information on The Glass Guru Franchise Business website, Franchisor has the right to require Franchisee to prepare all or a portion of the content, at Franchisee's expense, using a template that Franchisor provides. All such information must be subject to Franchisor's approval prior to posting. Franchisor retains the sole right to advertise or use the Marks on the Internet, including the use of websites, domain names, uniform resource locators, linking, search engines (and search engine optimization techniques), social media pages, banner ads, meta-tags, marketing, auction sites, e-commerce and co-branding arrangements. Franchisee may be requested to provide content for Franchisor's Internet marketing and must be required to follow Franchisor's intranet and Internet usage rules, policies, and requirements. Franchisor has the sole right to approve any linking to, or other use of, The Glass Guru website and The Glass Guru Franchise Business website.

11.3.1 Use of Websites, social media, and other Internet tools. Franchisee acknowledges that use of the websites, social media, such as Facebook, LinkedIn, Twitter, YouTube, blogs, vlogs (social videos), online social networks, wikis, forums, content sharing communities, etc., and other Internet tools in connection with the operation, advertising, and marketing of the Franchise Business are subject to the trademark, advertising, marketing, and other requirements of this Agreementand the Manuals. Franchisee must comply with any policies of Franchisor relating touse of websites, social media, social media advertising, and other Internet tools.

11.4 <u>Telemarketing</u>

Franchisee and any telemarketers employed or utilized by Franchisee must comply with any applicable "no call" or "do not call" and "home solicitation sales" laws or any other laws governing or affecting telemarketers to the extent Franchisee uses telemarketing to advertise and promote the Franchise Business.

11.5 <u>Required Advertising on Glass Trucks and Other Business Vehicles</u>

Each glass truck and other business vehicle must be professionally wrapped and display The Glass Guru Mark and any other of the Marks as Franchisor determines in accordance with marketing policy.

11.6 General Limitations on Advertising and Marketing

Franchisee must request written authorization from Franchisor to conduct advertising or advertise outside the Area of Primary Responsibility as described in Franchisor's policies.

11.7 Present and Future Advertising Policy May Limit Franchisee's Right to Advertise

Franchisee acknowledges that Franchisor has developed and will continue to develop advertising/marketing policies regarding the methods and manner of advertising in various media and that Franchisee is obligated to comply with all advertising policies. Franchisee understands that existing and/or future advertising policies may limit or eliminate Franchisee's right to use telephone numbers and/or Internet website addresses in advertising placed on Franchisee's vehicles and/or elsewhere. Franchisee also understands that existing and/or future policies may otherwise limit Franchisee's abilityto advertise in a particular manner. Such limitations, when established, are established for the benefit of all the franchise system's customers and/or to establish reasonable rules to govern the actions between franchisees.

11.8 Agreement to Comply with Franchisor Policy

Franchisee acknowledges and agrees that it is required to comply fully with Franchisor's advertising/marketing policies, which are set forth in writing and available to Franchiseevia the manuals or in another manner specified by Franchisor.

11.9 Agreement to Comply with Modifications and Changes to Franchisor Policy

Franchisor, in its sole discretion, reserves the right to modify or change marketing and advertising policies, and Franchisee is obligated to comply with such modified or changed policies.

11.10 Advertising Standards

Franchisee will make no misrepresentations or material omissions in any of itsadvertising Or marketing and promotional materials. All marketing, advertising, and promotional materials used by Franchisee must be factual, in good taste and not harmful to the Marks (as determined in Franchisor's sole discretion) and must conform to the highest standards of ethical advertising. Franchise must immediately cease using any advertising, marketing or promotional material that Franchisor notifies Franchisee is in violation of this Section 11.10.

11.11 Advertisements Produced by 3rd Parties

Franchisor does not, by virtue of its approval of any proposed advertisement or promotional material produced by 3rd parties, assume any responsibility for the contents of the advertisement. Franchisee agrees to indemnify and save harmless Franchisor from any claims, demands, liability, costs, and expenses that Franchisee suffers arising from the use of any such advertisement or promotional material.

11.12 Franchisor Remedies for Displaying Unapproved or Unauthorized Advertising

Except in the case of a minor violation that can be immediately cured (as determined In Franchisor's sole discretion) Franchisor may seek remedies, including injunctive reliefbarring

Franchisee from its ongoing advertising violations, assignment of Franchisee's telephone numbers to Franchisor, or other relief, up to and including termination of the Franchise Agreement.

12 ACCOUNTING RECORDS AND REPORTING OBLIGATIONS

12.1 Records

During the term of this Agreement, Franchisee must maintain full, complete, and accurate books, records, and accounts in accordance with the standard accounting system prescribed by Franchisor in the Manuals or otherwise in writing, and in accordance with generally accepted accounting principles. Franchisee must retain during the term of this Agreement, and for three (3) years thereafter, all books and records related to the Franchise Business including, without limitation, purchase orders, invoices, payroll records, sales tax records, state and federal tax returns, bank statements, cancelled checks, deposit receipts, cash receipts and disbursement journals, general ledgers, and any other financial records designated by Franchisor or required by law. In efforts to facilitate proper record-keeping between the software systems, in conjunction with entering into the Franchise Agreement, you must use an Approved Supplier, to facilitate your business book-keeping for the first three (3) full months of operation.

12.1.1 Providing Business Information to Franchisor; Customer Lists

Franchisee will supply Franchisor with such records, reports, and other information aboutFranchisee and the Franchise Business (in addition to that otherwise provided for in this Agreement) as Franchisor may require. It is hereby agreed and understood that the customer lists of the Franchise Business are and will remain Franchisor's property. Franchisee acknowledges and agrees that the records and reports that Franchisee maybe required to provide to Franchisor may not be uniformly imposed on all franchisees. Differences in required records and reports may be based on the Franchisee's experience, the demographics of the Area of Primary Responsibility, the density of thepopulation, and other reasonable factors.

12.2 Monthly Reports

Franchisee must provide to Franchisor monthly reports pertaining to Franchisee's Gross Sales and such other additional information specified by Franchisor in monthly report forms or formats specified by Franchisor in the Manuals (currently provided to Franchisor through GPS Insight software system). Monthly Royalty Reports for each calendar month must be provided on or before the fifth (5th) day following the end of that calendar month. Franchisee must provide additional monthly reports specified by Franchisor in the Manuals (e.g., routine month-end close financial entries in line with generally accepted accounting principles) on or before the twentieth (20th) day following the end of that calendar month. These reports must be provided in the manner specified by Franchisor (which may include Franchisor directly accessing the

information on the software systems). Franchisor can share information in these reports with other franchisees in the ordinary course of Franchisor's business as a tool to improve the franchise system's volume of business.

12.2.1 Separate Reporting for each Franchise Business

If Franchisee operates more than one Franchise Business location, Franchisee must provide separate accounting reports for each Franchise Business location. This includes monthly reports of Gross Sales, balance sheets and income (profit and loss) statements as well as any other accounting reports specified elsewhere in this Agreement.

12.2.2 Consequences of Failure to Timely and Accurately Report

If Franchisee fails to transmit reports, records, or other information to Franchisor in the manner, at the times, and in the formats specified by Franchisor, or fails to timely close jobs and process open jobs or fails to provide accurate information, Franchisee will incur a charge of \$50 for each calendar day that data is not transmitted. The charge is intended to cover expenses Franchisor incurs to collect this data. If a documented technology failure prevents electronic transmittal of the data, or some other bona fide emergency occurs preventing electronic transmittal of the data (as Franchisor determines in its sole discretion), the charge will not be imposed.

12.2.3 Consequences of Improper Reporting

If Franchisee fails to submit the monthly Gross Sales report in an improper or unapproved format or fails to submit the Gross Revenue report when due to the Franchisor, Franchisee will incur a charge of \$75 for each occurrence. If a documented technology failure prevents electronic transmittal of the data, or some other bona fide emergency occurs preventing electronic transmittal of

the data (as Franchisor determines in its sole discretion), the liquidated damages charge will not be imposed.

12.3 Financial Statements

- 12.3.1 Franchisee must, at its expense, submit to Franchisor within ninety (90) days after the end of each calendar year, an income statement for the calendar year just ended and a balance sheet as of the last day of the calendar year. Such financial statements must be prepared in accordance with generally accepted accounting principles applied on a consistent basis. If required by Franchisor, such financial statements must be reviewed or audited by a certified public accountant.
- 12.3.2 Franchisor also reserves the right that Franchisee supply to Franchisor, upon Franchisor's request, on or before the tenth (10th) day of each month, in a form approved by Franchisor, a balance sheet as of the end ofthe last day of

the preceding month and an income statement for the preceding month and the fiscal year-to-date.

12.3.3 Franchisee must submit to Franchisor such other periodic reports in the manner and at the time specified in the Manuals or otherwise in writing.

12.4 Other Reports

Franchisee must submit to Franchisor copies of all state sales tax returns that are required to be filed with the appropriate governmental agency and such other records as Franchisor may reasonably request from time to time or as specified in the Manual. Franchisor has the right to release financial and operational information relating to the Franchise Business to Franchisor's lenders or prospective lenders, or to use or disclose such information in an a financial performance representation created for the Franchisor's Franchise Disclosure Document. Franchisee must certify as true and correct all reports to be submitted under this Agreement.

12.4.1 Operational Information

Franchisee agrees to provide to Franchisor, as and when requested, operational information relating to the Franchise Business, whether of a financial nature or otherwise, including but not limited to: business organizational charts; corporate or company organizational charts; information on all management employees; federal and state operating authorities, and all changes and additions to the operating authorities.

12.5 Computer, Point-of-Sale, and Accounting Systems

Franchisee must purchase, install, and use computer, point-of-sale, and accounting systems consisting of hardware and software that is in accordance with Franchisor's specifications, including, but not limited to software installed locally at the Franchise Business location, cloud-based business applications, or other proprietary software. Franchisee is responsible for any and all subscription fees, license fees or other fees charged for obtaining and using the required systems.

Franchisor must have full access to all of Franchisee's computer, point-of-sale, and accounting systems and all related data. This information must be available by means of direct access, either in person or by telephone, modem, or Internet, as to permit Franchisor to verify Franchisee's compliance with its obligations under this Agreement.

Within forty-eight (48) hours from initial customer contact, Franchisee is required to enter customer contact information – e.g., including full name, address, phone number and e-mail address and other information Franchisor specifies in the computer system for the Franchise Business. Should Franchisee fail to enter customer information during three (3) or more occasions (three consecutive or non-consecutive months) during any calendar year, franchisee may be required to obtain and present to Franchisor a year-end audited financial statement for the year the breach occurred. The audited financial statements, if required, are to be prepared by an independent account firm and presented within ninety (90) days post year-end. Franchisor may charge an administrative fee currently in the amount of One

Hundred Seventy-Five Dollars (\$175.00) for each month in which Franchisee fails to enter customer data by the due date.

12.6 Right to Inspect

Franchisor, itself or through its designee, has the right, at any time during the term of this Agreement or upon termination or expiration of this Agreement, to inspect, examine, copy, and audit, the books, records. and tax returns of Franchisee's operation of Franchise Business, as well as tax returns for any person or entity holding a material interest in the Franchise Business. This may be accomplished by entering upon the premises of the Franchise Business during business hours, by requiring Franchisee to electronically provide require records or in any other manner designated by Franchisor.

Franchisee must cooperate fully with such audit and inspection. If the audit or any other inspection should reveal that any payments to Franchisor have been underpaid, then Franchisee must immediately pay to Franchisor the amount of the underpayment plus interest from the date such amount was due until paid at the rate of eighteen percent (18%) per annum (or the highest rate allowed by the law of the state where Franchisee islocated, whichever is lower). If the inspection discloses an underpayment of three percent (3%) or more of the amount due for any period covered by the audit, or if the inspection is conducted due to Franchisee's failure to submit required reports and records, Franchisee must, in addition, reimburse Franchisor for all costs and expenses connected with the inspection (including, without limitation, travel expenses and reasonable accounting and attorneys' fees). The foregoing remedies must be in addition to any other remedies Franchisor may have.

12.7 Release of Records

At Franchisor's request, Franchisee must authorize and direct any third parties, including accounting professionals and suppliers to release to Franchisor all accounting and financial records arising from or relating to the operation of the Franchise Business including, but not limited to, records evidencing Gross Sales, profits, losses, income, tax liabilities, tax payments, revenues, expenses, and any correspondence, notes, memoranda, audits, business records, or internal accounts within said third parties' possession, custody or control, and to continue to release such records to Franchisor ona monthly basis for the length of the unexpired term of this Agreement or until such time as Franchisor withdraws its request. Franchisee must execute all documents necessary to facilitate the release of records referenced here to Franchisor.

12.8 Chart of Accounts; Central Accounting Office

Franchisee must use the standard "chart of accounts" as specified by Franchisor and must have its chart of accounts approved by Franchisor before Franchisee begins operation of the Franchise Business. If Franchisee and its affiliates operate multiple Franchise Businesses and have a central accounting office for those Franchise Businesses, Franchisor may require Franchisee and its affiliates to store all business records at the central accounting office.

13 STANDARDS OF OPERATION

13.1 <u>Authorized Products, Services and Suppliers</u>

- 13.1.1 Franchisee acknowledges that the reputation and goodwill of the System is based on offering high quality services and products to its customers. Accordingly, Franchisee must provide or offer for sale or use at the Franchise Business only those products, supplies, signs, equipment, tools, components and other items and services that Franchisor from time to time approves (and which are not thereafter disapproved) and that comply with Franchisor's specifications and quality standards. If required by Franchisor, any such items or services must be purchased only from "Approved Suppliers" that Franchisor designates or approves (which mightinclude, or be limited to, Franchisor). Franchisee must not offer for sale, sell, or provide through the Franchise Business any services or products that Franchisor has not approved or has disapproved.
- 13.1.2 Franchisor shall provide Franchisee, in the Manuals or other written or electronic form, with a list of specifications and, if required, a list of Approved Suppliers for some or all the products, supplies, signs, equipment, tools, components and other approved or specified items and services, and Franchisor may from time-to-time issue revisions to such list. If Franchisor is an Approved Supplier of products or supplies, Franchisee must execute a standard form purchase or supply agreement for the items to be supplied by Franchisor. If Franchisee desires to utilize any services or products that Franchisor has not approved (for services and products that require supplier approval), Franchisee must first send Franchisor at its expense sufficient information, specifications, and samples (if required) for Franchisor to determine whether the service or product complies with its standards and specifications or whether the supplier meets its Approved Supplier criteria. Franchisor will decide within a reasonable time (usually thirty [30] days) after receiving the required information whether Franchisee may purchase or lease such items or services or from such supplier. Approval of a supplier may be conditioned on the supplier's ability to provide sufficient quantity of product; quality of products or services at competitive prices; production and delivery capability; anddependability and general reputation. Nothing in this Section must be construed to require Franchisor to approve any supplier, or to require Franchisor to make available to prospective suppliers, standards, and specifications Franchisor confidential. that deems
- 13.1.3 Notwithstanding anything contrary in this Agreement, Franchisor has the right to review from time to time its approval of any items or suppliers. Franchisor may revoke its approval of any item, service, or supplier at anytime by notifying Franchisee or the supplier. Franchisee must, at its own expense, promptly cease using, selling, or providing any items or services disapproved by Franchisor and must promptly cease purchasing from suppliers disapproved by Franchisor.
- 13.1.4 Franchisor has the right to designate certain products and services, not otherwise authorized for general use as part of the System, to be offered

locally or regionally based upon such factors as Franchisor determines including, but not limited to, franchisee qualifications, test marketing and regional or local differences. Franchisor has the right to allow one (1) or more franchisees to provide certain products or services not authorized forgeneral use as part of the System. Such consent will be based upon the factors described in Section 10.3 and must not create any rights in Franchisee to provide the same products or services.

13.1.5 While Franchisor has the right to retain volume rebates, markups, and other benefits from suppliers or in connection with the furnishing, approval or specification of suppliers, Franchisor may pass on a portion of these benefits to franchisees. Franchisor must keep a portion of any such benefits to compensate it for the administrative costs of the purchasing program.

13.2 <u>The Glass Guru Restoration Tools and Products</u>

- 13.2.1 Franchisor and its Affiliate have designed, developed and/or acquired and continue to design and develop certain window restoration tools, equipment and products, and other items especially suited for use in the operation of The Glass Guru Business ("The Glass Guru Restoration Tools"). In order to maintain the consistency, quality and uniformity of the System, Franchisor shall make The Glass Guru Restoration Tools available to Franchisee in reasonable quantities in accordance with the procedures for ordering, handling, and shipping that Franchisor may determine periodically, provided that Franchisee is in compliance with the Franchise Agreement and all other agreements with Franchisor.
- 13.2.2 Franchisee acknowledges and agrees that The Glass Guru RestorationTools developed and/or acquired by Franchisor and its Affiliate are distinctive as a result of being developed under Franchisor and its Affiliate's experience and are inextricably interrelated with the Marks. Franchisee agrees to order and purchase all its requirements of The Glass Guru Restoration Tools exclusively from Franchisor or a supplier designated by Franchisor. Franchisee agrees to, always, maintain an inventory of The Glass Guru Restoration Tools as necessary to operate the Franchise Business at full capacity.
- 13.2.3 Franchisor commits to provide The Glass Guru Restoration Tools supplied by Franchisor at competitive prices; however, Franchisee acknowledges that Franchisor has the right to earn a reasonable profit on the sale of its The Glass Guru Restoration Tools.

13.3 Appearance and Condition of the Franchise Business

Franchisee must maintain the Franchise Business, and the vehicle, equipment and signage used in connection with the Franchise Business in an attractive and safe condition and in good maintenance and repair and in compliance with the standards specified by Franchisor

in the Manuals or otherwise, and as necessary to comply with health and safety standards and any applicable laws or regulations. The expense of such maintenance must be borne by Franchisee and must be in addition to any required System modifications, as described in Section 10.2. If at any time, in Franchisor's sole discretion, the general state of repair, appearance, or cleanliness of the premises of the Franchise Business or the vehicles, equipment, fixtures, or signs of the Franchise Business does not meet Franchisor's standards, Franchisor may notify Franchisee in writing, specifying the action to be taken by Franchisee to correct the deficiency. Franchisee must initiate the specified action within thirty (30) days after receipt of the notice and diligently proceed to complete the specified action. If Franchisee fails to do so, then Franchisor will have the right, in addition to its other rights under this Agreement, to enter the Franchise Businessand cause the specified action to be taken on behalf of Franchisee and Franchisee must pay the entire cost to Franchisor.

13.4 Ownership and Management

The Franchise Business must always be under the direct supervision of Franchisee. The Designated Manager must devote sufficient efforts to the management of the day-to- day operation of the Franchise Business, but not less than thirty (30) hours per week, excluding vacation, sick leave and similar absences. Franchisee must keep Franchisor informed, in writing, at all times of the identity of its Designated Manager. Franchisee must not engage in any business or other activities that will conflict with its obligations under this Agreement.

13.5 <u>Obligation to Personally Supervise and Manage Day-to-Day Operations</u>

Franchisee's Designated Manger approved in writing by Franchisor, must always personally supervise the day–to–day operation of the Franchise Business and always personally exercise his or her best efforts to market the products and services of the Franchise Business. Franchisor reserves the right to approve any Designated Manager to whom Franchisee delegates any substantial portion of this responsibility. Franchisor has the right to require the designated manager to successfully complete Franchisor's training program as a condition to approval of the Designated Manager.

13.6 <u>Days of Operation</u>

Franchisee must keep the Franchise Business open for business during normal business hours on the days specified in the Manuals.

13.7 Franchise Business Staff

The Franchise Business must be staffed with sufficient personnel to provide optimumservices as specified in the Manuals.

13.8 Only Franchisee Has the Right to Control Employees

Franchisor does not control, and does not have the right to control, Franchisee's decisions regarding recruiting, hiring, disciplining, supervision, terminating, compensation, benefits, work schedules, work rules, or recordkeeping regarding Franchisee's employees or agents. Franchisor does not control or have the right to control Franchisees other day-to-day

business activities. Franchisee shall notify and communicate clearly with its employees in all dealings, including without limitation, its employment applications, written and electronic correspondence, paychecks, employee handbooks, employment policies and procedures, and other written materials that Franchisee (and only Franchisee) is their employer, and that Franchisor is not their employer.

Franchisee shall comply with any state workers compensation act, any state unemployment compensation benefit law or regulation, or any other federal, state, or local employment or employee benefit law and regulation, and shall establish employeraccounts as required by applicable federal and/or state law. Any required employee training and any mandatory and suggested standards, specifications, policies, and procedures of Franchisor relating to Franchisee's employees or employment matter are imposed not for the purpose of exercising control over Franchisee's or the Franchise Business but rather for the limited purpose of protecting Franchisor's Marks, system, confidential Information, goodwill and brand consistency. Franchisee must indemnify and hold harmless Franchisor from and against any liability relating to or arising from employment related decisions and obligations, including but not limited to labor and employment law violations by Franchisee and Franchisee's employees and claims that Franchisor is an employer or joint employer of Franchisee's Employees

13.9 Contributions and Donations

In order to protect the Marks, Franchisee must obtain Franchisor's prior written consent before making any contributions or donations of items, services or funds to any individual or entity, or provide any type of other benefit to any charitable, religious, political, social, civic or other type of organization (or to any individual on behalf of any organization).

13.10 Licenses and Permits; Compliance with Laws

Franchisee must secure and maintain in force all required licenses, permits and certificates necessary for the operation of the Franchise Business. Franchisee must operate the Franchise Business in full compliance with all applicable federal, state, county, municipal or other statutes, laws, ordinances and regulations, rules or orders applicable to the Franchise Business, including but not limited to state and federal labor and employmentlaws, such as the Fair Labor Standards Act (FLSA), Family and Medical Leave Act (FMLA), Occupational Safety and Health Act (OSHA), Employee Retirement Income Security Act (ERISA), Title VII, the Age Discrimination in Employment Act, and the Affordable Care Act and laws and regulations relating to worker's compensation insurance, unemployment insurance, and withholding and payment of federal and state income taxes, sales, and other taxes. Franchisee must promptly pay all payroll and business taxes, fees and expenses, and all other amounts required by law. Franchisor makes no representation to Franchisee with regard to any legal requirements that Franchisee must satisfy or comply with in connection with the operation of the Franchise Business. Franchisee must be solely responsible for investigating and complying with all such laws, ordinances, and regulations with regard to the operation of the Franchise Business.

13.11 Notification of Proceedings

Franchisee must notify Franchisor in writing of the commencement of any action, suit or proceeding involving Franchisee or the Franchise Business, and of the issuance of any order, writ, injunction, judgment, award, or decree which may affect the operation or financial condition of the Franchise Business not more than five (5) days after Franchisee receives notice of any such commencement or issuance. Franchisee must deliver to Franchisor not more than five (5) days after Franchisee's receipt of that notice, a copy of any inspection report, warning, certificate, or rating by any governmental agency relatingto any health or safety law, rule or regulation that reflects Franchisee's failure to meet and maintain the highest applicable rating or Franchisee's noncompliance or less than full compliance with any applicable law, rule, or regulation.

13.12 <u>Compliance with Good Business Practices</u>

Franchisee acknowledges that the quality of customer service, and every detail of appearance and demeanor of Franchisee and the Franchise Business, is material to this Agreement and the relationship created. Therefore, Franchisee must endeavor to maintain high standards of quality and service in the operation of the Franchise Business. Franchisee must always give prompt, courteous, and efficient service to customers of the Franchise Business. The Franchise Business must in all dealings with its customers, vendors, and the general public, adhere to the highest standards of honesty, fair dealing,

and ethical conduct.

13.13 <u>Obligation to be Governed by Highest Ethical Standards and to Comply with the Systems</u>

Franchisee acknowledges that every component of the System is important to Franchisorand to the operation of the Franchise Business. Franchisee must always operate the Franchise Business in a competent manner and in full compliance with all aspects of the System specified by Franchisor. In all business dealings with the public and with Franchisor, Franchisee will be governed by the highest standards of honesty, integrity, fair dealing, and ethical conduct and act always to support and grow the System. Franchisee must not engage in any activity or practice that results in or may reasonablybe anticipated to result in damage to Franchisor's business reputation or result in or reasonably be anticipated to result in any public criticism of the System or Marks. Franchisee will not use or engage any federal, state, or local law, regulation, court, or tribunal to retard or prevent another franchisee or prospective franchisee of the System from obtaining a license or authority to operate as a full-service window, door, and glassbusiness or in any other capacity authorized by this Agreement. Franchisee acknowledges that such violations will be good cause for immediate termination of this Agreement.

13.14 <u>Disputes Arising with Customers and Third Parties</u>

If Franchisor becomes aware of a bona fide dispute between Franchisee and one or more customers and/or other third parties regarding the Franchise Business, Franchisor may, inits sole discretion, undertake one or more of the following options:

take no action, except to direct the Franchisee to resolve the dispute in a manner that will not cause injury to the reputation of the Marks and The Glass Guru franchise system;

assist the parties in the resolution of the dispute, if Franchisor in its sole discretion, determines that it can constructively do so; and/or if Franchisor determines in its sole discretion that Franchisee cannot or will not resolve the dispute, and that such failure to resolve it has or is reasonably likely to cause damage to the Marks and/or The Glass Guru franchise system's business reputation, then upon notice to Franchisee, Franchisor may resolve the dispute directly with the third-party by payment of damages alleged and supported by documentary evidence by the third-party, including attorney's fees, and Franchisee agrees to reimburse and indemnify Franchisor for all such payments.

If Franchisor pays such damages to a third-party, Franchisor will invoice Franchisee for the damages paid, and payment from Franchisee will be due to Franchisor within fourteen (14) days from the date of invoice. In Franchisor's sole discretion, it may consult a designated franchisee group to provide it with an advisory opinion regarding resolution of the dispute, although Franchisor will not be obligated to comply with the advice of the designated franchisee group. If Franchisor consults with a designated franchisee group, it will provide the designated group with the facts and circumstances of the dispute, but Franchisor will not provide it with the identity of any of the parties to the dispute.

If Franchisor deems that Franchisee did not fairly handle a customer complaint, Franchisor has the right, but not the obligation, to intervene and satisfy the customer. Franchisor has the right to terminate this Agreement for violation of this Section. Franchisee must reimburse Franchisor for all costs incurred by Franchisor in servicing a customer of the Franchise Business under this Section.

13.15 Uniforms

For purposes of promoting the Marks and brand, Franchisee must abide by any uniform requirements stated in the Manual. Uniforms may be purchased from Franchisor or other Approved Supplier.

13.16 Credit Cards

Franchisee must, at its expense, lease or purchase the necessary equipment or software and must have arrangements in place with Visa, MasterCard, Discover, American Expressor such other credit card issuers as Franchisor may designate, from time to time, to enable the Franchise Business to accept such methods of payment from its customers. Franchisee must use such credit card processing and clearing house services as Franchisor specifies in the Manual.

13.16.1 Credit Card Payment; PCI Compliance.

Franchisee must make available at its Franchise Business, credit card services that enable all customers to pay for approved services with a valid credit card, so long as the credit card has sufficient credit to cover payment of the approved services. Franchisee must not charge an additional fee or a different price for products or services if the customer payswith a valid credit card. Franchisee must comply with the Payment Card Industry ("PCI") Data Security Standard ("DSS") Requirements and Security Assessment Procedures and other applicable PCI requirements ("PCI Requirements") in connection with the Franchise Business. It is

Franchisee's responsibility to research and understand the PCI Requirements and to ensure that its business policies and practices comply with the PCI Requirements. Although Franchisor may provide advice and/or specify or provide POS Systems or business software, Franchisor does not represent or warrant that those systems or software comply with the PCI Requirements and it will be the sole responsibility of Franchisee to ensure that its business practices comply with the PCI Requirements.

13.17 E-Mail

Franchisor will provide Franchisee with e-mail addresses for Franchise Business purposesas expressed in the Manual. Franchisee may not change its e-mail addresses unless approved in writing by Franchisor.

13.17.1 Centralized Email

If specified by Franchisor, Franchisee must use a centralized email system maintained by Franchisor. Currently, centralized email use is provided for through the aforementioned Technology Fee.

13.18 Participation in Call Center or Answering Service

Franchisor may provide or have a third-party provide call center or answering services for its franchisees, which may include responding to afterhours phone calls and emails and/or responding to calls and emails during business hours in certain circumstances or other similar services. Franchisor may require Franchisee to participate in these services and to pay a proportionate share of the costs of these services or a reasonable charge for these services.

13.19 Best Efforts

Franchisee must use its best efforts to promote and increase the sales and recognition of services offered through the Franchise Business. Franchisee must require all of Franchisee's employees, managers, officers, agents, and representatives to make a good faith effort to enhance and improve the System and the sales of all services and products provided as part of the System .

13.20 Automation Systems

Franchisee must use the Automation Systems specified by Franchisor in the operation of the Franchise Business (the "Automation Systems"). The Automation Systems may include proprietary computer systems (including specified hardware and software), accounting applications, credit card systems, learning management system (LMS), the Extranet, marketing automation system, mobile technology solutions, unified communications system, online training programs, telephone systems, call center systems, email, Internet access and other communication methods, secure websites, networks, and other or different components that may be designated by Franchisor. As acondition to using the Automation Systems, Franchisee must agree to comply with the terms of use specified by Franchisor.

Franchisee must keep accurate Automation Systems user accounts for its employees and must either notify Franchisor or make any necessary updates in the Automation Systemsof any user account changes or employee status changes within two business days.

Franchisee must employ other adequate measures to secure the Automation Systems and the information contained in the Automation Systems, as specified by Franchisor orany applicable Designated or Approved Supplier.

If Franchisor has not yet specified a particular system and/or Designated or Approved Supplier of a system as part of the required Automation Systems, Franchisee must obtain approval from Franchisor before obtaining the system or transitioning to a new system or Designated or Approved Supplier of the system. Current examples of systems in this category are telephone systems and call center systems. If Franchisor specifies these or other systems as part of the Automation Systems in the future, Franchisee must use the systems, and/or Designated or Approved Suppliers specified by Franchisor.

13.21 <u>Computer Systems</u>

Franchisee must purchase, install, and use in operating the Franchise Business. In addition, Franchisor may develop computer and point-of-sale systems consisting of hardware, software and other technology that is in accordance with Franchisor's specifications, including, but not limited to software installed locally at the Franchise Business location, could-based business applications, or proprietary software developed by Franchisor. If Franchisor introduces software designed exclusively for or modified for the System ("Proprietary Software"), Franchisee must use Proprietary Software in the operation of the Franchise Business and comply with all specifications and standards prescribed by Franchisor regarding the Proprietary Software, as provided from time to time in the Manuals, including but not limited to execution of a software license agreement and payment of license fees. Upgrades to the Proprietary Software may be implemented into the System at Franchisor's discretion. Once developed, Franchisee will be responsible for maintaining on-going services and support regarding the Proprietary Software, and Franchisor or a third party will license the Proprietary Software to Franchisee at the then current published rates.

In addition, Franchisor may develop computer systems and specifications for certain components of the computer system in the future and may modify such specifications and the modification of the specifications for the components of the computer systems and software may require Franchisee to incur costs to purchase, lease, and/or license new or modified computer hardware and/or software and to obtain service and support for the computer systems during the term of this Agreement. All such computer systems must be compatible with Franchisor's computer systems as modified from time to time, must be connected to Franchisor's facilities by high-speed Internet, and must be updated, maintained, and used in compliance with Franchisor's specifications. Franchisor may require Franchisee to electronically upload or transmit information on a periodic basis (including daily).

13.22 Franchisor Access to and Use of Information

Franchisor will have the right to independently access sales information, customer information, reports and other data produced by or stored in Franchisee's computer system, software or the Automation Systems and there are no contractual limitations on Franchisor's

right to access and use that information and data, even if the data is maintained by a third party. Franchisee must provide Franchisor access to the computer system, software, and Automation Systems information on the Automation Systems in the manner specified by Franchisor and must supply Franchisor with all security codes necessary to obtain such access. Franchisee agrees that Franchisor will not be liable to Franchisee for any claims, losses, or damages arising from or related to Franchisor's access to or use of the information and other data produced by computer systems, software or the Automation Systems, including but not limited to any errors or omissions in the information and other data obtained by Franchisor or in the informationand other data shared by Franchisor with third parties (including other franchisees or prospective franchisees). Franchisee waives and releases Franchisor from any such liability.

13.23 <u>Certain Telephone Technology Prohibited</u>

Franchisee is not permitted to use "roll over" or "hunt" telephone line system technology where Franchisee and/or its Affiliates operate multiple Franchise Businesses without the written consent of Franchisor.

14 FRANCHISOR'S ADDITIONAL OPERATIONS ASSISTANCE

14.1 General Advice and Guidance

Franchisor shall make a reasonable effort to discuss problems and offer general guidance to Franchisee by telephone, e-mail, intranet, or other methods with respect to planning, and operating the Franchise Business. However, Franchisee shall use its independent judgment to make all business decisions and should not rely solely upon any advice given or statements made by Franchisor. Franchisor must not charge for thisservice; however, Franchisor retains the right to refuse or charge a fee for this service should Franchisor deem Franchisee to be utilizing this service too frequently or in an unintended manner. Franchisor's advice or guidance to Franchisee relative to prices for products and services that, in Franchisor's judgment, constitutes good business practiceis based upon the experience of Franchisor and its franchisees in operating The Glass Guru Business and an analysis of costs and prices charged for competitive products and services. However, Franchisee has the sole right to determine the prices to be charged by the Franchise Business.

14.2 Periodic Visits

Franchisor or Franchisor's representative may, but is not required to, make periodic visits, which may be announced or unannounced, to the Franchise Business for the purposes of inspection, consultation, assistance, or guidance with respect to various aspects of the operation and management of the Franchise Business. Franchisor and Franchisor's representatives who visit the Franchise Business may prepare, for the benefit of both Franchisor and Franchisee, written reports detailing any problems or concerns discovered during any such visit and outlining any required or suggested changes or improvements in the operations of the Franchise Business. A copy of any such written report may be provided to Franchisee. Franchisee must implement any required changesor improvements in a timely manner.

14.3 System Improvements

Franchisor shall communicate improvements in the System to Franchisee as such improvements are developed or acquired by Franchisor and implemented as part of the System.

14.4 <u>Marketing and Promotional Materials</u>

Franchisor may periodically provide advertising and promotional materials including adslicks, brochures, fliers and other materials or logoed items to Franchisee for use in the operation or promotion of the Franchise Business.

15 INSURANCE

15.1 <u>Types and Amounts of Coverage</u>

At its sole expense, prior to initial training, but no later than ninety (90) days of the Effective Date, and in no event later than the commencement of operations Franchisee must procure and maintain in full force and effect during the term of this Agreement, thetypes of insurance listed below. All policies (except any workers' compensation insurance) must expressly name Franchisor as an additional insured or loss payee and all must contain a waiver of all subrogation rights against Franchisor and its successors and assigns. In addition to any other insurance that may be required by applicable law, or bylender or lessor, Franchisee must procure:

- 15.1.1 "Special Form" (all risk) property insurance coverage on all assets including inventory, furniture, fixtures, equipment, supplies, and leasehold improvements and betterments and other property used in the operation of the Franchise Business at full replacement cost; Franchisor must be listed as loss payee;
- 15.1.2 Workers' compensation insurance that complies with the statutory requirements of the state in which the Franchise Business is located and employer liability coverage with a minimum limit of ONE HUNDRED THOUSAND DOLLARS (\$100,000.00) or, if higher, the statutory minimum limit as required by state law; A waiver of subrogation endorsement in favor of The Glass Guru Enterprises, Inc. is required. Endorsement must be provided;
- 15.1.3 Comprehensive general liability insurance against claims for bodily and personal injury, death and property damage caused by, or occurring in conjunction with, the operation of the Franchise Business, or Franchisee's conduct of business under this Agreement, with a minimum liability coverage of ONE MILLION DOLLARS (\$1,000,000.00) per occurrence or higher if required by landlord or other lease agreements; The Glass Guru Enterprises,

Inc. is to be named as additional insured with waiver of subrogation. Endorsements must be provided;

Automobile liability insurance for company owned, leased, or hired vehicles, if applicable, with a combined single limit of at least ONE MILLION DOLLARS (\$1,000,000.00); and The Glass Guru Enterprises, Inc. isto be named as additional insured with waiver of subrogation. Endorsements must be provided;

Such insurance as necessary to provide coverage under the indemnity provisions described in Section 15.1.

15.2 Future Increases

Franchisor has the right to reasonably increase the minimum liability protection requirement annually and require different or additional insurance coverage(s) to reflectinflation, changes in standards of liability, future damage awards or other relevant changes in circumstances.

15.3 <u>Carrier Standards</u>

Such policies must be written by an insurance company approved by the state insurance department in the state in which Franchisee operates and having at least an "A" Rating Classification as indicated in the latest issue of <u>A.M. Best's Key Rating Guide</u>. Although A.M. Best groups "A" and "A-" in the same classification, Franchisor demands an "A" rating.

15.4 <u>Evidence of Coverage</u>

Franchisee's obligation to obtain and maintain the foregoing policies must not be limited in any way by reason of any insurance that may be maintained by Franchisor, nor must Franchisee's performance of this obligation relieve it of liability under the indemnity provisions described in Section 21.3. Franchisee must provide, annually, certificates of Insurance showing compliance with the foregoing requirements. Such certificates must state that said policy or policies must not be canceled or altered without at least thirty (30) days prior written notice to Franchisor.

15.5 <u>Failure to Maintain Coverage</u>

Should Franchisee not procure and maintain insurance coverage as required by this Agreement, Franchisor has the right (but not the obligation) to immediately procure such insurance coverage and to charge the premiums to Franchisee, which charges, together with a reasonable fee for expenses incurred by Franchisor in connection withsuch procurement, must be payable by Franchisee immediately upon notice.

16 DEFAULT AND TERMINATION

16.1 <u>Termination by Franchisee</u>

If Franchisee is in substantial compliance with this Agreement and Franchisor materially breaches this Agreement and fails to commence reasonable efforts to cure such breachwithin thirty (30) days after receiving written notice identifying the claimed breach, Franchisee has the right to terminate this Agreement unless the breach cannot reasonably be cured within such thirty (30) days. If the breach cannot reasonably be cured in such thirty (30) days, Franchisee has the right to terminate this Agreement onlyif Franchisor does not promptly undertake and continue efforts to cure such material breach within a reasonable period of time and furnish Franchisee reasonable proof of such efforts.

16.2 <u>Termination by Franchisor</u>

- 16.2.1 Franchisor has the right to terminate this Agreement, without anyopportunity to cure by Franchisee, if Franchisee:
 - 16.2.1.1 fails to timely select an approved site for or establish, equip and commence operations of the Franchise Business under Section 5.
 - 16.2.1.2 fails to have its Franchisor approved Designated Manager satisfactorily complete the initial training program under Section 8.1;
 - 16.2.1.3 made any material misrepresentation or omission in its application for the Franchise, or otherwise to Franchisor in the course of entering into this Agreement;
 - 16.2.1.4 is convicted of or pleads no contest to a felony or other crime or offense that islikely to adversely affect the reputation of Franchisor, Franchisee, or the Franchise Business;
 - 16.2.1.5 discloses, duplicates or otherwise uses in an unauthorized manner any portion of the Manuals or any Trade Secret or other Confidential Information;
 - if required by Franchisor, and allowable under applicable state law, fails to have any holder of a legal or beneficial interestin Franchisee, and any officer, director, executive, manager or member of the professional staff and all employees of Franchisee, execute a nondisclosure and noncompetition agreement, in a form the same as or similar to the Nondisclosure and Non-Competition Agreement attached as Exhibit 3, upon the signing of this Agreement or prior to each such person's affiliation with Franchisee or fails to provide Franchisor with copies of all nondisclosure and non-competition agreements signed under Section 7.4, if requested by Franchisor;

- 16.2.1.7 abandons, fails, or refuses to actively operate the Franchise Business for five (5) or more consecutive days (unless the Franchise Business has not been operational for a purpose approved by Franchisor);
- 16.2.1.8 if Franchisor and Franchisee do not agree upon a substitute site for relation within ninety (90) days after the lease expires or is terminated or the Approved Location is rendered unusable; or Franchisee fails to promptly relocate the Franchise Business once approval is given by Franchisor for the Approved Location;
- 16.2.1.9 surrenders or transfers control of the operation of the Franchise Business, makes or attempts to make an unauthorized direct or indirect assignment of the Franchise or an ownership interest in Franchisee;
- 16.2.1.10 fails to or refuses to assign the Franchise or the interest in Franchisee of a deceased or incapacitated owner; within one hundred eighty (180) days following the death or Incapacity of Franchisee or any holder of a legal or beneficial interest as required by Section 18.6, or fails to maintain the Franchise Business under the primary supervision of a Franchisor approved Designated Manager until assignment of the Franchisee or the interest in Franchisee:
- adjudicated as bankrupt, becomes insolvent, commits any affirmative act ofinsolvency, or files any action or petition of insolvency; if a receiver of its property or any part of that is appointed by a court; if it makes a general assignment for the benefit of its creditors; if a final judgment remains unsatisfied of record for thirty (30) days or longer (unless supersedeas bond is filed); if execution is levied against Franchisee's business or property; if a suit toforeclose any lien or mortgage against its property or equipment is instituted against Franchisee and not dismissed within thirty (30) days or is not in the process of being dismissed;
- 16.2.1.12 misuses or makes an unauthorized use of any of the Marks or commits any other act which can reasonably be expected to impair the goodwill associated with any of the Marks;
- 16.2.1.13 fails on three (3) or more separate occasions within any period of twelve (12) consecutive months to submit reports or other information or supporting records when due, to pay any Royalty Fee, Non-APR Service Fee, Technology Fee, Marketing Fee, amounts due for purchases from Franchisor, or other payment when due to Franchisor, whether or not such failures to comply are corrected after notice is delivered to Franchisee;
- 16.2.1.14 violates any health or safety law, ordinance, or regulation, or operates the Franchise Business in a manner that presents a health or safety hazard to itscustomers, employees, or the public and fails to cure the

- violation within forty-eight (48) hours of receipt of notice of such violation or health or safety hazard;
- 16.2.1.15 engages in any activity exclusively reserved to Franchisor;
- 16.2.1.16 fails to comply with any applicable law or regulation within ten (10) days afterbeing given notice of noncompliance;
- 16.2.1.17 repeatedly breaches this Agreement and/or repeatedly fails to comply with mandatory specifications, customer service standards or operating proceduresprescribed in the Manual, whether or not previous breaches or failures are cured; or
- 16.2.1.18 If Franchisor has the right to terminate any other agreement between Franchisor and Franchisee, or such agreement automatically terminates.
- 16.2.2 Except as otherwise provided in Section 16.2.1, Franchisor has the right to terminate this Agreement for the following breaches and defaults by giving notice of such termination stating the nature of the default; provided, however, that Franchisee may avoid termination by curing such default or failure (or by providing proof acceptable to Franchisor that Franchisee has made all reasonable efforts to cure such default or failure and must continue to make all reasonable efforts to cure until a cure is effected if such default or failure cannot reasonably be cured before the effective dateof the termination) within the specified period:
 - 16.2.2.1 within ten (10) days of receiving notice of Franchisee's failure to pay any amounts due to Franchisor;
 - 16.2.2.2 within ten (10) days of receiving notice of Franchisee's failure to maintain insurance as specified in Section 15 of this Agreement; or
 - 16.2.2.3 within thirty (30) days of receiving notice of Franchisee's failure to comply with any other provision of this Agreement or any mandatory specification, standard or operating policy or procedure Franchisor prescribes in the Manuals or otherwise in writing from time to time;
 - 16.2.2.4 within thirty (30) days of receiving notice of Franchisee's failure to use and cooperate in the use of operating systems and tools provided by Franchisor to improve the Franchise Business and/or the franchise system as a whole;
 - 16.2.2.5 within thirty (30) days of receiving notice of Franchisee's failure to regularly attend and actively participate in required training, conference calls, or meetings, conventions, and other events required by Franchisor, or within thirty (30) days of receiving notice of failing to comply with

Performance Improvement Requirements described in Section 8.8;

- 16.2.2.6 within thirty (30) days of receiving notice of Franchisee's failure to embrace newprograms and business building initiatives developed for the enhancement of the performance of the Franchise Business as part of the System;
- 16.2.2.7 within thirty (30) days of receiving notice of Franchisee's regularly acting in a combative or confrontational manner with vendors, customers, other franchisees, or Franchisor's staff; or
- 16.2.2.8 within thirty (30) days of receiving notice that Franchisee has had an excessive number of customer complaints and/or has not acted reasonably and in the best interests of the franchise system in resolving customer complaints.
- 16.2.2.9 within thirty (30) days of receiving notice of any other default by Franchisee or upon Franchisee's failure to comply with any mandatory specification, standardor operating procedure prescribed in the Manual or otherwise prescribed in writing;

16.3 Reinstatement and Extension

If provisions of this Agreement provide for periods of notice less than those required by applicable law, or provide for termination, cancellation, or non-renewal other than in accordance with applicable law, Franchisor may reinstate or extend the term of this Agreement for the purpose of complying with applicable law by submitting a writtennotice to Franchisee without waiving any of Franchisor's rights under this Agreement.

16.4 Right of Franchisor to Discontinue Services to Franchisee

If Franchisee is in breach of any obligation under this Agreement, and Franchisor deliversto Franchisee a notice of default or termination under Section 16.2.2, Franchisor has the right to suspend its performance of any of its obligations under this Agreement including, without limitation, the sale or supply of any services or products for which Franchisor is an Approved Supplier to Franchisee, until such time as Franchisee corrects the breach.

16.5 Right of Franchisor to Operate Franchise Business

Following the delivery of a notice of default or termination under Section 16.2.2, if necessary, in Franchisor's discretion, Franchisor shall have the right, but not the obligation, to assumethe operation of the Franchise Business until such time as Franchisee corrects the breachFranchisor may charge a management fee as stated in the Manuals from time to time, currently equal to FIVE HUNDRED DOLLARS (\$500.00) per day, or twenty-five percent (25%) of the Gross Sales during that period (whichever is greater), and Franchisor must be entitled to reimbursement of all expenses Franchisor incurs that are not paid out of the operating cash flow of the Franchise Business. In the state of Washington, this fee will not

17 RIGHTS AND DUTIES UPON TERMINATION

17.1 Actions to be Taken

Except as otherwise provided, upon termination or expiration, this Agreement and all rights granted hereunder to Franchisee must terminate and Franchisee must:

- 17.1.1 immediately cease to operate the Franchise Business and must not thereafter, directly or indirectly, represent to the public or hold itself out as a present or former franchisee of Franchisor;
- 17.1.2 cease to use the Trade Secrets and other Confidential Information, the System and the Marks including, without limitation, all signs, slogans, symbols, logos, advertising materials, stationery, forms, and any other items which display or are associated with the Marks;
- 17.1.3 and upon demand by Franchisor, at Franchisor's sole discretion, immediately assign (or, if an assignment is prohibited, sublease for the fullremaining term, and on the same terms and conditions as Franchisee's lease) its interest in the lease then in effect for the Franchise Business to Franchisor and Franchisee must furnish Franchisor with evidence satisfactory to Franchisor of compliance with this obligation within thirty (30) days after termination or expiration of this Agreement, and Franchisorhas the right to pay rent and other expenses directly to the party to whom such payment is ultimately due;
- 17.1.4 take such action as may be necessary to cancel or assign to Franchisor, at Franchisor's option, any assumed name or equivalent registration filed with state, city or county authorities which contains the name "THE GLASS GURU" or any other Mark, and Franchisee must furnish Franchisor with evidence satisfactory to Franchisor of compliance with this obligation within thirty (30) days after termination or expiration of this Agreement;
- 17.1.5 return any leased, loaned, or other third-party equipment used in theoperation of the Franchise Business:
- 17.1.6 pay all sums owing to Franchisor, which may include, but not be limited to,all damages, costs and expenses, including reasonable attorneys' fees, unpaid Royalty Fees, and any other amounts due to Franchisor;
- 17.1.7 pay to Franchisor all costs and expenses, including reasonable attorneys' fees, incurred by Franchisor subsequent to the termination or expiration of the Franchise in obtaining injunctive or other relief if required for the enforcement of any provisions of this Agreement;

- 17.1.8 immediately return to Franchisor the Manuals, all materials containing Trade Secrets and all other Confidential Information including records, files, instructions, brochures, agreements, disclosure statements and any and all other materials provided by Franchisor to Franchisee relating to the operation of the Franchise Business (all of which are acknowledged to be Franchisor's property);
- 17.1.9 assign all telephone listings and numbers and e-mail addresses for the Franchise Business to Franchisor, and must notify the telephone company, Internet service provider and all listing agencies of the termination or expiration of Franchisee's right to use any telephone numbers, facsimile numbers and e-mail addresses associated with the Marks in any regular, classified, or other telephone directory or Internet listing and must authorize transfer of same to or at the direction of Franchisor;
- 17.1.10 assign to Franchisor all of Franchisee's rights in and to Internet domain names, email addresses, websites, social media accounts (such as Facebook, LinkedIn, Twitter, YouTube), blogs, vlogs (social videos), onlinesocial networks, wikis, forums, content sharing communities, and other Internet tools ("Internet Tools") used in connection with the Franchise Business and associated with Franchisor's Marks, including passwords and account manager access, and shall notify the necessary parties of the termination of the Franchisee's right to use any the Internet Tools and toauthorize an assignment of same at the direction of Franchisor.
- 17.1.11 Except in the case of termination in connection with a transfer of the Franchise Business under Section 18 or expiration of this Agreement without renewal, Franchisee must pay to Franchisor the greater of an amount equal to Five Thousand Dollars (\$5,000.00) or five percent (5%) of the Franchisee's average Gross Sales for the six (6) calendar months preceding the date of termination or expiration without renewal (the "Holdback Amount").

Franchisor will have the right to ACH transfer the Holdback Amount from Franchisee's account five (5) days before the date of termination or expiration without renewal. Franchisor will hold the Holdback Amount for a period of six (6) months and may use the Holdback Amount during that period to pay damages in connection with the resolution of disputes with Franchisee's customers, vendors, or other third parties, under Section 13.14 of this Agreement. Any portion of the Holdback Amount remaining at the end of the six (6) month period will be paid back to the Franchisee or Franchisee's owners within fourteen (14) days of the end of the six (6) month period.

17.1.12 comply with all other applicable provisions of this Agreement.

17.2 <u>Post-Termination Covenant Not to Compete</u>

17.2.1 Franchisee acknowledges that the restrictive covenants contained in this

Section and in Section 7.5 are fair and reasonable and are justifiably required for purposes including, but not limited to, the following (not applicable in the state of Washington):

- 17.2.1.1 to protect the Trade Secrets and other Confidential Information of Franchisor;
- 17.2.1.2 to induce Franchisor to grant a Franchise to Franchisee; and
- 17.2.1.3 to protect Franchisor against its costs in training Franchisee and its officers, directors, executives, professional staff, and Designated Managers.
- 17.2.2 Except as otherwise approved in writing by Franchisor, neither Franchisee, nor any holder of a legal or beneficial interest in Franchisee, nor any officer, director, executive, manager or member of the professional staff of Franchisee, must, for a period of two (2) years after the expiration or termination of this Agreement, regardless of the cause of termination, either directly or indirectly, for themselves or through, on behalf of or in conjunction with, any person, persons, partnership, corporation, limited liability company or other business entity:
 - own an interest in, manage, operate, or provide services to any Competitive Business located or operating (a) within a fifteen (15) mile radius of the Area of Primary Responsibility, or (b) within a fifteen (15) mile radius of the location of any other Franchise Business in existence at the time of termination or expiration; or
 - 17.2.2.2 solicit or otherwise attempt to induce or influence any customer, employee, orother business associate of Franchisor to terminate or modify his, her or its business relationship with Franchisor, any The Glass Guru Business, or to compete against Franchisor or any The Glass Guru Business; or
- 17.2.3 In furtherance of this Section, Franchisor has the right to require certain individuals to execute standard form nondisclosure or non-competition agreements in a form the same as or similar to the Nondisclosure and Non-Competition Agreement attached as Exhibit 3.
- 17.2.4 Franchisee acknowledges that that this form of Agreement is prepared for use in many jurisdictions with differing laws and public policies and that such laws and public policies may change. Accordingly, Franchisee and Franchisor agree that the non-competition restrictions set forth above may be modified by a Court to the extent necessary to make the non-competition agreements valid and enforceable against franchisee.
- 17.2.5 Franchisee acknowledges that violation of the covenants not to compete contained in this Agreement would result in immediate and irreparable injury

to franchisor for which no adequate remedy at law will be available. Accordingly Franchisee hereby acknowledges that Franchisor may seek to obtain the entry of an injunction prohibiting any conduct by Franchisee or its owners in violation of the terms of the covenants not to compete set forth in this Agreement. Franchisee expressly agrees that it may conclusively be presumed that any violation of the terms of said covenants not to compete was accomplished by and through Franchisees unlawful utilization of our Trade Secrets or Confidential Information. Further, Franchisee expressly agrees that the existence of any claims franchisee may have against Franchisor whether or not arising from this Agreement shall not constitute a defense to the enforcement by Franchisor of the covenants not to compete set forth in this Agreement. Franchisee further agrees to pay all costs and expenses including reasonable attorneys' and experts' fees incurred by Franchisor in connection with the enforcement of those covenants not to compete set forth in this Agreement.

17.3 <u>Unfair Competition</u>

If Franchisee operates any other business during or after termination of this Agreement, Franchisee must not use any reproduction, counterfeit, copy or colorable imitation of the Marks, either in connection with such other business or the promotion of that business, that is likely to cause confusion, mistake, or deception, or that is likely to dilute Franchisor's rights in the Marks. Franchisee must not utilize any designation of origin, description or representation that falsely suggests or represents an association or connection with Franchisor. This Section is not intended as an approval of Franchisee's right to operate other businesses and in no way is it intended to contradict Section 17.1or 17.2. Upon termination or expiration of this Agreement, Franchisee must immediatelycease all use of the Marks, change the telephone and fax numbers, website address andtake such other actions as may be necessary to prevent any association between Franchisor or the System and Franchisee and any business subsequently operated by Franchisee. If Franchisee fails or refuses to comply with the requirements of this Section, Franchisor has the right to make or cause to be made such changes as may be required, at the expense of Franchisee, which expense Franchisee must pay upon demand.

17.4 Franchisor's Option to Purchase Certain Business Assets

Franchisor has the right (but not the duty), for a period of thirty (30) days after termination or expiration of this Agreement, to purchase any or all assets of the Franchise Business including, without limitation, leasehold improvements, equipment, supplies, and other inventory. The purchase price must be equal to the assets' fair market value, excluding any goodwill. If fair market value cannot be mutually agreed upon between Franchisor and Franchisee, a third-party valuation must be ordered by a mutually agreed upon appraiser. If a mutually agreed upon appraiser cannot be established, Franchisor and Franchisee must each order their own appraisals. If after review of the two appraisals a market value cannot be agreed upon, a third appraisal may be ordered to make the final determination. The costs of the appraisal services shall be shared equally by Franchisor and Franchisee. If Franchisor elects to exercise this option to purchase, it has the right to set off all amounts due from Franchisee under this Agreement, if any, against such purchase price.

17.5 Survival of Certain Provisions

All obligations of Franchisor and Franchisee, which expressly or by their nature survive the expiration or termination of this Agreement, must continue in full force and effect subsequent to and notwithstanding their expiration or termination and until satisfied orby their nature expire.

18 TRANSFERABILITY OF INTEREST

18.1 <u>Transfer by Franchisor</u>

This Agreement and all rights and duties hereunder are fully transferable in whole or in part by Franchisor and such rights will inure to the benefit of any person or entity to whom transferred; provided, however, that with respect to any assignment resulting in the subsequent performance by the assignee of the functions of Franchisor, the assigneemust assume the obligations of Franchisor hereunder and Franchisor shall thereafter have no liability for the performance of any obligations contained in this Agreement.

18.2 <u>Transfer by Franchisee to a Third Party</u>

The rights and duties of Franchisee as described in this Agreement, and the Franchise granted, are personal to Franchisee (or its owners), and Franchisor has entered into this Agreement in reliance upon Franchisee's personal or collective skill and financial ability. Accordingly, neither Franchisee nor any holder of a legal or beneficial interest in Franchisee may sell, assign, convey, give away, pledge, mortgage, sublicense or otherwise transfer, whether by operation of law or otherwise, any interest in this Agreement, the Franchise granted, its assets or any part or all the ownership interest in Franchisee without the prior written approval of Franchisor, which approval must not be unreasonably withheld or delayed. Any purported transfer without such approval must be null and void and must constitute a material breach of this Agreement. If Franchisee is in compliance with this Agreement, Franchisor's consent to such transfer must be conditioned upon the satisfaction of the following requirements:

- 18.2.1 Franchisee has complied with the requirements described in Section 19;
- 18.2.2 all obligations owed to Franchisor, and all other outstanding obligations relating to the Franchise Business, are fully paid and satisfied;
- 18.2.3 Franchisee and all transferring owners, have executed a general release, ina form the same as or similar to the General Release attached as Exhibit 2, releasing any and all claims against Franchisor, any Affiliate and their officers, directors, shareholders, managers, members, partners, owners and employees (in their corporate and individual capacities) including, without limitation, claims incident to the termination of this Agreement or to the transfer of Franchisee's interest or to the transfer of Franchisee's ownership all or any

- part of the Franchise; provided, however, that if a general release is prohibited, Franchisee must give the maximum release allowed by law;
- 18.2.4 the prospective transferee has satisfied Franchisor that it meets Franchisor's management, business and financial standards, and otherwise possesses the character and capabilities, including business reputation and redit rating, as Franchisor may require demonstrating ability to conduct the Franchise Business;
- 18.2.5 the transferee and, if Franchisor require, all persons owning any interest inthe transferee, have executed the then-current Franchise Agreement for new franchisees, which may be substantially different from this Agreement, including different Royalty Fee, Non-APR Service Fee, Technology Fee, and Marketing Fee rates, a different Area of Primary Responsibility, and other material provisions, and the franchise agreement then executed must be for the term specified in such agreement; At Franchisor's option, the parties shall execute an assignment and assumption agreement satisfactory to Franchisor whereby the transferee assumes Franchisee's obligations under this Agreement in lieu of execution of a new franchise agreement;
- 18.2.6 Franchisee has provided Franchisor with a complete copy of all contracts and agreements and related documentation between Franchisee and the prospective transferee relating to the intended sale or transfer of the Franchise and Franchisor must have determined that the price and terms of payment are not so burdensome to adversely affect the future operations of transferee's Franchise Business;
- 18.2.7 Franchisee, or the transferee, has paid to Franchisor a transfer fee in theamount of SEVEN THOUSAND FIVE HUNDRED DOLLARS (\$7,500.00);
- 18.2.8 the transferee and all holders of a legal or beneficial interest in the transferee, have agreed to be personally bound jointly and severally by all provisions of the new Franchise Agreement , or this Agreement for the remainder of its term if this Agreement is assigned to transferee, to guarantee the full performance of that franchise agreement;
- 18.2.9 the lessor of the Approved Location has consented to an assignment of the lease or sublease of the Approved Location to transferee;
- 18.2.10 transferee has obtained all necessary consents and approvals by third parties and all applicable federal, state and local laws, rules, ordinances and requirements applicable to the transfer have been complied with or satisfied;
- 18.2.11 the transferee agrees that Franchisee and its new Designated Manager must complete, to Franchisor's satisfaction, a training program in substance similar to the initial training described in Section 8.1 prior to assuming the management of the day-to-day operation of the Franchise Business. The

training must be at the expense of the transferee at the then-current standard rates for technician training.

18.2.12 If required by Franchisor, Franchisee shall enter into an agreement with Franchisor agreeing that any obligations of the transferee to make installment payments of the purchase price to Franchisee will be subordinate to the transferee's ongoing obligations to Franchisor, including without limitation, payment of fees and other amounts due to Franchisor.

18.3 <u>Franchisor's Disclosure to Transferee</u>

Franchisor has the right, without liability of any kind or nature whatsoever to Franchisee, to make available for inspection by any intended transferee of Franchisee all or any part of Franchisor's records relating to this Agreement, the Franchise Business or to the history of the relationship of the parties to this Agreement. Franchisee specifically consents to such disclosure by Franchisor and must release and hold Franchisor harmlessfrom and against any claim, loss or injury resulting from an inspection of Franchisor's records relating to the Franchise Business by an intended transferee identified by Franchisee.

18.4 Franchisee Disclosure to Transferee

Franchisee must provide the new franchisee with information specified by Franchisor, regardless of whether it is required under the purchase agreement between the parties. This information may include, but is not limited to, employment and personnel records for existing employees and employee candidates, currently valued loss run reports for the past five policy years (or for all years of operation if less than 5 years) for all insurance policies, annual premium amounts for each year for the past five policy years (or for all years of operation if less than 5 years), and payrolls for the past five years (or for all years of operation if less than 5 years). Franchisee agrees that Franchisor shall have the right to communicate and confer with transferee on any aspect of the proposed assignment of the Franchise Business and to provide the specified information directly to the new franchisee to the extent that Franchisor has that information.

18.5 For-Sale Advertising

Franchisee must not, without prior written consent of Franchisor, place in, on or upon the location of the Franchise Business, or in any communication media, any form of advertising relating to the sale of the Franchise Business or the rights granted hereunder. Franchisee must not, whether in or on the location of the Franchise Business or in any communication media, advertise a "going-out-of-business," inventory liquidation or similar sale or event.

18.6 Transfer by Death or Incapacity

18.6.1 Upon the death or Incapacity (as determined by a court of competent jurisdiction) of any individual franchisee or any holder of a legal or beneficial interest in Franchisee, the appropriate representative of such person (whether administrator, personal representative or trustee) will, within a reasonable time not exceeding one hundred eighty (180) days following such event,

transfer such individual's interest in the Franchise Business or in Franchisee to a third party approved by Franchisor. Such transfers, including transfers by will or inheritance, must be subject to the conditions for assignments and transfers contained in this Agreement. During such one hundred eighty (180) day period, the Franchise Business must always remain under the primary management of a Designated Manager who otherwise meets Franchisor's management qualifications.

18.6.2 Following the death or Incapacity of an owner of the Franchise Business, if necessary, in Franchisor's discretion, Franchisor shall have the right, but not the obligation, to assume operation of the Franchise Business until the deceased or incapacitated owner's interest is transferred to a third party approved by Franchisor. Franchisor may charge a management fee as stated in the Manuals from time to time, currently equal to FIVE HUNDRED DOLLARS (\$500.00) per day or twenty-five percent (25%) of the Gross Sales for that period (whichever is greater), and Franchisor shall be entitled to reimbursement of any expenses Franchisor incurs that are not paid out of the operating cash flow of the Franchise Business.

19 RIGHT OF FIRST REFUSAL

19.1 <u>Submission of Offer</u>

If Franchisee, or any of its owners, proposes to sell the Franchise Business (or any of its assets outside of the normal course of business), any ownership interest in Franchisee or any ownership interest in the Franchise granted hereunder, Franchisee must obtain and deliver a bona fide, executed written offer or proposal to purchase, along with all pertinent documents including any contract or due diligence materials, to Franchisor.

The offer must apply only to an approved sale of the assets or interests listed above and may not include any other property or rights of Franchisee or any of its owners.

19.2 <u>Franchisor's Right to Purchase</u>

Franchisor shall, for thirty (30) days from the date of delivery of all such documents, have the right, exercisable by written notice to Franchisee, to purchase the offered assets or interest for the price and on the same terms and conditions contained in such offer communicated to Franchisee. Franchisor has the right to substitute cash for the fair market value of any form of payment proposed in such offer. Franchisor's credit shall be deemed at least equal to the credit of any proposed buyer. After providing notice to Franchisee of Franchisor's intent to exercise this right of first refusal, Franchisor must have up to sixty (60) days to close the purchase. Franchisor shall be entitled to receive from Franchisee all customary representations and warranties given by Franchisee as the seller of the assets or such ownership interest or, at Franchisor's election, such representations and warranties contained in the proposal.

19.3 Non-Exercise of Right of First Refusal

If Franchisor does not exercise this right of first refusal within thirty (30) days, the offer or proposal may be accepted by Franchisee or any of its owners, subject to Franchisor's prior written approval as required by Section 18.2. Should the sale fail to close within onehundred twenty (120) days after the offer is delivered to Franchisor, Franchisor's right of first refusal must renew and be implemented in accordance with this Section.

20 BENEFICIAL OWNERS OF FRANCHISE

Franchisee represents, and Franchisor enters into this Agreement in reliance upon such representation, that the name, entity classification, state of organization, and all persons with a beneficial interest in Franchisee and percentages of ownership of those persons in Franchisee are set forth on the Holders of Legal or Beneficial Interest in Franchisee; Officers; Directors, attached as Exhibit 5. If Franchisee entity is owned by one or more other entities, this form must be completed for all entities, directly or indirectly, owning an interest in Franchisee. Franchisee represents that the information stated in Exhibit 5 is accurate and complete. Franchisee agrees that it will immediately notify Franchisor (and comply with the provisions of Section 18 of this Agreement, if applicable) if there is any change in the ownership of Franchisee or other information set forth in Exhibit 5, including if Franchisee awards some ownership interest to an employee as an incentive or other compensation. In addition, Franchisee must provide to Franchisor an update of the information in Exhibit 5 or confirm that is remains unchanged on an annual basis (or as otherwise specified by Franchisor in the Manuals). Each of the persons named in Exhibit 5 must guaranty the obligations of Franchisee to Franchisor, unless Franchisor consents otherwise. Failure to comply with these requirements will be a material default under this Agreement.

Franchisee's board of directors, members or other governing body must pass a resolution requiring, and the corporation or other entity must otherwise require, the prominent placement of the following notation regarding transfer restrictions on each certificate representing ownership in the entity:

20.1.1 "The transfer of the shares or other ownership represented by this certificate is subject to the terms and conditions of a certain written franchise agreement entered into with The Glass Guru Enterprises, Inc."

Franchisee must provide Franchisor with proof of complying with this provision withinfifteen (15) days following the date on which any entity obtains rights under this Agreement, in whole or in part.

21 RELATIONSHIP AND INDEMNIFICATION

21.1 Relationship

This Agreement is purely a contractual relationship between the parties and does not appoint or make Franchisee an agent, legal representative, joint venturer, partner, employee, servant, or independent contractor of Franchisor for any purpose whatsoever. Franchisee may not represent or imply to third parties that Franchisee is an agent of Franchisor, and Franchisee

is in no way authorized to make any contract, agreement, warranty, or representation on behalf of Franchisor, or to create any obligation, express or implied, on Franchisor's behalf. Franchisee agrees not to use any Mark in signing any contract, purchase agreement, mortgage, lease, negotiable instrument, application for any license or permit, or any other legal obligation, or in any manner that may result in liability to Franchisor for any indebtedness or obligation of Franchisee. Except as expressly authorized by this Agreement, neither Franchisee nor Franchisor will make any express or implied agreements, warranties, quarantees or representations, or incur any debt, in the name of or on behalf of the other or represent that the relationship between the parties is other than that of franchisor and franchisee. During the term of this Agreement, and any extension or renewal of this Agreement, Franchisee must hold itself out to the public only as a franchisee and an owner of the Franchise Business operating the Franchise Business under a franchise from Franchisor. Franchisee must take such affirmative action as may be necessary to doso including, without limitation, exhibiting a notice in the form specified in Section 6.2 and on all forms, stationery or other written materials, the content of which Franchisor has the right to specify. Under no circumstances must Franchisor be liable for any act, omission, contract, debt, or any other obligation of Franchisee. Franchisor shall in no way be responsible for any injuries to persons or property resulting from the operation of the Franchise Business.

21.2 Standard of Care

This Agreement does not establish a fiduciary relationship between the parties. Unless otherwise specifically provided in this Agreement with respect to certain issues, whenever this Agreement requires Franchisee to obtain Franchisor's written consent or permits Franchisee to take any action or refrain from taking any action, Franchisor is freeto act in its own self-interest without any obligation to act reasonably, to consider the impact on Franchisee or to act subject to any other standard of care limiting Franchisor's right, except as may be provided by statute or regulation.

21.3 <u>No Employment Relationship</u>

Franchisee expressly acknowledges that Franchisor is not Franchisee's employer or an employer of any of Franchisee's employees or subcontractors. In addition, Franchisor is not a joint employer with Franchisee. Franchisee acknowledges that the training, guidance, advice and assistance provided by Franchisor, for obligations under this Agreement and the standards, specifications, policies, and procedures required by Franchisor under this Agreement and in the Manuals are imposed not for the purpose of exercising control over Franchisee or the operation of the Franchise Business, but rather for the limited purpose of protecting the Marks, System, Trade Secrets and Confidential Information, goodwill and brand consistency. Franchisee shall notify and communicate clearly with its employees and subcontractors in all dealings, including without limitations, employment applications and other employment forms, written and electronic correspondence, agreements or contracts, paychecks, employee handbooks, employment policies and procedures, and other written materials that Franchisee (and only Franchisee) is their employer and that Franchisor is not their employer. Franchisee is solely responsible for the management and supervision of the Franchise Business as an independent franchise owner/operator.

21.4 <u>Indemnification</u>

Franchisee must hold harmless and indemnify Franchisor, any Affiliate, all holders of a legal or beneficial interest in Franchisor and all officers, directors, executives, managers, members, partners, owners, employees, agents, successors and assigns (collectively "Franchisor Indemnitees") from and against all losses, damages, fines, costs, expenses or liability (including attorneys' fees and all other costs of litigation) incurred in connectionwith any action, suit, demand, claim, investigation or proceeding, or any settlement,

which arises from or is based upon Franchisee's (a) ownership or operation of the Franchise Business; (b) violation, breach or asserted violation or breach of any federal, state or local law, regulation or rule; (c) breach of any representation, warranty, covenant, or provision of this Agreement or any other agreement between Franchisee and Franchisor (or an Affiliate); (d) defamation of Franchisor or the System; (e) acts, errors or omissions committed or incurred in connection with the Franchise Business, including any negligent or intentional acts; or (f) infringement, violation or alleged infringement orviolation of any Mark, patent or copyright or any misuse of the Confidential Information. This indemnification includes without limitation actual and consequential damages, reasonable arbitrators', attorneys', accountants' and expert witness fees (including those for appeal), costs of investigation and proof of fats, court costs, other litigation expenses and travel and living expenses. The indemnification obligations described herein will continue in full force and effect after, and notwithstanding, the expiration or termination of this Agreement.

21.5 Right to Retain Counsel

Franchisee must give Franchisor immediate notice of any such action, suit, demand, claim, investigation or proceeding that may give rise to a claim for indemnification by aFranchisor Indemnitee. Franchisor has the right to retain counsel of its own choosing inconnection with any such action, suit, demand, claim, investigation or proceeding. In order to protect persons, property, Franchisor's reputation or the goodwill of others, Franchisor has the right to, at any time without notice, take such remedial or corrective actions as it deems expedient with respect to any action, suit, demand, claim, investigation or proceeding if, in Franchisor's sole judgment, there are grounds to believe any of the acts or circumstances listed above have occurred. If Franchisor's exercise of its rights under this Section causes any of Franchisee's insurers to refuse to pay a third-party claim, all cause of action and legal remedies Franchisee might have against such insurer must automatically be assigned to Franchisor without the need for any further action on either party's part. Under no circumstances must Franchisor be required or obligated to seek coverage from third parties or otherwise mitigate losses in order to maintain a claim against Franchisee. The failure to pursue such remedy or mitigate such loss must in no way reduce the amounts recoverable by Franchisor from Franchisee.

22 GENERAL CONDITIONS AND PROVISIONS

22.1 No Waiver

No failure of Franchisor to exercise any right or power reserved to it hereunder, or to insist uponstrict compliance by Franchisee with any obligation or condition hereunder, and no custom nor practice of the parties in variance with the terms of this Agreement, shall constitute a waiver of Franchisor's right to demand exact compliance with the terms of this

Agreement. Waiver by Franchisor of any particular default by Franchisee shall not bebinding unless in writing and executed by Franchisor and must not affect nor impair Franchisor's right with respect to any subsequent default of the same or of a different nature. Subsequent acceptance by Franchisor of any payment(s) due shall not be deemed to be a waiver by Franchisor of any preceding breach by Franchisee of any terms, covenants, or conditions of this Agreement.

22.2 <u>Injunctive Relief</u>

As any breach by Franchisee of any of the restrictions contained in Sections 6, 7 and 17 would result in irreparable injury to Franchisor, and as the damages arising out of any such breach would be difficult to ascertain, in addition to all other remedies provided bylaw or in equity, Franchisor must be entitled to seek injunctive relief (whether a restraining order, a preliminary injunction or a permanent injunction) against any such breach, whether actual or contemplated, and Franchisee specifically waives any and all defenses to injunctive relief.

If Franchisor obtains injunctive relief, Franchisee shall pay Franchisor an amount equal to the total of its costs of obtaining it, including, without limitation, reasonable arbitrators', attorneys' and expert witness fees, costs of investigation and proof of facts court costs, other arbitration or litigation expenses and travel and living expenses, and any damages Franchisor incurs as a a result of the breach of any such provision. Franchisee further agrees to waive any claims for damage in the event there is a later determination that an injunction or specific performance order was issued improperly.

22.3 Notices

All notices required or permitted under this Agreement must be in writing and must be deemed received: (a) at the time delivered by hand to the recipient party (or to an officer, director or partner of the recipient party); (b) on the next business day after transmission by email; (c) two (2) business days after being sent via guaranteed overnight delivery by a commercial courier service; or (d) five (5) business days after being sent by U.S. PostalService Certified or Registered Mail, return receipt requested. Either party may change its address for service by a written notice sent in accordance with this Section. All notices, payments and reports required by this Agreement must be sent to Franchisee at its approved location address on file with Franchisor and the email address provided by Franchisor and to Franchisor at the following address:

The Glass Guru Enterprises, Inc
ATTN: President
5550 Granite Parkway Suite 280
Plano, TX 75024
info@theglassguru.com and compliance@theglassguru.com

22.4 Cost of Enforcement or Defense

If Franchisor or Franchisee is required to enforce this Agreement in a judicial or arbitration proceeding, the prevailing party must be entitled to reimbursement of itscosts, including reasonable accounting and attorneys' fees, in connection with such proceeding.

22.5 <u>Guaranty and Assumption of Obligations</u>

All holders of a legal or beneficial interest in Franchisee must be required to execute, as of the date of this Agreement, the Guaranty and Assumption of Obligations attached as Exhibit 4, through which such holders agree to assume and discharge all of Franchisee's obligations under this Agreement and to be personally liable hereunder for all of the same.

22.6 Approvals

Whenever this Agreement requires the prior approval or consent of Franchisor, Franchisee must make a timely written request to Franchisor for such approval and, except as otherwise provided, any approval or consent granted must be effective only if in writing. Franchisor makes no warranties or guarantees upon which Franchisee may relyand assumes no liability or obligation to Franchisee or any third party to which it would not otherwise be subject, by providing any waiver, approval, advice, consent, or services to Franchisee in connection with this Agreement, or by reason of any neglect, delay, ordenial of any request for approval.

22.7 Entire Agreement

This Agreement, its exhibits and the documents referred to must be construed together and constitute the entire, full, and complete agreement between Franchisor and Franchisee concerning the subject matter of this Agreement and must supersede all prioragreements. No other representation, oral or otherwise, has induced Franchisee to execute this Agreement, and there are no representations [other than those within Franchisor's (The Glass Guru Enterprises, Inc.) [Franchise Disclosure Document], inducements, promises or agreements, oral or otherwise, between the parties not embodied in the Agreement, which are of any force or effect with respect to the matters described in or contemplated by this Agreement or otherwise. No amendment, change, or variance from this Agreement will be binding on either party unless executed in writing by both parties. Nothing in this Agreement or any related agreement requires the franchisee to waive reliance on the representations made in the disclosure document. Nothing in this Agreement requires the franchisee to waive reliance on, or in any related agreement is intended to disclaim the representations made in the Franchise Disclosure Document.

22.8 <u>Severability and Modification</u>

22.8.1 Except as noted below, each paragraph, part, term, and provision of this Agreement must be considered severable. If any paragraph, part, term, or provision of this Agreement is ruled to be unenforceable, unreasonable, or invalid, such ruling must not impair the operation of or affect the remaining portions, paragraphs, parts, terms and provisions of this Agreement, and the latter must continue to be given full force and effect and bind the parties; and such unenforceable, unreasonable, or invalid paragraphs, parts, terms, or provisions must be deemed not part of this Agreement. If Franchisor determines that a finding of invalidity adversely affects the basic consideration of this Agreement, Franchisor has the right to, at its option, terminate this Agreement.

22.8.2 Notwithstanding the above, each of the covenants contained in Sections 6 and 7 and 17 must be construed as independent of any other covenant or provision of this Agreement. If all or any portion of any such covenant is held to be unenforceable, unreasonable, or invalid, then it must be amended to provide for limitations on use of Marks or disclosure of Trade Secrets and other Confidential Information or on competition to the maximum extent provided or permitted by law.

22.9 Construction

All captions are intended solely for the convenience of the parties, and none must be deemed to affect the meaning or construction of any provision of this Agreement.

22.10 Force Majeure

Whenever a period of time is provided in this Agreement for either party to perform anyact, except pay monies, neither party will be liable nor responsible for any delays due to strikes, lockouts, casualties, acts of God, war, terrorism, pandemic, governmental regulation or control or other causes beyond the reasonable control of the parties, and the time period for the performance of such act must be extended for the amount of time of the delay. This clause must not result in an extension of the term of this Agreement.

22.11 <u>Timing</u>

Time is of the essence. Except as described in Section 22.10, failure to perform any activithin the time required or permitted by this Agreement is a material breach.

22.12 <u>Withholding Payments</u>

Franchisee must not, for any reason, withhold payment of any Royalty Fees or other amounts due to Franchisor or to an Affiliate. Franchisee must not withhold or offset any amounts, damages, or other monies allegedly due to Franchisee against any amounts due to Franchisor. No endorsement or statement on any payment for less than the full amount due to Franchisor will be construed as an acknowledgment of payment in full, oran accord and satisfaction, and Franchisor has the right to accept and cash any such payment without prejudice to Franchisor's right to recover the full amount due, or pursue any other remedy provided in this Agreement or by law. Franchisor has the right to apply any payments made by Franchisee against any of Franchisee's past due indebtedness as Franchisor deems appropriate. Franchisor shall set off sums Franchisorowes to Franchisee against any unpaid debts owed by Franchisee to Franchisor.

22.13 Further Assurances

Each party to this Agreement will execute and deliver such further instruments, contracts, forms, or other documents, and will perform such further acts, as may be necessary or desirable to perform or complete any term, covenant or obligation contained in this Agreement.

22.14 Third-Party Beneficiaries

Anything to the contrary notwithstanding, nothing in this Agreement is intended, nor willbe deemed, to confer upon any person or legal entity other than Franchisor or Franchisee, and their respective successors and assigns as may be contemplated by this Agreement, any rights, or remedies under this Agreement.

22.15 Multiple Originals; Electronic Signature

Both parties may execute multiple copies of this Agreement, and each executed copy will be deemed an original. This Agreement may be signed with full legal force and effect using electronic signatures and records. Delivery of this Agreement by electronic mail or other functionally equivalent means of transmission constitutes valid and effective delivery.

23 DISPUTE RESOLUTION

23.1 Choice of Law

Except to the extent this Agreement or any particular dispute is governed by the U.S. Trademark Act of 1946 or other federal law, this Agreement must be governed by and construed in accordance with the laws of the State of Texas (without reference to its conflict of laws principles excluding any law regulating the sale of franchises or governing the relationship between a franchisor and franchisee, unless the jurisdictional requirements of such laws are met independently without reference to this Section. The Federal Arbitration Act must govern all matters subject to arbitration. References to any law refer also to any successor laws and to any published regulations for such law as in effect at the relevant time. References to a governmental agency also refer to any regulatory body that succeeds in the function of such agency.

23.2 Consent to Jurisdiction

Any action brought by either party, except those claims required to be submitted to arbitration, must be brought exclusively in the appropriate state or federal court located in or serving Collin County, Texas. The parties waive all questions of personal jurisdiction or venue for the purposes of carrying out this provision. The parties submit to service of process by registered mail, return receipt requested or by any other manner provided by law. Claims for injunctive relief may be brought by Franchisor where Franchisee is located. This exclusive choice of jurisdiction and venue provision shall not restrict the ability of the parties to confirm or enforce judgments or arbitration awards in any appropriate jurisdiction.

23.3 <u>Cumulative Rights and Remedies</u>

No right or remedy conferred upon or reserved to Franchisor or Franchisee by this Agreement

is intended to be, nor must be deemed, exclusive of any other right or remedy of this Agreement or by law or equity provided or permitted, but each must be in addition to every other right or remedy. Nothing contained in this Agreement will bar Franchisor's right to obtain injunctive relief against threatened conduct that may cause it loss or damages, including obtaining restraining orders and preliminary and permanent injunctions.

23.4 <u>Limitations of Claims</u>

Any claim concerning the Franchise Business or this Agreement or any related agreement will be barred unless an action for a claim is commenced within one (1) yearfrom the date on which Franchisee or Franchisor knew or should have known, in the exercise of reasonable diligence, of the facts giving rise to the claim.

23.5 <u>Limitation of Damages</u>

Franchisee and Franchisor each waive, to the fullest extent permitted by law, any right orclaim for any punitive or exemplary damages against the other and agree that if there is a dispute with the other, each will be limited to the recovery of actual damages sustained by it including reasonable accounting and legal fees as provided in Section 22.4. Franchisee waives and disclaims any right to consequential damages in any action or claim against Franchisor concerning this Agreement or any related agreement. In anyclaim or action brought by Franchisee against Franchisor concerning this Agreement, Franchisee's contract damages must not exceed and must be limited to refund of Franchisee's Franchise Fee and Royalty Fees.

23.6 Waiver of Jury Trial

FRANCHISEE AND FRANCHISOR EACH IRREVOCABLY WAIVE TRIAL BY JURY IN ANY ACTION, WHETHER AT LAW OR EQUITY, BROUGHT BY EITHER OF THEM.

23.7 Arbitration

This Agreement evidences a transaction involving commerce and, therefore, the Federal Arbitration Act, Title 9 of the United States Code is applicable to the subject matter contained under this Agreement. Except for controversies or claims relating to the ownership of any of Franchisor's Marks or the unauthorized use or disclosure of Franchisor's Trade Secrets or other Confidential Information, covenants against competition and other claims for injunctive relief, all disputes arising out of or relating to this Agreement or to any other agreements between the parties, or with regard to interpretation, formation or breach of this or any other agreement between the parties, must be settled by binding arbitration conducted in Denton County, Texas, in accordance with the Commercial Arbitration Rules of the American Arbitration Association then in effect. The proceedings will be held by a single arbitrator agreed upon by the parties or otherwise appointed by the Circuit Court for the State of Texas located in or serving Denton County, Texas. The decision of the arbitrator must be final and binding upon the parties. Judgment upon the award rendered by the arbitrator maybe entered in any court having personal and subject matter jurisdiction.

Franchisee acknowledges that it has read the terms of this binding arbitration provision and affirms that this provision is entered into willingly and voluntarily and without any fraud, duress, or undue influence on the part of Franchisor or any of Franchisor's agents or employees. Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of Texas and federal laws (such the Federal Arbitration Act) toany provisions of a franchise agreement restricting venue to a forum outside the State of Texas.

24 AFFIRMATIONS

24.1 Receipt of this Agreement and the Franchise Disclosure Document

Franchisee represents and acknowledges that it has received an exact copy of this Agreement and its exhibits at least seven (7) days prior to the date on which this Agreement was executed. Franchisee has received, at least fourteen (14) calendar days prior to the date on which this Agreement was executed, the Franchise Disclosure Document required by the Trade Regulation Rule of the Federal Trade Commission entitled Disclosure Requirements and Prohibitions Concerning Franchising.

24.2 <u>True and Accurate Information</u>

Franchisee represents that all information described in all applications, financial statements and submissions to Franchisor is true, complete, and accurate in all respects and Franchisee acknowledges that Franchisor is relying upon the truthfulness, completeness, and accuracy of such information.

24.3 <u>No Violation of Other Agreements</u>

Franchisee represents that entering into (signing) this Agreement will not violate any other agreement or commitment to which Franchisee or any holder of a legal or beneficial interest in which Franchisee is a party.

[BALANCE OF THIS PAGE INTENTIONALLY LEFT BLANK.]

The parties to this Agreement, intending to be legally bound have duly executed this Agreement on the dates below each signature.

FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	
BY:	Entity Name
NAME:	Entity (Valle)
TITLE:	BY:
	NAME:

TITLE:	
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:

EXHIBIT 1

TO THE FRANCHISE AGREEMENT THE GLASS GURU ENTERPRISES, INC.

SPECIFICS

Franch	nisee's Name:			
A.	Approved Locati Business	on (Section 2.2) - The pro	oposed initial location for the I	Franchise is:
В.	Area of Primary I include the follow	Responsibility (Section 2	2.3) – The Area of Primary Resp	onsibilityshall
	Postal Service du Franchisor reserve of the Area of Pr	ring the term of this A s the right to determine mary Responsibility. Ot	p codes are subdivided by the agreement and new zip code whether the new zip codes wil herwise, the Area of Primary Term, except as specifically p	s are created, I become part Responsibility
C.	The Franchise For	ee (Section 3.1) – is \$_	US dollars unl	ess otherwise
D.	DBA - The permit	ted "doing business as" ı	name is	
	The Glass Guru of		(Citv)	

This Exhibit 1 is signed on this date: _	
3	

FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	
	Entity Name
BY:	BY:
NAME:	NAME:
TITLE:	TITLE:
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:
	BY:
	NAME:

TITLE:

EXHIBIT 2

TO THE FRANCHISE AGREEMENT THE GLASS GURU ENTERPRISES, INC.

GENERAL RELEASE

Γhis General Release is made and given on thisday of, 20 by	,
("Releasor") an	
ndividual / corporation / limited liability company / partnership with a principal address of	
, in consideratio	n of:
[Initial one as applicable]	
Franchisor will not refund the Franchise Fee Franchisee paid to Franchisor u	•
Franchisor's termination of the franchise agreement entered into between Franchisee	and
Franchisor for Franchisee's failure to select a site for the Franchise Business under Section 5.2o	of the
Franchise Agreement.	
Franchisor will not refund the Franchise Fee Franchisee paid to Franchisor uffranchisor's termination of the franchise agreement entered into between Franchisee	•
Franchisor for Franchisee's failure to develop an Approved Location under Section 5.2 o	
Franchise Agreement.	
Franchisor will not refund the Franchise Fee Franchisee paid to Franchisor u	upon
Franchisor's termination of the franchise agreement entered into between Franchisee	and
Franchisor for Franchisee's failure to complete the initial training program under Section 8.4o Franchise Agreement	f the

In consideration of the mutual and several agreements recited above, Franchisee does forever release and discharge Franchisor, and its affiliates, and their respective officers, directors, shareholders, manager, members, partners, owners, and employees, in that capacity and individually, its guarantors, successors, and assigns on behalf of himself, herself, and itself or hisor her heirs, executors, and administrators and its successors and assigns from all manner of actions, cause, causes of action, suits, debts, sums of money, accounts, promises, variances, trespasses, damages, judgments, execution, claims and demands, whatsoever, in law or in equity, arising out of or related to the Franchise Agreement between the parties which he, she or it has, or has had, or which his or her heirs, executors or administrators and its successors and assigns hereafter can, or may have, for upon or by reason of any matter, cause or thing whatsoever at any time prior to the date of this Agreement.

Warranty of Volitional Agreement. The parties warrant and represent that this Agreement is freely and voluntarily executed by the parties, and each of them, after having been apprised of all of the relevant information and data by their respective attorneys. The parties executing this Agreement warrant and represent that they have not relied on any inducements, promises or representations made by any party or its representative, or any other person, except for those expressly described in this Agreement.

This release is inapplicable with respect to claims arising under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted thereunder.

This General Release must not be amended or modified unless such amendment or modification is in writing and is signed by Releasor and Releasee.

FRANCHISEE:

Releasor has executed this General Release as of the date first above written.

	Entity Name
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:
ACKNOWLEDGMENT	
State/Province of:	
On thisday of, 20before m	ne personally appeared
known to me to be the	same person or who provided
	d number) as proof thereof, and whose name is
signed to the foregoing General Release, an uses and purposes therein set forth, [and who(title) of	d acknowledged the execution thereof for the didswear and say that he/she is the
	ne), and he/she has the authority to executesaid
General Release].	
•	
IN WITNESS WHEREOF, I have hereunto set m	y hand and official seal.
(NOTARIAL SEAL)	
Notary Public	
My Commission expires:	

EXHIBIT 3

TO THE FRANCHISE AGREEMENT THE GLASS GURU ENTERPRISES, INC.

NONDISCLOSURE AND NON-COMPETITION AGREEMENT

Between Franchisor & Franchisee

This "Agreement" made as of this date:	<u></u>
	("Franchisee").
WITNESSETH:	
WHEREAS, Franchisee is a party to that certain Franchise Agreement with Th Enterprises, Inc., ("Franchisor") dated	ne Glass Guru -
("Franchise Agreement");	
WHEREAS, Franchisor desires Franchisee to have access to and/or to review	certain Trade Secrets,

WHEREAS, Franchisor is required by the Franchise Agreement to have Franchisee execute this Agreement prior to providing Franchisee access to said Trade Secrets; and

WHEREAS, Franchisee agrees not to disclose any such Trade Secrets to any other party and/oruse such Trade Secrets to compete against Franchisor or any other the Glass Guru franchisee, ("Competitive Business", as defined below) now or in the future.

NOW, THEREFORE, in consideration of the mutual promises and undertakings described here, and intending to be legally bound, the parties now mutually agree as follows:

1. Trade Secrets

Franchisor possesses and will possess Trade Secrets, which is important to its business. For purposes of this Agreement, "Trade Secrets" is information, without regard to formincluding, but not limited to, technical or non-technical data, formulas, patterns, compilations, programs, devices, methods, techniques, drawings, processes, financial data, financial plans, strategic plans, product plans, lists of actual or potential customers or suppliers which are not commonly known by or available to the public and which information: (a) derives economic value, actual or potential, from not being generally known to, and not being readily ascertainable by proper means by, other persons who can obtain economic value from its disclosure or use; and, (b)is the subject of efforts that are reasonable under the circumstances to maintain its secrecy. Franchisee understands Franchisor's providing of access to the Trade Secrets creates a relationship of confidence and trust between Franchisee and Franchisor with respect to the Trade Secrets.

2. Confidentiality/Non-Disclosure

which are more particularly described below;

a) Franchisee acknowledges that Franchisor will disclose Trade Secrets and other Confidential Information to Franchisee during the training program, through the Manual, and as a result of guidance furnished to Franchisee during the term of the Franchise Agreement. Franchisee must notacquire any interest in the Trade Secrets or other Confidential Information, other than the right to use it in the development

and operation of the Franchise Business and in performing its duties during the term of this Agreement. Franchisee acknowledges that the use or duplication of the Trade Secrets or other Confidential Information in any other business venture would constitute an unfair method of competition. Franchisee acknowledges that the Trade Secrets and other Confidential Information are proprietary and are disclosed to Franchisee solely on the condition that Franchisee (and all holders of a legal or beneficial interest in Franchisee and all officers, directors, executives, managers and members of the professional staff of Franchisee): (a) must not use the Trade Secrets or other Confidential Information in any other business or capacity; (b) mustmaintain the absolute confidentiality of the Trade Secrets and other Confidential Information during and after the term of this Agreement; (c) must not make any unauthorized copies of any portion of the Trade Secrets or other Confidential Information disclosed in written or other tangible form; and (d) must adopt and implement all reasonable procedures prescribed from time to time by Franchisor to prevent unauthorized use or disclosure of the Trade Secrets and other Confidential Information. Franchisee must enforce this Section as to its employees, agents and representatives and must be liable to Franchisor for any unauthorized disclosure or use of Trade Secrets or other Confidential Information by any of them.

b) "Confidential Information" means technical and non-technical information used in or related to TheGlass Guru Businesses and not commonly known by or available to the public, including, without limitation, Trade Secrets and any other information identified as confidential when delivered by Franchisor. Confidential Information does not include, however, any information that: (a) is now or subsequently becomes generally available to the public through no fault of Franchisee; (b) Franchisee can demonstrate was rightfully in its possession, without obligation of nondisclosure, prior to disclosure under this Agreement; (c) is independently developed without the use of any Confidential Information; or (d) is rightfully obtained from a third party who has the right, without obligation of nondisclosure, to transfer or disclose such information.

3. Non-Competition

- a) During the period Franchisee owns any interest in the Franchise Business (as defined in the Franchise Agreement), Franchisee must not, directly or indirectly, carry on, be engaged in or take part in, render services to, or own or share in the earnings of any Competitive Business anywhere without the express written consent of Franchisor;
- b) For a period to two (2) years after the termination of Franchisee's interest in Franchise Business, Franchisee must not, directly or indirectly, carry on, be engaged in or take part in, render services to, or own or share in the earnings of any Competitive Business located or operating (a) within a fifteen (15) mile radius of the Area of Primary Responsibility of Franchisee in existence at the timeof termination or expiration; or, (b) within a fifteen (15) mile radius of the location of any other Franchise Business in existence at the time of termination or expiration.
- c) Franchisee must not solicit or otherwise attempt to induce or influence any employee or otherbusiness associate of Franchisee or Franchisor to terminate or modify his, her or its business relationship with Franchisee or Franchisor or to compete with Franchisee or Franchisor.

d) "Competitive Business" means any business, whether direct or indirect, that offers (or grants franchises or licenses to others to operate a business that offers) products and services the same as or similar to those provided by Franchise Business or in which Trade Secrets or other Confidential Information could be used to the disadvantage of Franchisor, any Affiliate or its otherfranchisees; provided, however, that the term "Competitive Business" must not apply to (a) any business operated by Franchisee under a Franchise Agreement with Franchisor, or (b) any business operated by a publicly-held entity in which Franchisee owns less than a five percent (5%) legal or beneficial interest.

4. Post Termination

a) Franchisee obligations under paragraphs 2(a), 3(b) and 3(c) of this Agreement must continue in effect after termination of Franchisee's relationship with Franchisor, regardless of the reason or reasons for termination, and whether such termination is voluntary or involuntary, and Franchisor is entitled to communicate Franchisee's obligations under this Agreement to any future Franchisee, vendor, customer or employer to the extent deemed necessary by Franchisor for protection of Franchisor's rights hereunder.

5. Miscellaneous

- a) This Agreement constitutes the entire Agreement between the parties with respect to the subject matter hereof. This Agreement cannot be altered or amended except by an agreement in writing signed by the duly authorized representatives of the parties.
- b) Franchisee must reimburse Franchisor for any, and all costs and attorney fees incurred by Franchisor in the enforcement of the terms of this Agreement.
- c) This Agreement will be effective as of the date this Agreement is executed and must be binding upon the successors and assigns of Franchisee and must inure to the benefit of Franchisor, its subsidiaries, successors, and assigns.
- d) The failure of either party to insist in any one (1) or more instances upon performance of any terms and conditions of this Agreement must not be construed a waiver of future performance of any such term, covenant or condition of this Agreement and the obligations of either party with respect to this Agreement must continue in full force and effect.
- e) The paragraph headings in this Agreement are included solely for convenience and must not affect, or be used in connection with, the interpretation of this Agreement.
- f) In the event that any part of this Agreement must be held to be unenforceable or invalid, the remaining parts hereof must nevertheless continue to be valid and enforceable as though the invalid portions were not a part hereof.

Franchisor has caused this Agreement to be executed by its duly authorized officer, and Franchisee has executed this Agreement, all being done in duplicate originals with one (1) original being delivered to each party as of the day and year first above written.

FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	Entity Name
BY:	BY:
NAME:	NAME:
TITLE:	TITLE:

EXHIBIT 4

TO THE FRANCHISE AGREEMENT THE GLASS GURU ENTERPRISES, INC.

GUARANTY AND ASSUMPTION OF OBLIGATIONS

THIS GUARANTY AND ASSUMPTION OF OBL	IGATIONS is given this date
In consideration of, and as an inducement to date herewith ("Agreement") by The Glass Gu personally and unconditionally guarantees to	the signing of that certain Franchise Agreement of even ru Enterprises, Inc. ("Franchisor"), each of the undersigned Franchisor and its successors and assigns, for the term of the Agreement, that:
perform each and every undertaking, agreer the undersigned must be personally bound provision in the Agreement, including those or refrain from taking specific actions or engasections 7 and 17 of the Agreement. Each acceptance by Franchisor of the foregoing indebtedness or non-performance of any obany party with respect to the indebtedness or right it may have to require that an action	ment and covenant described in the Agreement. Each of by, and personally liable for, Franchisee's breach of any relating to monetary obligations and obligations to take aging in specific activities, such as those contemplated by of the undersigned waives: (a) acceptance and notice of undertakings; (b) notice of demand for payment of any oligations guaranteed; (c) protest and notice of default to r non-performance of any obligations guaranteed; (d) any be brought against Franchisee or any other person as a renotices and legal or equitable defenses to which it may
Guaranty must be joint and several; (b) it must Agreement upon demand if Franchisee fails be contingent or conditioned upon pursuit other person or entity; and (d) such liability rany extension of time, credit or other including, or performance, or the compromise or release	ees that: (a) its direct and immediate liability under this st render any payment or performance required under the or refuses punctually to do so; (c) such liability must not by Franchisor of any remedies against Franchisee or any must not be diminished, relieved or otherwise affected by gence which Franchisor may from time to time grant to without limitation, the acceptance of any partial payment se of any claims, none of which must in any way modify or ing and irrevocable during the term of the Agreement.
This Guaranty has been entered into the day	and year first before written.
PERSONAL GUARANTOR	PERSONAL GUARANTOR
Personally, and Individually (Printed Name)	Personally, and Individually (Printed Name)
Personally and Individually (Signature)	Personally and Individually (Signature)
HOME ADDRESS:	HOME ADDRESS:
TELEPHONE:	TELEPHONE:

PERCENT OWNERSHIP IN FRANCHISE: _____ %

PERCENT OWNERSHIP IN FRANCHISE: _____%

PERSONAL GUARANTOR PERSONAL GUARANTOR Personally, and Individually (Printed Name) Personally, and Individually (Printed Name) Personally and Individually (Signature) Personally and Individually (Signature) HOME ADDRESS: HOME ADDRESS: TELEPHONE: TELEPHONE: PERCENT OWNERSHIP IN FRANCHISE: _____ % PERCENT OWNERSHIP IN FRANCHISE: _____ % **PERSONAL GUARANTOR PERSONAL GUARANTOR** Personally, and Individually (Printed Name) Personally, and Individually (Printed Name) Personally and Individually (Signature) Personally and Individually (Signature) HOME ADDRESS: HOME ADDRESS: TELEPHONE: TELEPHONE:

PERCENT OWNERSHIP IN FRANCHISE: _____ %

PERCENT OWNERSHIP IN FRANCHISE: _____ %

EXHIBIT 5

TO THE FRANCHISE AGREEMENTTHE GLASS GURU ENTERPRISES, INC.

HOLDERS OF LEGAL OR BENEFICIAL INTEREST IN FRANCHISEE; OFFICERS, DIRECTORS, AND MANAGERS

Holders of Legal or Beneficial Interest (i.e. Shareholders, Members):

NAME:	 NAME:	
TITLE:	TITLE:	
HOME ADDRESS:	 HOME ADDRESS:	
TELEPHONE:	TELEPHONE:	
EMAIL:	EMAIL:	
PERCENT OWNERSHIP IN FRANCHISE:	PERCENT OWNERSHIP IN FRANCHISE:	
NAME:	 NAME:	
TITLE:	TITLE:	
HOME ADDRESS:	HOME ADDRESS:	
TELEPHONE:	 TELEPHONE:	
EMAIL:	EMAIL:	
PERCENT OWNERSHIP IN FRANCHISE:	PERCENT OWNERSHIP IN FRANCHISE:	
NAME:	 NAME:	
TITLE:	TITLE:	
HOME ADDRESS:	 HOME ADDRESS:	
TELEPHONE:	 TELEPHONE:	
EMAIL:	EMAIL:	
PERCENT OWNERSHIP IN FRANCHISE:	PERCENT OWNERSHIP IN FRANCHISE:	

Officers, Directors, and Managers:

NAME:	NAME:
TITLE:	TITLE:
HOME ADDRESS:	HOME ADDRESS:
TELEPHONE:	TELEPHONE:
EMAIL:	EMAIL:
PERCENT OWNERSHIP IN FRANCHISE:%	PERCENT OWNERSHIP IN FRANCHISE:%
NAME:	NAME:
TITLE:	TITLE:
HOME ADDRESS:	HOME ADDRESS:
TELEPHONE:	TELEPHONE:
EMAIL:	EMAIL:
PERCENT OWNERSHIP IN FRANCHISE:%	PERCENT OWNERSHIP IN FRANCHISE:%
NAME:	NAME:
TITLE:	TITLE:
HOME ADDRESS:	HOME ADDRESS:
TELEPHONE:	TELEPHONE:
EMAIL:	EMAIL:
PERCENT OWNERSHIP IN FRANCHISE:%	PERCENT OWNERSHIP IN FRANCHISE:%

EXHIBIT 6

TO THE FRANCHISE AGREEMENT THE GLASS GURU ENTERPRISES, INC.

ASSIGNMENT OF TELEPHONE NUMBERS AND INTERNET TOOLS

This Assignment is made between The Glass Guru Enterprises, Inc. of 5550 Granite Parkway,
Suite 280, Plano, Texas 75024 ("Franchisor") and
whose address is
("Franchisee").

- 1. Introduction: Franchisee has obtained a license from Franchisor for the operation of a businessusing Franchisor's THE GLASS GURU Franchise Business system ("System"), which business Franchisee acquired by signing a Franchise Agreement dated_______ (the "Franchise Agreement"). In consideration of Franchisor granting the license to Franchisee, Franchisee has agreed to assign all Telephone Numbers and Internet Tools (as defined below) that are associated with Franchisee's THE GLASS GURU Franchise Business (the "Franchise Business") and/or the System to Franchisor. For purposes of this Agreement, "Telephone Numbers" includes all telephone numbers used in connection with the Franchise Business, including in connection with advertising and marketing for the Franchise Business. For purposes of this Agreement, "Internet Tools" means Internet domain names, email addresses, websites, social media accounts (such as Facebook, LinkedIn, Twitter, YouTube), blogs, vlogs (social videos), online social networks, wikis, forums, content sharing communities, and other Internet tools used in connection with the Franchise Business, including in connection with advertising and marketing for the Franchise Business.
- 2. **Assignment of Telephone Numbers/Power of Attorney:** Franchisee assigns all Telephone Numbers to Franchisor or its successor or assign. Franchisee hereby appoints an officer of Franchisor as Franchisee's attorney-in-fact to transfer the Telephone Numbers to Franchisor and to sign on behalf of Franchisee, all documents necessary to accomplish the transfer.
- 3. **Assignment of Internet Tools/Power of Attorney:** Franchisee assigns all Internet Tools to Franchisor or its successor or assign. Franchisee also hereby appoints an officer of Franchisor as Franchisee's attorney-in-fact to transfer the Internet Tools to Franchisor and to sign on behalf of Franchisee, all documents necessary to accomplish the transfer.
- 4. Limited License; Responsibility for Costs: Franchisor grants Franchisee a limited license to usethe Telephone Numbers and Internet Tools in connection with the Franchise Business only during the term of the Franchise Agreement and only as long as Franchisee complies with the policies and procedures specified by Franchisor. On the expiration without renewal or termination of the Franchise Agreement, this limited license will terminate, and Franchisee mustcease all use of the Telephone Numbers and Internet Tools. On the termination of this license, Franchisee must cooperate with Franchisor and provide any authorizations as may be necessaryfor Franchisor to assert its rights in the Telephone Numbers and Internet Tools. While this limited license is in effect, Franchisee is responsible for all costs associated with the Telephone Numbers and Internet Tools and, unless otherwise specified by Franchisor, must pay those costsdirectly to the providers of the Telephone Numbers and Internet Tools.
- 5. **Access to Telephone Numbers and Internet Tools:** Franchisor will have the right to access all accounts relating to the Telephone Numbers and Internet Tools. Franchisee mustprovide to Franchisor all information necessary to allow Franchisor to access those accounts, including usernames, passwords, security codes, and all changes to any of that information.

- 6. Consent: Franchisee hereby consents and authorizes any and all telephone companies, telephone directory services, Internet companies and other public or private businesses using, authorizing or providing any of the Telephone Numbers and Internet Tools to immediately recognize this Assignment upon receipt of written notice from Franchisor. Franchisee agrees that a copy of this Assignment, certified by an officer of Franchisor, will be as valid and binding as the original.
- 7. **Notices:** Franchisor may give notice of its acceptance of the Assignment of the Telephone Numbers and Internet Tools by sending written notice by first class mail and certified or registered mail with postage fully paid and depositing them in the United States Mails. Notices may be sent in accordance with this Section to Franchisee and to all telephone companies, Internet companies and other businesses that are to recognize the Assignment. All notices to the Franchisee must be addressed to the address indicated above, or to any subsequent address of which Franchisor receives written notice. Any notice delivered by mail in the manner set forth in this Section will be deemed delivered and received three days after mailing.
- 8. **Miscellaneous:** If any part of this Agreement is found to be unenforceable, such findings will not invalidate the other parts of this Agreement. This Agreement expresses the entire understanding of the parties with respect to the subject matter herein. This Agreement will be construed in accordance with the laws of the State of Texas and will be deemed to have been made in the State of Texas. This Agreement may not be changed orally, but only by an agreement in writing and signed by the party against whom enforcement of any change is sought.

Exhibit 6 signed and effective this date: ______.

FRANCHISOR:	FRANCHISEE:	
THE GLASS GURU ENTERPRISES, INC.		
	Entity Name	
BY:		
NAME:	BY:	
TITLE:	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	

EXHIBIT 7

TO THE FRANCHISE AGREEMENT THE GLASS GURU ENTERPRISES, INC.

ADDENDUM TO LEASE AND COLLATERAL ASSIGMENT OF LEASE

This P	Addend	dum to) Lease, d	ated			_ , 20	_	, is entered	d into by and
betwe	een								<u> </u>	("Lessor")
and _										("Lessee").
	A.	The	parties	hereto	have	entered	into a	certain	Lease A	greement,
dated,					20	,and	pertainir	ng to th	ne premise	es located
at								("Location"). Lessor
acknowl	ledges	that I	Lessee in	tends to	operat	te a The G	lass Guru	Franchi	se Business	from the
leased p	oremise	es ("Pı	remises")	pursuar	nt to a	Franchise	Agreeme	ent (" Fra i	nchise Ag	reement")
-				•			•		under the	
			•			•		-		Business").

B. The parties now desire to amend the Lease in accordance with the terms and conditions contained herein.

NOW, THEREFORE, it is hereby mutually covenanted and agreed betweenLessor and Lessee as follows:

- 1. Use of Premises. Lessor and Lessee agree that, during the term of the Franchise Agreement, the Premises shall be used only for the operation of a The Glass Guru Franchise Business under a Franchise Agreement with Franchisor.
- 2. Signs. Lessor agrees that Lessee shall have the right to display the proprietary marks and signs on the interior and exterior of the Premises as Lessee is reasonably required to do pursuant to the Franchise Agreement and any successor Franchise Agreement under which Lessee may operate a Franchise Business on the Premises.
- 3. Franchisor's Right to Enter. Lessor and Lessee agree that the employees of Franchisor, or its affiliates, shall have the right to enter Premises to make any modifications necessary to protect their proprietary marks.
- 4. Assignment. Lessee shall have the right to assign all of its right, title and interest in the Lease to Franchisor or its affiliate, or another franchisee, at any time during the term of the Lease, including any extensions or renewals thereof, without first obtaining Lessor's consent in accordance with the Collateral Assignment of Lease attached hereto as Attachment A (the "Collateral Assignment"). However, no assignment shall be effective until the time as Franchisor or its designated affiliate or franchisee gives Lessor written notice of its acceptance of the assignment, and nothing contained herein or in any other document shall constitute Franchisor or its designated transferee a party to the Lease, or guarantor thereof, and shall

not create any liability or obligation of Franchisor or its designated transferee unless and until the Lease is assigned to, and accepted in writing by, Franchisor or its designated transferee. In the event of any assignment, Lessee shall remain liable under the terms of the Lease. Franchisor shall have the right to reassign the Lease to another franchisee without the Lessor's consent in accordance with this Section.

5. Default and Notice.

- a) In the event there is a default or violation by Lessee under the termsof the Lease, Lessor shall give Lessee and Franchisor written notice of the default or violation within a reasonable time after Lessor receives knowledge of its occurrence.If Lessor gives Lessee a default notice, Lessor shall contemporaneously give Franchisor a copy of the notice. Franchisor shall have the right, but not the obligation, to cure the default. Franchisor will notify Lessor whether it intends to cure the default andtake an automatic assignment of Lessee's interest as provided in the Collateral Assignment. Franchisor will have an additional fifteen (15) days from the expiration of Lessee's cure period in which it may exercise the option, but it is not obligated, to cure the default or violation.
- b) All notices to Franchisor shall be sent by registered or certified U.S. mail, postage prepaid, to the following address:

The Glass Guru Enterprises, Inc. 5550 Granite Parkway Suite 280 Plano, TX 75024 Attn: Dan Frey, President

Franchisor may change its address for receiving notices by giving Lessor writtennotice of the new address. Lessor agrees that it will notify both Lessee and Franchisor of any change in Lessor's mailing address to which notices should be sent.

c) Following Franchisor's approval of the Lease, Lessee agrees not to terminate, or in any way alter or amend the same during the term of the Franchise Agreement, including any renewal thereof, without Franchisor's prior written consent, and any attempted termination, alteration or amendment shall be null and void and have no effect as to Franchisor's interests thereunder; and a clause to the effect shall be included in the Lease.

6. Termination or Expiration.

a) Upon Lessee's default and failure to cure the default within the applicable cure period, if any, under either the Lease or the Franchise Agreement, Franchisor will, at its option, have the right, but not the obligation, to take an automatic assignment of Lessee's interest and at any later time to re-assign the Lease to a new franchisee without Lessor's consent and to be fully released from any and all liability to Lessor upon the reassignment, provided the franchisee agrees to assume Lessee's obligations and the Lease.

b) Upon the expiration or termination of either the Lease or the Franchise Agreement, Lessor will cooperate with and assist Franchisor in securing possession of the Premises and if Franchisor does not elect to take an assignment of the Lessee's interest, Lessor will allow Franchisor to enter the Premises, without being guilty of trespass and without incurring any liability to Lessor, to remove all signs and all other items identifying the Premises as a Franchise Business and to make other modifications as are reasonably necessary to protect the Franchisor's marks and system. In the event Franchisor exercises its option to purchase assets of Lessee, Lessor shall permit Franchisor to remove all the assets being purchased by Franchisor.

7. Consideration; No Liability.

- a) Lessor hereby acknowledges that the provisions of this Addendum to Lease are required pursuant to the Franchise Agreement under which Lessee plans to operate its business and Lessee would not lease the Premises without this Addendum. Lessor also hereby consents to the Collateral Assignment from Lessee to Franchisor as evidenced by Attachment A hereto.
- b) Lessor further acknowledges that Lessee is not an agent or employee of Franchisor and Lessee has no authority or power to act for, or to create any liability on behalf of, or to in any way bind Franchisor or any affiliate of Franchisor, and that Lessor has entered into this Addendum to Lease with full understanding that it creates no duties, obligations or liabilities of or against Franchisor or any affiliate of Franchisor.
- 8. Sales Reports. If requested by Franchisor, Lessor will provide Franchisor with whatever reports, information or data Lessor has regarding Lessee's sales from its Franchise Business
- 9. Amendments. No amendment or variation of the terms of the Lease or this Addendum to the Lease shall be valid unless made in writing and signed by the parties hereto.
- 10. Reaffirmation of Lease. Except as amended or modified herein, all of the t e r m s, conditions and covenants of the Lease shall remain in full force and effect and are incorporated herein by reference and made a part of this Addendum as though copies herein in full.
- 11. Beneficiary. Lessor and Lessee expressly agree that Franchisor is a third-party beneficiary of this Addendum.

LESSOR:	LESSEE:	
Name: Title	 Name: Title	

IN WITNESS WHEREOF, witness the signatures of the parties hereto as of the day, month

and year first written above.

Attachment A COLLATERAL ASSIGNMENT OF LEASE

FOR VALUE RECEIVED, as of thetheundersigned,	day_of	, 20 " Assignor ") ber	("Effective Date"),
and sets over unto The Glass Guru Er Assignor's right, title and interest as te is attached hereto as Attachment A ("	nterprises, Inc., a enant, in, to and u	Texas corporations and a certain that certain the certain that certain the cer	tion (" Assignee ") all of n lease, a copy of which
This Collateral Assignment of Le only and except as specified herein, As whatsoever arising from or in connection take possession of the premises demise assume the obligations of Assignor there	ssignee shall hav on with this Colla ed by the Lease	e no liability or teral Assignmer	obligation of any kind at unless Assignee shall
Assignor represents and warrant assign the Lease and its interest therein a to, assign or transfer any of its interest in	and that Assignor	has not previou	sly, and is not obligated
Upon a default by Assignor under a The Glass Guru Franchise Business be or in the event of a default by Assignor un Agreement, Assignee shall have the rig premises demised by the Lease, expel As no further right, title or interest in the L	tween Assignee ander any document that and is hereby ssignor therefror	and Assignor (" F nt or instrument rempowered to	ranchise Agreement"), securing the Franchise take possession of the
Assignor agrees it will not suffer modification of the Lease without the pri Franchise Agreement and any renewals all options to extend the term of or rene that said option must be exercised, unless Assignee to otherwise agree in writing renew the Lease as stated herein, Assign lawful attorney-in-fact, which appoint extension or renewal options in the name effecting the extension or renewal.	or written consent thereto, Assigno w the Lease at lease at lease ss Assignee othe and upon failunor hereby irrevo atment is couple	nt of Assignee. To agrees that it seast thirty (30) or rwise agrees in vore of Assignor to eably appoints Act with an int	Through the term of the shall elect and exercise lays before the last day writing. Upon failure of o so elect to extend or Assignee as its true and erest, to exercise the
IN WITNESS WHEREOF , Assignor and A as of the Effective Date first above writ	-	ned this Collate	ralAssignment of Lease
ASSIGNOR:	ASSIG	NEE:	
	THE G	LASS GURU EN	TERPRISES, INC.
Ву:	By:	_	

Title:_____

EXHIBIT 8

TO THE FRANCHISE AGREEMENT THE GLASS GURU ENTERPRISES, INC.

MULTI-STATE ADDENDA

ADDENDUM TO THE FRANCHISE AGREEMENT THE GLASS GURU ENTERPRISES, INC.

FOR THE STATE OF CALIFORNIA

This Addendum to the Franchise Agreement is agreed to this date	
between The Glass Guru Enterprises, Inc. and	

- 1. New Section 17.6 is inserted into the Franchise Agreement and states as follows:
 - If termination is the result of Franchisee's default, Franchisee will pay to Franchisor a lump sum payment (as liquidated damages for causing the premature termination of this Agreement and notas a penalty) equal to the total of all Royalty Fee payments for: (a) the twenty-four (24) calendar months of operation of Franchisee preceding Franchisee's default; (b) the period of time Franchisee has been in operation preceding the notice, if less than twenty-four (24) calendar months, projected on a twenty-four (24) calendar month basis; or (c) any shorter period as equals the unexpired term at the time of termination. The parties acknowledge that a precise calculation of the full extent of the damages that Franchisor will incur on termination of this Agreement as a result of Franchisee's default is difficult and the parties desire certainty in this matter and acknowledge that the lump sum payment provided under this Section is reasonable in light of the damages for premature termination that Franchisor will incur. This payment is not exclusive of any other remedies that Franchisor may have, including attorneys' fees and costs.
- 2. In recognition of the requirements of the California Franchise Investment Law, Cal. Corp. Code §§31000-3516 and the California Franchise Relations Act, Cal. Bus. And Prof. Code §§20000-20043, the FranchiseAgreement for The Glass Guru Enterprises, Inc. is amended as follows:
 - The California Franchise Relations Act provides rights to the Franchisee concerning termination or non-renewal of the Franchise Agreement, which may supersede provisions in the Franchise Agreement, specifically Sections 4.2 and 16.2.
 - Section 16.2.1.10, which terminates the Franchise Agreement upon the bankruptcy of Franchisee, may not be enforceable under federal bankruptcy law (11 U.S.C. Section 101, etseg.).
 - Section 16.2.2.1 which terminates the Franchise Agreement upon (5) days of receiving noticeof Franchisee's failure to pay amounts due Franchisor, Franchisor's services provided for saidtime period if default is not cured.
 - Section 16.2.2.2 which terminates the Franchise Agreement upon (10) days of receiving notice of Franchisee's failure to maintain insurance as specified in Section 15 of this Agreement, Franchisor's service provided for said time period if default is not cured; or
 - Section 16.2.2.3 which terminates the Franchise Agreement within thirty (30) days of receiving notice of Franchisee's failure to achieve acceptable status on other compliance requirements; or
 - Section 16.2.2.4 which terminates the Franchise Agreement within thirty (30) days of receiving notice of Franchisee's failure to use and cooperate in the use of operating systems and tools provided by Franchisor to improve the Franchise Business and/or the franchise systemas a whole; or
 - Section 16.2.2.5 which terminates the Franchise Agreement within thirty (30) days of receiving notice
 of Franchisee's failure to regularly attend and actively participate in required conference calls,
 or meetings, conventions, and other events sponsored or suggested by Franchisor to increase
 the chance of success and/or maximize the performance of the Franchise Business and/or the
 franchise system as a whole; or
 - Section 16.2.2.6 which terminates the Franchise Agreement within thirty (30) days of receiving notice of Franchisee's failure to embrace new programs and business building initiatives developed for the enhancement of the performance of the Franchise Business; or

- Section 16.2.2.7 which terminates the Franchise Agreement within thirty (30) days of receiving notice of Franchisee's regularly acting in a combative or confrontational manner with vendors, customers, or other franchisees; or
- Section 16.2.2.8 which terminates within thirty (30) days of receiving notice that Franchisee has had an excessive number of customer complaints and/or has not acted reasonably and in the best interests of the franchise system in resolving customer complaints.
- Section 16.2.2.9 which terminates the Franchise Agreement upon (30) days of receiving notice of
 any other default by Franchisee's failure to comply with any mandatory specification, standard
 or operating procedure prescribed in the Manual or otherwise prescribed in writing, Franchisor's
 services provided for said time period if default is not cured.
- Section 17.2 contains a covenant not to compete that extends beyond the expiration or termination of the Agreement; this covenant may not be enforceable under California Law.
- Paragraph 1 of this Addendum contains a liquidated damages clause. Under California CivilCode Section 1671, certain liquidated damages clauses are unenforceable.
- Section 23.7 requires binding arbitration. The arbitration must occur at the forum indicatedin Section 23.7, with the costs being borne by the non-prevailing party. Prospective franchisees are encouraged to consult legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of the Franchise Agreement restricting venue to a forum outside of the State of California.
- 3. California law requires that you obtain a Contractor's license from the California Contractors State License Board (CSLB) if the total cost (labor and materials) of one or more contracts on the project is \$500 or more. Licenses may be issued to individuals, partnerships, corporations joint ventures, or Limited Liability Companies (LLCs).
- 4. To the extent this Addendum must be deemed to be inconsistent with any terms or conditions of saidFranchise Agreement or exhibits or attachments to the Franchise Agreement, the terms of this Addendum must govern.

Each of the undersigned acknowledges having read this California Addendum, and understands and consents to be bound by all of its terms.

FRANCHISOR:	FRANCHISEE:	
THE GLASS GURU ENTERPRISES, INC.		
	Entity Name	
BY:	BY:	
NAME:	NAME:	
TITLE:	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	11122.	
	BY:	
	NAME:	
	TITLE:	

FOR THE STATE OF HAWAII

This Addendum to the Franchise Agreement is agreed to this date	
petween The Glass Guru Enterprises, Inc. and	

- 1. In recognition of the requirements of the Hawaii Franchise Investment Law, Hawaii Revised Statutes, Title 26, Chapter 482E *et seq.*, the Franchise Agreement for The Glass Guru Enterprises, Inc. is amended as follows:
- The Hawaii Franchise Investment Law provides rights to you concerning non-renewal, terminationand transfer of the Franchise Agreement. If the Agreement, and more specifically Sections 4.2, 16.2and
 - 18.2 contain a provision that is inconsistent with the Hawaii Franchise Investment Law, the HawaiiFranchise Investment Law must control.
- Sections 4.2.8, 18.2.3 and 18.2.6 require Franchisee to sign a general release as a condition of renewal
 and transfer of the franchise, and Sections 5.2, 5.5 and 8.4 require Franchisee to sign a general
 release as a condition to receiving a refund of a portion of the Franchise Fee following a
 termination of thefranchise; such release must exclude claims arising under the Hawaii Franchise
 Investment Law.
- Section 16.2, which terminates the Franchise Agreement upon the bankruptcy of the Franchisee, maynot be enforceable under federal bankruptcy law (11 U.S.C. Section 101, et seq.).
- 2. Each provision of this Addendum is effective only to the extent that the jurisdictional requirements of the Hawaii Franchise Investment Law are met independently of this Addendum. To the extent this Addendum must be deemed to be inconsistent with any terms or conditions of said Franchise Agreement or exhibits or attachments to that Agreement, the terms of this Addendum must govern.

Each of the undersigned acknowledges having read this Hawaii Addendum, and understands and consents tobe bound by all of its terms.

FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	Entity Name
BY: NAME: TITLE:	BY: NAME: TITLE:
	BY: NAME: TITLE:
	BY: NAME: TITLE:
	BY: NAME: TITLE:

FOR THE STATE OF ILLINOIS

This Addendum to the Franchise Agreement is agreed to this date	
between The Glass Guru Enterprises, Inc. and	

- 1. In recognition of the requirements of the Illinois Franchise Disclosure Act, 815 ILCS 705, the FranchiseAgreement for The Glass Guru Enterprises, Inc. is amended as follows:
- Sections 4.2, 8.2 and 18.2 are amended to add: No general release shall be required as a condition of renewal, termination and/or transfer that is intended to require Franchisee to waive compliance with the Illinois Franchise Disclosure Act, 815 ILCS 705.
- Sections 16.2, 17.2 and 23 are amended to add: The conditions under which the Franchise Agreementcan be terminated and your rights upon termination or non-renewal, as well as the application by which you must bring any claims, may be governed by the Illinois Franchise Disclosure Act, 815 ILCS 705/19 and 705/20.
- Sections 23.1 and 23.2 are amended to add: The Franchise Agreement shall be governed by Illinois law. Jurisdiction and venue for court litigations shall be in Illinois. Any provision in the Franchise Agreement that designates jurisdiction or venue in a forum outside the State is void, provided that a Franchise Agreement may provide for arbitration in a forum outside of Illinois.
- Section 23.4 is amended to add: No action for liability under the Illinois Franchise Disclosure Act shall be maintained unless brought before the expiration of three (3) years after the act or transaction constituting the violation upon which it is based, the expiration of one (1) year after the Franchisee becomes aware of facts or circumstances reasonably indicating that he may have a claim for relief in respect to conduct governed by the Act, or ninety (90) days after delivery to the Franchisee of a writtennotice disclosing the violation, whichever shall first expire.
- Section 23.6 is deleted in its entirety.
- 2. The Franchise Agreement is amended to add the following: Any condition, stipulation, or provision purporting to bind any person acquiring any franchise to waive compliance with anyprovision of the Illinois Franchise Disclosure Act or any other law of the State of Illinois is void.
- 3. Each provision of this Addendum is effective only to the extent that the jurisdictional requirements of the Illinois Franchise Disclosure Act are met independently of this Addendum. To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of said Franchise Agreement or exhibits or attachments to that Agreement, the terms of this Addendum shall govern.
- 4. No statement, questionnaire or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on behalf of the Franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Each of the undersigned acknowledges having read this Illinois Addendum and understands and consents tobe bound by all of its terms.

FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	
	Entity Name
BY:	BY:
NAME:	NAME:
TITLE:	
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:

FOR THE STATE OF INDIANA

This Addendum to the Franchise Agreement is agreed to this date	
between The Glass Guru Enterprises, Inc. and	

- 1. In recognition of the requirements of the Indiana Deceptive Franchise Practices Law, IC 23-2.2.7 and the Indiana Franchise Disclosure Law, IC 23-2-2-2.5, the Franchise Agreement for The Glass Guru Enterprises, Inc. is amended as follows:
 - Sections 4.2, 8.2 and 18.2 do not provide for a prospective general release of claims against Franchisorthat may be subject to the Indiana Deceptive Franchise Practices Law or the Indiana Franchise Disclosure Law.
 - Section 16 is amended to prohibit unlawful unilateral termination of a franchise unless there is a material violation of the Franchise Agreement and termination is not in bad faith.
 - Section 17.2 is amended subject to Indiana Code 23-2-2.7-1(9) to provide that post-term non-competitor covenants shall have a geographical limitation of the territory granted to Franchisee.
 - Section 23.1 is amended to provide that Franchisee shall not be required to indemnify Franchisor for any liability imposed upon Franchisor as a result of Franchisee's reliance upon or use of procedures or products which were required by Franchisor, if such procedures or products were utilized by Franchisee in the manner required by Franchisor.
 - Section 23.1 is amended to provide that, in the event of a conflict of law, the Indiana Franchise Disclosure Law, IC 23-2-2.5, and the Indiana Deceptive Franchise Practices Law shall prevail.
 - Section 23.2 is amended to provide that Franchisee may commence litigation in Indiana for any cause of action under Indiana law.
 - Section 23.7 is amended to provide that arbitration between Franchisor and Franchisee, shall be conducted at a mutually agreed upon location.
- 2. Each provision of this Addendum is effective only to the extent that the jurisdictional requirements of the Indiana Law applicable to the provisions are met independently of this Addendum. To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of said Franchise Agreement or exhibits or attachments to that Agreement, the terms of this Addendum shall govern.

Each of the undersigned acknowledges having read this Indiana Addendum and understands and consents tobe bound by all of its terms.

FRANCHISOR:	FRANCHISEE:	
THE GLASS GURU ENTERPRISES, INC.		
	Entity Name	
BY:	BY:	
NAME:	NAME:	
TITLE:		
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITI E-	

FOR THE STATE OF MARYLAND

This Addendum to the Franchise Agreement is agreed to this date	
between The Glass Guru Enterprises, Inc. and	

- 1. In recognition of the requirements of the Maryland Franchise Registration and Disclosure Law, Md. Code Ann., Bus. Reg. §§14-201-14-233, the Franchise Agreement for The Glass Guru Enterprises, Inc. is amended as follows:
 - Sections 4.2, 18.2.3 and 18.2.6 require Franchisee to sign a general release as a condition of renewal, termination and transfer of the franchise,; such release shall exclude claims arising under the Maryland Franchise Registration and Disclosure Law.
 - Section 16.2.1.10, which terminates the Franchise Agreement upon the bankruptcy of the Franchisee, may not be enforceable under federal bankruptcy law (11 U.S.C. Section 101, et seq.).
 - Section 23.1 requires that the franchise be governed by the laws of the State of Texas; however,in
 the event of a conflict of laws to the extent required by the Maryland Franchise Registration
 and Disclosure Law, the laws of the State of Maryland shall prevail.
 - Sections 23.2 and 23.7 require litigation or arbitration to be conducted in the State of Texas; the requirement shall not limit any rights Franchisee may have under the Maryland Franchise Registration and Disclosure Law to bring suit in the State of Maryland.
 - Any Section of the Franchise Agreement requiring Franchisee to assent to any release, estoppel or
 waiver of liability as a condition of purchasing the Franchise are not intended to, nor shall they act as
 a, release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and
 Disclosure Law.
 - Section 23.4 is amended to the extent that any claims arising under the Maryland Franchise Registration and Disclosure Law may be brought within three (3) years after the grant of the franchise.
- 2. Any portion of the Franchise Agreement which requires prospective franchisees to disclaim the occurrence and/or acknowledge the non-occurrence of acts would constitute a violation of the Maryland Franchise Registration and Disclosure Law. Any such representations are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.
- 3. Each provision of this Addendum is effective only to the extent that the jurisdictional requirements of the Maryland Franchise Registration and Disclosure Law applicable to the provisions are met independently of this Addendum. To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of said Franchise Agreement or exhibits or attachments to that Agreement, the terms of this Addendum shall govern.
- 4. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Each of the undersigned acknowledges having read this Maryland Addendum and understands and consents to be bound by all of its terms.

FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	Entity Name
BY: NAME: TITLE:	BY: NAME: TITLE:
	BY: NAME: TITLE:
	BY: NAME: TITLE:
	BY: NAME: TITLE:

FOR THE STATE OF MINNESOTA

This Addendum to the Franchise Agreement is agreed to this date	
between The Glass Guru Enterprises, Inc. and	

- 1. In recognition of the Minnesota Franchise Law, Minn. Stat., Chapter 80C, Sections 80C.01 through 80C.22, and the Rules and Regulations promulgated pursuant thereto by the Minnesota Commission of Securities, Minnesota Rule 2860.4400, et. seq., the parties to the attached Franchise Agreement agree as follows:
 - Sections 4.2 and 16.2 are amended to add that with respect to franchises governed by MinnesotaLaw, Franchisor shall comply with the Minnesota Franchise Law that requires, except in certain specified cases, that a Franchisee be given 90 days notice of termination (with 60 days to cure) and 180 days notice of non-renewal of the Agreement.
 - Sections 4.2.8, 5.2, 5.5, 8.4, 18.2.3 and 18.2.6 do not provide for a prospective general release of anyclaims against Franchisor that may be subject to the Minnesota Franchise Law. Minn. Rule 2860.4400D prohibits a franchisor from requiring a franchisee to assent to a general release.
 - Section 6 is amended to add that as required by Minnesota Franchise Act, Franchisor shall reimburse
 Franchisee for any costs incurred by Franchisee in the defense of Franchisee's right to use the
 Marks, so long as you were using the Marks in the manner authorized by Franchisor, and so
 long as Franchisor is timely notified of the claim and is given the right to manage the defense
 of the claim including the right to compromise, settle or otherwise resolve the claim, and to
 determine whether to appeal a final determination of the claim.
 - Section 23.4 is amended to state that any claim concerning the Franchise Business or this Agreement or any related agreement shall be barred unless an arbitration or an action for a claimthat cannot be the subject of arbitration is commenced within three (3) years from the date on which Franchisee or Franchisor knew or should have known, in the exercise of reasonable diligence, of the facts giving rise to or the claim.
 - Section 23.5 is deleted in its entirety.
 - Section 23.6 is deleted in its entirety.
 - Minn. Stat. §80C.21 and Minn. Rule 2860.4400J prohibit Franchisor from requiring litigation to be conducted outside Minnesota. In addition, nothing in the Disclosure Document or Franchise Agreement can abrogate or reduce any of Franchisee's rights as provided for in Minnesota Statutes, Chapter 80C, or Franchisee's rights to any procedure, forum or remedies provided for by the lawsof the jurisdiction.
- 2. Each provision of this Addendum is effective only to the extent that the jurisdictional requirements of the Minnesota Franchise Law applicable to the provisions are met independently of this Addendum. To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of said Franchise Agreement or exhibits or attachments to them, the terms of this Addendum shall govern.

Each of the undersigned acknowledges having read this Minnesota Addendum and understands and consents tobe bound by all of its terms.

FRANCHISOR:	FRANCHISEE:	
THE GLASS GURU ENTERPRISES, INC.	Entity Name	
	Entity Name	
BY:	BY:	
NAME:	NAME:	
TITLE:	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITLE:	

FOR THE STATE OF NEW YORK

This Addendum to the Franchise Agreement is agreed to this date .	
between The Glass Guru Enterprises, Inc. and	

- 1. In recognition of the requirements of the General Business Laws of the State of New York, Article 33, §§680 through 695, the Franchise Agreement for The Glass Guru Enterprises, Inc. is amended as follows:
 - Sections 4.2.8, 5.2, 5.5, 8.4, 18.2.3 and 18.2.6 require Franchisee to sign a general release as a condition of renewal, termination and transfer of the franchise; such release shall exclude claims arising underthe General Business Laws.
 - Under Section 18.1, Franchisor shall not transfer and assign its rights and obligations under the
 Franchise Agreement unless the transferee will be able to perform the Franchisor's obligations
 underthe Franchise Agreement, in Franchisor's good faith judgment, so long as it remains
 subject to theGeneral Business Laws of the State of New York.
 - Section 21.3 is amended to provide that Franchisee shall not be required to indemnify
 Franchisor for any liability imposed upon Franchisor as a result of Franchisee's reliance upon or
 use of procedures or products that were required by Franchisor, if such procedures or products
 were utilized by Franchisee in the manner required by Franchisor.
 - Section 23.1 requires that the franchise be governed by the laws of the state the Franchisor's principalbusiness is then located, such a requirement shall not be considered a waiver of any right conferred upon the Franchisee by Article 33 of the General Business Laws.
 - Each provision of this Addendum is effective only to the extent that the jurisdictional requirements of the New York Law applicable to the provisions are met independently of this Addendum. To the extent this Addendum may be deemed to be inconsistent with any terms or conditions of said Franchise Agreement or exhibits or attachments to them, the terms of this Addendum shall govern.

Each of the undersigned acknowledges having read this New York Addendum and understands and consents tobe bound by all of its terms.

FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	Fatitu Nama
	Entity Name
BY:	BY:
NAME:	NAME:
TITLE:	TITLE:
	BY:
	NAME:
	TITLE:
	DV.
	BY: NAME:
	TITLE:
	BY:
	NAME:
	TITLE:

FOR THE STATE OF NORTH DAKOTA

This Addendum to the Franchise Agreement is agreed to this date	
between The Glass Guru Enterprises, Inc. and	

- 1. The North Dakota Securities Commission requires that certain provisions contained in the Agreement be amended to be consistent with North Dakota Law, including the North Dakota Franchise Investment Law, North Dakota Century Code Addendum, Chapter 51-19, Sections 51-19-01 *et seq.* Such provisions in the Agreement are amended as follows:
- Under Sections 4.2.8, 5.2, 5.5, 8.4, 18.2.3 and 18.2.6, the signing of a general release upon renewal, transfer,or termination of the franchise will be inapplicable to franchises operating under the North Dakota Franchise Investment Law to the extent that such a release excludes claims arising under the North DakotaFranchise Investment Law.
- Section 7 is amended to add that the prevailing party in any enforcement action is entitled to recoverall costs and expenses including attorneys' fees.
- Sections 17.1.6 and 17.1.7 are amended to state:
- If Franchisor or Franchisee is required to enforce this Agreement via judicial or arbitration proceedings, the prevailing party shall be entitled to reimbursement of its costs, including reasonable accounting and legal fees in connection with such proceeding.
- Section 17.2 is amended to add that covenants not to compete upon termination or expiration of theFranchise Agreement are generally unenforceable in the State of North Dakota except in limited instances as provided by law.
- Section 23.1 is amended to state that in the event of a conflict of laws, North Dakota Law shall prevail.
- Section 23.2 is amended to add that any action may be brought in the appropriate state or federal courtin North Dakota with respect to claims under North Dakota Law.
- Section 23.4 is amended to state that the statute of limitations under North Dakota Law shall apply.
- Sections 23.5 and 23.6 are deleted in their entireties.
- Section 23.7 is amended to state that arbitration involving a franchise purchased in North Dakota must be held either in a location mutually agreed upon prior to the arbitration, or if the parties cannot agreeon a location, the arbitrator shall determine the location.
- 2. Each provision of this Addendum is effective only to the extent that the jurisdictional requirements of the North Dakota Law applicable to the provisions are met independently of this Addendum. To the extentthis Addendum shall be deemed to be inconsistent with any terms or conditions of said Franchise Agreement or exhibits or attachments to them, the terms of this Addendum shall govern.

Each of the undersigned acknowledges having read this North Dakota Addendum and understands and consents tobe bound by all of its terms.

FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	
	Entity Name
BY:	BY:
NAME:	NAME:
TITLE:	TITLE:
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:

FOR THE STATE OF RHODE ISLAND

- 1. In recognition of the requirements of The Rhode Island Franchise Investment Act §19-28.1-14, the Franchise Agreement for The Glass Guru Enterprises, Inc. is amended as follows:
 - Sections 4.2.8, 5.2, 5.5, 8.4, 18.2.3, and 18.2.6 require Franchisee to sign a general release asa condition of renewal, transfer, or termination of the franchise; such release shall exclude claims arising under The Rhode Island Franchise Investment Act.
 - Sections 23.1, 23.2 and 23.7 are amended to state that restricting jurisdiction or venue to aforum outside the state of Rhode Island or requiring the application of the laws of anotherstate is void with respect to a claim otherwise enforceable under The Rhode Island Franchise Investment Act.
- 2. Each provision of this Addendum is effective only to the extent that the jurisdictional requirements of the Rhode Island Law applicable to the provisions are met independently of this Addendum. To the extentthis Addendum shall be deemed to be inconsistent with any terms or conditions of said Franchise Agreement or exhibits or attachments, the terms of this Addendum shall govern.

Each of the undersigned acknowledges having read this Rhode Island Addendum and understands and consents tobe bound by all of its terms.

FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	
	Entity Name
	BY:
BY:	NAME:
NAME:	TITLE:
TITLE:	_
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE:
	BY:
	NAME:
	TITLE

FOR THE COMMONWEALTH OF VIRGINIA

This Addendum to the Franchise Agreement	is agreed to this date
between The Glass Guru Enterprises, Inc. and	

amend and revise said Franchise Agreement as follows:

- Section 16.2.1.10, which terminates the Franchise Agreement upon the bankruptcy of the Franchisee, may not be enforceable under federal bankruptcy law (11 U.S.C. Section 101, et seq.)
- In recognition of the restrictions contained in Section 13.1-564 of the Retail Franchising Act, the Franchise Disclosure Document for The Glass Guru Enterprises, Inc. for use in the Commonwealthof Virginia shall be amended as follows:

Additional Disclosure: The following statements are added to Item 17.h. Under Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any ground for default or termination stated in the franchise agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

Each of the undersigned acknowledges having read this Addendum and understands and consents to bebound by all of its terms.

FRANCHISOR:	FRANCHISEE:	
THE GLASS GURU ENTERPRISES, INC.	Entity Name	
BY:	BY:	
NAME:	NAME:	
TITLE:	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITI E.	

to

FOR THE STATE OF WASHINGTON

This Addendum to the Franchise Agreement hereby amends the Franchise Agreement and Related	d Agreements
and is agreed to this date	
between The Glass Guru Enterprises, Inc. and	

- In the event of a conflict of laws, the provisions of the Washington Franchise InvestmentProtection Act, Chapter 19.100 RCW will prevail.
- RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor
 including the areas of termination and renewal of your franchise. There may also be court decisions
 which may supersede the franchise agreement in your relationship with the franchisor including the
 areas of termination and renewal of your franchise.
- In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.
- A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.
- Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.
- Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceableagainst an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 peryear (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.
- Section 23.5 is removed in its entirety.

- RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) solicitingor hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.
- Franchisor will defer collection of the initial franchise fee until the franchisor has fulfilled its initial preopening obligations to the franchisee and the franchisee is open for business. Please note that the initial fees for the purposes of the deferral include all initial franchisee fees described in Item 5 of the Franchise Disclosure Document.
- No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

The undersigned does hereby acknowledge receipt of this addendum.

Dated thisday of	20
FRANCHISOR:	FRANCHISEE:
THE GLASS GURU ENTERPRISES, INC.	Entity Name
BY: NAME: TITLE:	BY: NAME: TITLE:
	BY: NAME: TITLE:
	BY: NAME: TITLE:
	BY: NAME: TITLE:

FOR THE STATE OF WISCONSIN

This Addendum to the Franchise Agreement is agreed to this date	
between The Glass Guru Enterprises, Inc. and	

to amend and revise said Franchise Agreement as follows:

- 1. The Wisconsin Fair Dealership Law Title XIV-A Ch. 135, Sec. 135.01-135.07 shall supersede any conflicting terms of the Franchise Agreement.
- 2. This provision of this Addendum is effective only to the extent that the jurisdictional requirements of the Wisconsin Fair Dealership Law applicable to the provisions are met independently of this Addendum. To the extent this Addendum shall be deemed to be inconsistent with any terms or conditions of saidFranchise Agreement or exhibits or attachments, the terms of this Addendum shall govern.

Each of the undersigned acknowledges having read this Addendum and understands and consents to bebound by all of its terms.

FRANCHISOR:	FRANCHISEE:	
THE GLASS GURU ENTERPRISES, INC.		
	Entity Name	
BY:	BY:	
NAME:	NAME:	
TITLE:	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITLE:	
	BY:	
	NAME:	
	TITI C.	

EXHIBIT B

TO THE DISCLOSURE DOCUMENT THE GLASS GURU ENTERPRISES, INC.

LIST OF STATE ADMINISTRATORS

The following is a list of state administrators responsible for registration and review of franchises and the Effective Date of this Disclosure Document for these states. The Effective Date of this Disclosure Document for any state that is not included in this list is as shown on the cover of the Disclosure Document. We may register in one or more of these states.

California

Department of Financial Protection and Innovation One Sansome St., Suite 600 San Francisco, California 94104 (415)972-8565

Department of Financial Protection and Innovation 320 W. 4th Street, Suite 750 Los Angeles, California 90013 (213)576-7500

Department of Financial Protection and Innovation 2101 Arena Blvd. Sacramento, California 95834 (866) 275-2677 Toll Free

Florida

Division of Consumer Services Attn: Business Opportunities 2005 Apalachee Parkway Tallahassee, Florida 32399-6500

Hawaii

Commissioner of Securities Department of Commerce & Consumer Affairs 335 Merchant Street, Room 203 Honolulu, Hawaii 96813 (808) 586-2722

Illinois

Illinois Attorney General 500 South Second Street Springfield, Illinois 62701 (217)782-4465

Indiana

Indiana Secretary of State Securities Division 302 West Washington Street, Room E-111 Indianapolis, Indiana 46204 (317)232-6681

Kentucky

Office of the Attorney General Consumer Protection Division Attn: Business Opportunity 1024 Capital Center Drive Frankfort, Kentucky 40601-8204

Maryland

Office of the Attorney General Securities Division 200 St. Paul Place Baltimore, Maryland 21202 (410)576-6360

<u>Michigan</u>

Department of the Attorney General Consumer Protection Division, Franchise Unit 525 Ottawa Street Mennen Williams Building, 1st Floor Lansing, Michigan 48933 (517)373-7117

Minnesota

Minnesota Department of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 539-1500

<u>Nebraska</u>

Nebraska Department of Banking and Finance Commerce Court 1230 O Street, Suite 400 Lincoln, Nebraska 68509 (402)471-2171

New York

NYS Department of Law Investor Protection Bureau Franchise Section 28 Liberty Street, 21st Floor New York, New York 10271 (212)416-8285

North Dakota

North Dakota Securities Department 600 East Boulevard Avenue State Capitol – 5th Floor Department 414 Bismarck, North Dakota 58505-0510 (701)328-4712

Rhode Island

Department of Business Regulation Securities Division 1511 Pontiac Avenue John O. Pastore Complex – Building 68-2 Cranston, RI 02920 (401)222-3048

South Dakota

Division of Insurance Securities Regulation 124 S Euclid, Second Floor Pierre, SD 57501 (605)773-3563

Texas

Office of the Secretary of State Statutory Document Section 1019 Brazos Street Austin, Texas 78701

<u>Utah</u>

State of Utah Department of Commerce Division of Consumer Protection 160 East Three Hundred South P.O. Box 146704 Salt Lake City, Utah 84114-6704

<u>Virginia</u>

State Corporation Commission Division of Securities and Retail Franchising 1300 East Main Street, 1st Floor Richmond, Virginia 23219 (804)371-9051

Washington

Department of Financial Institutions Securities Division PO Box 41200 Olympia, WA 98504-98501 (360)902-8760

Wisconsin

Division of Securities Department of Financial Institutions 201 W West Washington Avenue, Suite 300 Madison, Wisconsin 53703 (608)267-9140

EXHIBIT C

TO THE DISCLOSURE DOCUMENT THE GLASS GURU ENTERPRISES, INC.

LIST OF STATE AGENTS FOR SERVICE OF PROCESS

The following state agencies are designated as our agent for service of process in accordance with the applicable state laws. We may register in one or more of these states.

California

Department of Financial Protection and Innovation One Sansome St., Suite 600 San Francisco, California 94104 (415)972-8565

Department of Financial Protection and Innovation 320 W. 4th Street, Suite 750 Los Angeles, California 90013 (213)576-7500

Department of Financial Protection and Innovation 2101 Arena Blvd. Sacramento, California 95834 (866) 275-2677 Toll Free

Hawaii

Commissioner of Securities Department of Commerce & Consumer Affairs 335 Merchant Street, Room 203 Honolulu, Hawaii 96813 (808) 586-2722

<u>Illinois</u>

Illinois Attorney General 500 South Second Street Springfield, Illinois 62701 (217) 782-4465

<u>Indiana</u>

Indiana Secretary of State Securities Division 302 West Washington Street, Room E-111 Indianapolis, Indiana 46204 (317) 232-6681

Maryland

Office of the Attorney General Securities Division 200 St. Paul Place Baltimore, Maryland 21202 (410) 576-6360

Michigan

Department of the Attorney General Consumer Protection Division, Franchise Unit 525 West Ottawa Street F. Mennen Williams Building .1st Floor Lansing, Michigan 48913 (517)-373-7117

Minnesota

Minnesota Department of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101

New York

State of New York Office of the Attorney General 28 Liberty Street 21st Floor New York, NY 10005 (212) 416-8285

North Dakota

North Dakota Securities Department State Capitol – 5th Floor 600 East Boulevard Avenue Bismarck, North Dakota 58505-0510 (701) 328-4712

Rhode Island

Department of Business Regulation Securities Division 1511 Pontiac Avenue John O. Pastore Complex – Building 68-2 Cranston, RI 02920 (401) 222-3048

South Dakota

Division of Insurance Securities Regulation 124 S Euclid, Second Floor Pierre, SD 57501 (605)773-3563

Virginia

State Corporation Commission Division of Securities and Retail 1300 East Main Street 9th Floor Richmond, Virginia 23219 (804) 371-9051

Washington

Department of Financial Institutions Securities Division - 3rd Floor 150 Israel Road SW Tumwater Washington 98501 (360) 902-8760

Wisconsin

Division of Securities Department of Financial Institutions 4822 Madison Yards Way, North Tower Madison, Wisconsin 53705 (608) 267-9140

EXHIBIT D

TO THE DISCLOSURE DOCUMENT THE GLASS GURU ENTERPRISES, INC.

TABLE OF CONTENTS OF CONFIDENTIAL OPERATIONS MANUAL

The Glass Guru™ Franchise Operations Manual

Table of Contents

Welcome to The Glass Guru	7
How to Use This Manual	8
Updates to The Manual	8
Confidentiality	8
History of The Glass Guru	9
Advisory Council	9
The Glass Guru Mission Statement	9
The Glass Guru Core Values	10
The Glass Guru Guiding Principles	10
Service of Customers	10
Each Customer is the Most Important Customer	10
We Make It Easy for the Customer	10
Stand Behind the Work	11
The Glass Guru Support Resources	11
Guru Support	11
Directory	11
The Glass Guru Affiliations and Associations	11
National Glass Association	11
Glass Association of North America	11
International Franchise Association	11
Window & Door Dealers Alliance	12

Independently Owned and Operated Posting Requirements	12
Location	12
Customer Facing Materials	12
Only Franchisee Has the Right to Control Employees	12
Trademarks	13
Franchisor Ownership	13
The Glass Guru	13
General Trademark Use	13
Manner of Using Trademarks	14
Trademark Goodwill	14
Permitted Use of Trademark in Franchisee Business Name	15
Contributions and Donations Restrictions	15
Notification of Trademark Infringements and Claims	15
Indemnification for Use of Marks	15
Discontinuance of Trademark Use	16
Trade Secrets and other Confidential Information	16
Trade Secrets Usage and Corporate Identity	17
Designated Manager Training Requirements	19
The Glass Guru Franchise Training Process	19
Pre-Launch	19
Initial Franchise Training	20
Failure to Complete Initial Training Program	20
Additional Training & Refresher Courses	21
Designated Manager Mandatory Retraining	21
Training and Performance Improvement Requirements	22
Reporting Requirements	23

Financial Reporting Requirements	24
Reporting Overview	24
Related Documents	24
Royalty Reporting and Instructions on Completion	24
Penalties for Late Payment	24
Marketing Fee Reporting and Payment Requirements	25
Non-APR Fee	25
3rd Party Technology Fees*	26
Fee Calculation	27
APR Tracking	27
Fee Disputes	27
Workflow	28
Important Dates	29
Disputing Non-APR Fees	30
Submitting a Support Ticket	30
Accounting Requirements	32
Overview	32
Accounting Records and Reports Maintenance Requirements	32
Franchise Location Inspections	32
Right to Inspect	32
Books and Records	32
Contact with Other Related to Franchise Business	34
Insurance Requirements	34
General Insurance Coverage	34
You Must Have the Following Insurance Coverage:	34
Comprehensive General Liability Insurance	35

Workers Compensation Insurance	35
Automobile Liability Insurance	36
"Special Form" (all risk) Property Insurance Coverage	36
Business Interruption Insurance	36
Possible Increase the Minimum Liability Protection	36
Franchisor Right to Procure Insurance on Your Behalf	36
Risk Management	36
Reporting Incidents	36
Safety Recordkeeping	37
Marketing Requirements	38
Minimum Local Advertising Expenditure	38
Marketing Fee	39
Exclusive Use of The Glass Guru Brand	39
Special Marketing Approval Request	39
Telemarketing	39
Marketing, Solicitation and Service Restrictions	40
No Direct Marketing in Another Franchisee's Territory	40
Service Restrictions for Areas not Granted	40
No Right to Customers Outside of Your Designated Area of Primary Responsibility	41
Warranty & Financial Obligation for Areas Not Granted	41
Data Security Requirements, and Guidance	41
State and Federal Data Security Requirements	41
PCI Standards	42
PCI-DSS Requirements	42
The Twelve requirements of PCI-DSS:	42
Software Attack Notification Requirements	43

Social Media Usage Policy	44
Authorized Business Use of Social Media	45
Creating Social Media Accounts; Approval	45
Posting Content on Social Media	46
Personal Use of Social Media	48
Franchisor's Sole Right to Domain Name	50
Technology Requirements	50
Technology Introduction	50
Technology Usage	50
Technology Ecosystem	50
Intranet Guru HQ	51
Library	51
Learning (LMS) Guru University	51
Support Portal Guru Support	51
Third-Party Software Required for Business Operation (billed monthly)	52
Electronic Equipment Requirements	53
Station Computers Station Computers	53
Printer	53
Office/Telephone Lines	53
Cellular Phones	53
Tablets	54
Vehicle Requirements	54
Minimum Requirements	54
Vehicle Wrap Requirements	55
Uniform and Apparel Requirements	55
Office Personnel Attire	55

Estimator/ Sales Personnel Attire	55
Glazier / Helper Personnel Attire	55
Handling Customer Complaints and Problems	56
Escalation of Customer Complaints	56
Hours of Operation	56
Visitors in the Workplace	57
Employee Relations	57
A Word on Legal Compliance	57
Staffing Your Franchise	58
Designated Manager	58
Risk Management	59
Managing Risk at the Franchise Location or Job Site	60
New Employees Safety Training	60
Training Provides the Following Benefits:	60
Employee Training is Required to be Provided at the Following Times:	60
nitial Training Documents	61
Collections and Accounts Receivable Management	61
90 Day Notice	61
60-90-120 Days Overdue for Residential/Commercial	61
Expansion and Relocation Requirements	61
Franchise Territory, Resale Purchase or Territory Expansion	61
Convention Attendance Requirements	62
Conclusion	62

^{*}ALL THE GLASS GURU FRANCHISE TEMPLATES & DOCUMENTS ARE ACCESSIBLE THROUGH GURU HQ.

EXHIBIT E

TO THE DISCLOSURE DOCUMENT THE GLASS GURU ENTERPRISES, INC.

FINANCIAL STATEMENTS



INDEPENDENT AUDITORS' ACKNOWLEDGMENT

The Glass Guru Enterprises, Inc. Plano, Texas

We agree to the inclusion in the Franchise Disclosure Document dated April 30, 2024, issued by The Glass Guru Enterprises, Inc. (Franchisor) of our report dated April 4, 2024, relating to the financial statements of Franchisor as of and for the years ended December 31, 2023, 2022, and 2021.

CliftonLarsonAllen LLP

Clifton Larson Allen LLP

Roseville, California April 4, 2024

THE GLASS GURU ENTERPRISES, INC. FINANCIAL STATEMENTS YEARS ENDED DECEMBER 31, 2023, 2022 AND 2021



THE GLASS GURU ENTERPRISES, INC. TABLE OF CONTENTS YEARS ENDED DECEMBER 31, 2023, 2022 AND 2021

INDEPENDENT AUDITORS' REPORT		1
F	INANCIAL STATEMENTS	
	BALANCE SHEETS	4
	STATEMENTS OF INCOME	6
	STATEMENTS OF CHANGES IN STOCKHOLDERS' EQUITY	7
	STATEMENTS OF CASH FLOWS	8
	NOTES TO FINANCIAL STATEMENTS	10



INDEPENDENT AUDITORS' REPORT

Board of Directors The Glass Guru Enterprises, Inc. Plano, Texas

Report on the Audit of the Financial Statements Opinion

We have audited the accompanying financial statements of The Glass Guru Enterprises, Inc. (a Texas corporation), which comprise the balance sheets as of December 31, 2023, 2022 and 2021, and the related statements of income, changes in stockholders' equity, and cash flows for the years then ended, and the related notes to the financial statements.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of The Glass Guru Enterprises, Inc. as of December 31, 2023, 2022 and 2021, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the Auditors' Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of The Glass Guru Enterprises, Inc. and to meet our other ethical responsibilities in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Emphasis of Matter – Changes in Accounting Principles

As discussed in Note 1 to the financial statements, in 2023 the Company adopted new accounting guidance for the measurement of credit losses on financial instruments. Additionally, in 2022 the Company adopted new accounting guidance for leases. The guidance requires lessees to recognize a right-of-use asset and corresponding lease liability for all operating and finance leases with lease terms greater than one year. Our opinion is not modified with respect to this matter.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

Board of Directors
The Glass Guru Enterprises, Inc.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about The Glass Guru Enterprises, Inc.'s ability to continue as a going concern for one year after the date the financial statements are available to be issued.

Auditors' Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditors' report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due
 to fraud or error, and design and perform audit procedures responsive to those risks. Such
 procedures include examining, on a test basis, evidence regarding the amounts and disclosures
 in the financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit
 procedures that are appropriate in the circumstances, but not for the purpose of expressing an
 opinion on the effectiveness of The Glass Guru Enterprises, Inc.'s internal control. Accordingly,
 no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about The Glass Guru Enterprises, Inc.'s ability to continue as a going concern for a reasonable period of time.

Board of Directors The Glass Guru Enterprises, Inc.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control related matters that we identified during the audit.

CliftonLarsonAllen LLP

Clifton Larson Allen LLP

Roseville, California April 4, 2024

THE GLASS GURU ENTERPRISES, INC. BALANCE SHEETS DECEMBER 31, 2023, 2022 AND 2021

	2023	2022		2021
ASSETS				
CURRENT ASSETS				
Cash and Cash Equivalents	\$ 1,894,599	\$ 1,070,554	\$	726,876
Accounts Receivable, Net	361,725	359,089		317,445
Current Portion of Franchise Fee				
Financing Receivables	11,886	26,981		35,264
Prepaid Expenses	52,402	25,884		243,434
Total Current Assets	2,320,612	1,482,508		1,323,019
PROPERTY AND EQUIPMENT, NET	45,100	63,280		61,798
OTHER ASSETS				
Right-of-Use Assets - Operating	183,231	254,905		-
Franchise Fee Financing Receivables, Net of	,			
Current Portion	7,646	33,138		40,025
Loans to Stockholders	471,649	471,649		471,649
Deposits	8,284	8,284		14,784
Total Other Assets	670,810	767,976		526,458
Total Assets	\$ 3,036,522	\$ 2,313,764	_\$_	1,911,275

THE GLASS GURU ENTERPRISES, INC. BALANCE SHEETS (CONTINUED) DECEMBER 31, 2023, 2022 AND 2021

		2023	2022		2021
LIABILITIES AND STOCKHOLDERS' EQUITY					
CURRENT LIABILITIES					
Accounts Payable	\$	112,017	\$ 81,958	\$	135,171
Accrued Expenses	-	65,207	29,648	•	74,593
Income Taxes Payable		125,000	13,400		74,600
Current Portion of Lease Liability - Operating		72,847	70,758		- 1,000
Current Portion of Note Payable		10,356	10,051		9,754
Current Portion of Deferred Initial Franchise Fees		56,795	70,867		84,751
Total Current Liabilities		442,222	276,682		378,869
LONG-TERM LIABILITIES					
Lease Liability - Operating, Net of			ST .		
Current Portion		113,097	185,944		-
Note Payable, Net of Current Portion		32,661	43,016		53,067
Deferred Initial Franchise Fees, Net of					
Current Portion		60,794	96,250		166,097
Deferred Tax Liability		38,600	48,300		48,400
Total Long-Term Liabilities		245,152	373,510		267,564
Total Liabilities		687,374	650,192		646,433
STOCKHOLDERS' EQUITY					
Common Stock, \$0.0001 par value, 1,000					
shares authorized, 237.5 shares issued					
and outstanding		-	-		-
Additional Paid-in Capital		100,000	50,000		_
Retained Earnings		2,249,148	1,613,572		1,264,842
Total Stockholders' Equity		2,349,148	1,663,572		1,264,842
Total Liabilities and Stockholders' Equity	\$	3,036,522	\$ 2,313,764	_\$_	1,911,275

THE GLASS GURU ENTERPRISES, INC. STATEMENTS OF INCOME YEARS ENDED DECEMBER 31, 2023, 2022 AND 2021

	2023	2022	2021
REVENUE			
Franchise Royalty, Technology, and Marketing Fees Initial Franchise Fees Product Sales Revenue Other Revenue	\$ 4,680,611 132,382 38,154 22,650	\$ 4,359,076 201,645 43,812 73,383	\$ 3,774,437 267,017 101,892
Total Revenues	4,873,797	4,677,916	4,143,346
COST OF REVENUE	1,424,494	1,859,156	1,066,643
GROSS PROFIT	3,449,303	2,818,760	3,076,703
OPERATING EXPENSES	2,696,666	2,481,450	2,399,326
INCOME FROM OPERATIONS	752,637	337,310	677,377
OTHER INCOME (EXPENSE) Extinguishment of Debt Income - Paycheck Protection Program Loan	-	-	171,300
Interest Expense	(1,455)	(1,751)	(2,039)
Miscellaneous Income (Expense), Net	36,294	6,873	(2,184)
Total Other Income, Net	34,839	5,122	167,077
INCOME BEFORE INCOME TAXES	787,476	342,432	844,454
INCOME TAX EXPENSE (BENEFIT)	151,900	(6,298)	220,914
NET INCOME	\$ 635,576	\$ 348,730	\$ 623,540

THE GLASS GURU ENTERPRISES, INC. STATEMENTS OF CHANGES IN STOCKHOLDERS' EQUITY YEARS ENDED DECEMBER 31, 2023, 2022 AND 2021

	Сотто	Common Stock	Treasury	Additional	Retained	
	Shares	Amount	Stock	Paid-in Capital	Earnings	Total
BALANCE - JANUARY 1, 2021	238	€	. ←	. ↔	\$ 641,302	\$ 641,302
Net Income					623,540	623,540
BALANCE - JANUARY 1, 2022	238	r	1	ı	1,264,842	1,264,842
Receipt of Contributed Treasury Stock	(2)	,	(50,000)	20,000	í	
Sale of Treasury Stock	2	•	50,000	•	•	900'09
Net Income	11		31	1	348,730	348,730
BALANCE - DECEMBER 31, 2022	238	,	1	50,000	1,613,572	1,663,572
Receipt of Contributed Treasury Stock	(2)	•	(50,000)	900'09	ï	•
Sale of Treasury Stock	2	1	20,000	1.	ï	20,000
Net Income					635,576	635,576
BALANCE - DECEMBER 31, 2023	238	€	φ.	\$ 100,000	\$ 2,249,148	\$ 2,349,148

THE GLASS GURU ENTERPRISES, INC. STATEMENTS OF CASH FLOWS YEARS ENDED DECEMBER 31, 2023, 2022 AND 2021

		2023		2022	2021
CASH FLOWS FROM OPERATING ACTIVITIES Net Income Adjustments to Reconcile Net Income to Net	\$	635,576	\$	348,730	\$ 623,540
Cash Provided by Operating Activities:		40.400		47.000	4.000
Depreciation Expense		18,180		17,030	4,606
Financed Initial Franchise Fees		(23,760)		(23,760)	(9,090)
Loss on Disposal of Fixed Assets		2.000			13,053
Bad Debt Expense		3,920		23,798	7,918
Change in Deferred Tax Liability		(9,700)		(100)	27,100
Noncash Lease Expense		916		1,797	-
Extinguishment of Debt Income - Paycheck					(474.000)
Protection Program Loan		-		-	(171,300)
Changes in Operating Assets and Liabilities:		(0.550)		(05.440)	(00 500)
Accounts Receivable		(6,556)		(65,442)	(60,569)
Inventory		(00 F40)		- 047.550	12,298
Prepaid Expenses		(26,518)		217,550	(243,434)
Franchise Fee Financing Receivables		64,347		38,930	67,068
Deposits		20.050		6,500	(6,373)
Accounts Payable Accrued Expenses		30,059		(53,213)	109,725
Income Taxes Payable		35,559		(44,945)	70,196
Deferred Initial Franchise Fees		111,600		(61,200)	(89,541)
		(49,528)	-	(83,731)	 20,801
Net Cash Provided by Operating Activities		784,095		321,944	375,998
CASH FLOWS FROM INVESTING ACTIVITIES					
Loans to Stockholders		=		,	(110,825)
Cash Paid for Property and Equipment		**		(18,512)	(63,971)
Net Cash Used by Investing Activities	St.	-	-	(18,512)	(174,796)
CASH ELOWS EDOM EINANCING ACTIVITIES					
CASH FLOWS FROM FINANCING ACTIVITIES Proceeds from Sale of Treasury Stock		50,000		50,000	-
Principal Payments on Related Party Note Payable	7	(10,050)	8	(9,754)	(9,466)
Net Cash Provided (Used) by Financing Activities		39,950		40,246	(9,466)
NET INCREASE IN CASH AND CASH EQUIVALENTS		824,045		343,678	191,736
Cash and Cash Equivalents - Beginning of Year		1,070,554		726,876	535,140
CASH AND CASH EQUIVALENTS - END OF YEAR	\$	1,894,599	\$	1,070,554	\$ 726,876

THE GLASS GURU ENTERPRISES, INC. STATEMENTS OF CASH FLOWS (CONTINUED) YEARS ENDED DECEMBER 31, 2023, 2022 AND 2021

	 2023		2022		2021
SUPPLEMENTAL DISCLOSURES OF CASH FLOW INFORMATION				_	
Cash Paid for Interest	\$ 1,455	<u>\$</u>	1,751	\$	2,039
Cash Paid for Income Taxes	\$ 50,000	\$	55,000	\$	287,776

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

The following items comprise the significant accounting policies of The Glass Guru Enterprises, Inc. (the Company). The policies reflect industry practices and conform to accounting principles generally accepted in the United States of America (U.S. GAAP).

Principal Business Activity

The Glass Guru Enterprises, Inc., a Texas corporation, is a full-service glass and window restoration and replacement franchise business offering unique niche repair services as well as traditional glass shop services to residential customers. The Company offers franchises throughout the United States.

Variable Interest Entities

Accounting guidance issued by the Financial Accounting Standards Board (FASB) requires a variable interest entity (VIE) to be consolidated by a company if that company is considered to be the primary beneficiary of the VIE. Management has deemed certain related entities to be a VIE, with the Company being the primary beneficiary. Under Accounting Standards Update (ASU) No. 2018-17, Consolidation (Topic 810): Targeted Improvements to Related Party Guidance for Variable Interest Entities, a nonpublic entity has the option to exempt itself from applying the VIE consolidation model to qualifying common control entities. The Company applied the provisions of the ASU and has elected not to consolidate the related affiliates with the Company.

Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenue and expenses during the reporting period. Actual results could differ from those estimates.

Adoption of New Accounting Pronouncements

At the beginning of 2023, the Company adopted FASB ASU 2016-13, Financial Instruments – Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, as amended, which modifies the measurement of expected credit losses. The Company adopted this new guidance using the modified retrospective transition method, with no material impact on the Company's financial statements for the year ended December 31, 2023.

In February 2016, the FASB issued ASU No. 2016-02, *Leases (Topic 842)*. This new standard increases transparency and comparability among organizations by requiring the recognition of right-of-use (ROU) assets and lease liabilities on the balance sheet. Most prominent among the changes in the standard is the recognition of ROU assets and lease liabilities by lessees for those leases classified as operating leases. Under the standard, disclosures are required to meet the objective of enabling users of the financial statements to assess the amount, timing, and uncertainty of cash flows.

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Adoption of New Accounting Pronouncements (Continued)

The Company adopted the requirements of the guidance effective January 1, 2022, and has elected to apply the provisions of this standard to the beginning of the period of adoption, while continuing to present the comparative period in accordance with guidance under the lease standard effective during that period. The Company has elected to adopt the package of practical expedients available in the year of adoption.

Revenue and Cost Recognition

The Company recognizes revenue in accordance with FASB Accounting Standards Codification (ASC) 606, Revenue from Contracts with Customers (Topic 606) when its customer obtains control of promised goods or services in an amount that reflects the consideration to which the Company expects to receive in exchange for those goods or services. To determine revenue recognition for the arrangements that the Company determines are within the scope of Topic 606, the Company performs the following five steps: (1) identify the contract(s) with the customer, (2) identify the performance obligations in the contract, (3) determine the transaction price, (4) allocate the transaction price to the performance obligations in the contract, and (5) recognize revenue when (or as) the Company satisfies a performance obligation.

A contract exists when it has approval and commitment from both parties, the rights of the parties are identified, payment terms are identified, the contract has commercial substance, and collectability of consideration is probable. The Company enters into contracts with its franchisees to provide pre-opening training and support and ongoing use of the Company's branding and intellectual property, as well as marketing and technology support. These contracts are generally entered into for a period of five years. The Company charges an initial franchise fee at the inception of the contract, which is either collected at the inception of the contract or financed over the term of the contract (see Note 2), as well as continuing royalty, marketing, technology, and other fees for ongoing services provided by the Company based on a percentage of the franchisee's adjusted gross revenue. The Company also makes sales of promotional products and supplies to customers.

The Company considers initial franchise fees to be for initial franchise development activities and ongoing use of the Company's branding and intellectual property. Therefore, under the guidance in Topic 606, the Company recognizes revenue from initial franchise fees in the amount of costs incurred by the Company for pre-opening training and support, with the remainder of the initial franchise fee recognized evenly over the contract term as the Company fulfills the performance obligation to its customer to grant the right to use, and benefit from, the Company's branding and intellectual property.

The Company recognizes revenue from continuing royalty, marketing, technology, and other fees over time as performance obligations are satisfied, as the customer simultaneously receives and consumes the benefits provided by the Company's ongoing marketing and technology support as the Company performs. Revenue is recognized at contractually agreed upon monthly percentages of franchisee adjusted gross revenue in the month that services are provided. Management believes that this method is the most accurate measurement of progress to date for this performance obligation.

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Revenue and Cost Recognition (Continued)

The Company recognizes revenue from sales of promotional products and supplies when control of the products and supplies has transferred to customers. For the majority of the Company's sales of products and supplies, control transfers at a point in time when the items have been delivered as that is generally when legal title, physical possession, and the risks and rewards of the items are transferred to the customer. Amounts billed to customers related to shipping and handling costs are included in net sales in the statements of income. The Company has elected to account for shipping and handling costs as fulfillment costs which are included in cost of sales in the statements of income.

The Company has determined that the nature, amount, timing, and uncertainty of revenue and cash flows are not materially affected by customer type, line of business, geographic location, or other differentiating factor.

Opening and closing contract balances were as follows:

				nchise Fee	Į.	Deferred		
	Accounts		Accounts		Financing			Initial
	Rec	Receivable, Net		Receivables		Franchise Fees		
January 1, 2021	\$	257,151	\$	140,910	\$	230,047		
December 31, 2021		317,445		75,289		250,848		
December 31, 2022		359,089		60,119		167,117		
December 31, 2023		361,725		19,532		117,589		

Cash and Cash Equivalents

The Company considers all highly liquid investments purchased with a maturity of three months or less to be cash equivalents.

Accounts Receivable and Allowance for Credit Losses

The Company has adopted the current expected credit losses (CECL) methodology for estimating credit losses on financial assets, effective January 1, 2023, utilizing the modified retrospective transition method. The adoption of CECL resulted in changes to the Company's accounting policies, including recognition of credit losses based on expected future credit losses rather than incurred credit losses. The Company also updated its accounting policies for determining the recoverability of trade receivables, loans, and other financial assets. The adoption of this standard did not have a material impact on the Company's financial statements, but did change how the allowance for credit losses is determined.

The Company uses a combination of historical loss experience, current economic conditions, and forward-looking information to estimate credit losses for financial assets. The Company considers various factors, such as borrower creditworthiness and collateral values to estimate credit losses.

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Accounts Receivable and Allowance for Credit Losses (Continued)

Accounts receivable are stated at their estimated collectible amounts and comprise amounts billed and currently due from customers. The Company extends credit to customers in the normal course of business. The Company establishes an allowance for credit losses to present the net amount of accounts receivable expected to be collected. The allowance represents the estimate of expected credit losses based on historical experience, current economic conditions, and certain forward-looking information. At December 31, 2023, 2022 and 2021, management established an allowance for credit losses of \$2,813, \$13,348, and \$-0-, respectively.

Property and Equipment

Property and equipment, carried at cost, is depreciated over the estimated useful life of the related asset. Costs of repairs and maintenance are charged to expense. Upon retirement or disposal of property and equipment, the costs and related depreciation are removed from the accounts, and gains or losses, if any, are reflected in earnings for financial and income tax reporting purposes. Depreciation is provided for under the straight-line method for financial reporting and accelerated methods for income tax reporting. The estimated useful lives used for calculating depreciation for property and equipment are as follows:

Computer and Technology Equipment 5 Years Furniture 5 Years

Leases

The Company determines if an arrangement is a lease at inception. Operating leases are included in Right-of-Use (ROU) Assets – Operating and Lease Liability – Operating on the balance sheets. Finance leases, if any, are included in Right-of-Use Assets – Finance and Lease Liability – Finance on the balance sheets.

ROU assets represent the Company's right to use an underlying asset for the lease term and lease liabilities represent the Company's obligation to make lease payments arising from the lease. ROU assets and liabilities are recognized at commencement date based on the present value of lease payments over the lease term.

Lease terms may include options to extend or terminate the lease when it is reasonably certain that the Company will exercise that option. Lease expense for operating lease payments is recognized on a straight-line basis over the lease term. The Company has elected to recognize payments for short-term leases with a lease term of 12 months or less as an expense when incurred and these leases are not included as right-of-use assets or lease liabilities on the balance sheets.

The individual lease contracts do not provide information about the discount rate implicit in the lease. Therefore, the Company has elected to use a risk-free discount rate determined using a period comparable with that of the lease term for computing the present value of lease liabilities.

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Leases (Continued)

The Company has elected not to separate nonlease components from lease components and instead accounts for each separate lease component and the nonlease component as a single lease component.

Impairment of Long-Lived Assets

The Company reviews long-lived assets for impairment whenever events or changes in circumstances indicate that the carrying amount of such assets may not be recoverable. Recoverability of these assets is determined by comparing the forecasted undiscounted net cash flows of the operation to which the assets relate to the carrying amount. If the operation is determined to be unable to recover the carrying amount of its assets, then assets are written down first, followed by other long-lived assets of the operation to fair value. Fair value is determined based on discounted cash flows or appraised values, depending on the nature of the assets. There were no impairment losses recognized for long-lived assets during the years ended December 31, 2023, 2022 and 2021.

Income Taxes

The Company pays taxes based on its income for federal and state purposes. Some states in which the Company pays income taxes have minimum franchise taxes, to which the Company is subject.

Income taxes are provided for the tax effects of transactions reported in the financial statements and consist of taxes currently due plus deferred taxes related primarily to differences in reporting for accrual and income tax purposes, and depreciation. Deferred taxes represent the future tax return consequences of those differences, which will either be taxable or deductible when the assets are recovered and liabilities settled. The Company has not recorded a valuation allowance against deferred tax assets, as management believes that the deferred tax assets will be realized at a point in the future.

Uncertain Tax Positions

Accounting guidance issued by the Financial Accounting Standards Board (FASB) prescribes a recognition threshold and measurement attribute for financial statement recognition and measurement of a tax position taken or expected to be taken in a tax return. For those benefits to be recognized, a tax position must be more-likely-than-not to be sustained upon examination by taxing authorities. The Company did not have unrecognized tax benefits as of December 31, 2023, 2022 and 2021 and does not expect this to change significantly over the next twelve months. The Company will recognize interest and penalties accrued on any unrecognized tax benefits as a component of income tax expense. As of December 31, 2023, 2022 and 2021 the Company has not accrued interest or penalties related to uncertain tax positions.

Advertising Costs

Advertising costs are expensed as incurred. Advertising expense was \$244,830, \$181,805 and \$128,977 for the years ended December 31, 2023, 2022 and 2021, respectively and is included in operating expenses on the statements of income.

NOTE 1 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Concentrations of Credit Risk

Credit Risk

Financial instruments which potentially subject the Company to concentrations of credit risk consist primarily of cash and cash equivalents.

Cash and Cash Equivalents

The Company maintains its cash in bank deposit accounts which, at times, may exceed federally insured limits. The Company has not experienced any losses in such accounts. The Company believes it is not exposed to any significant credit risk on cash and cash equivalents.

Subsequent Events

In preparing these financial statements, the Company has evaluated events and transactions for potential recognition or disclosure through April 4, 2024, the date the financial statements were available to be issued.

NOTE 2 FRANCHISE FEE FINANCING RECEIVABLES

Franchise fee financing receivables consist of initial franchisee fees that the Company and franchisee enter into a financing agreement for payments of principal and interest to be made over a contractually agreed upon period of time at interest rates between 6.00% and 7.00%.

An allowance for uncollectible franchise fee financing receivables is maintained at a level management believes is adequate to provide for potential losses on financing arrangements. Management's periodic evaluation of the adequacy of the allowance is based on a specific identification of receivables considered uncollectible. The evaluation is inherently subjective as it requires estimates that are susceptible to significant revision as more information becomes available. The allowance for uncollectible franchise fee financing receivables was \$-0-, \$-0- and \$7,643 for the years ended December 31, 2023, 2022 and 2021, respectively.

Long-term maturities of franchise fee financing receivables are as follows:

Year Ending December 31,	Amount		
2024	\$	11,886	
2025		3,828	
2026		3,818	
Total	\$	19,532	

NOTE 3 PROPERTY AND EQUIPMENT

Property and equipment consists of the following at December 31:

		2023	2022	2021
Computer and Technology Equipment	\$	49,798	\$ 49,798	\$ 36,177
Furniture		35,384	 35,384	30,493
Total	,	85,182	 85,182	66,670
Less: Accumulated Depreciation		40,082	21,902	4,872
Property and Equipment, Net	\$	45,100	\$ 63,280	\$ 61,798

Depreciation expense of property and equipment was \$18,180, \$17,030 and \$4,606 for the years ended December 31, 2023, 2022 and 2021, respectively and is included in operating expenses on the statements of income.

NOTE 4 LEASES - ASC 842

The Company leases office space under a long-term, noncancelable lease agreement that expires in June 2026. In the normal course of business, it is expected that this lease will be renewed or replaced by a similar lease.

The following table provides quantitative information concerning the Company's lease:

		2023		2022	
Lease Cost:					
Operating Lease Cost	\$	74,447	\$	74,446	
Other Information:					
Cash Paid for Amounts Included in the					
Measurement of Lease Liabilities					
Operating Cash Flows from Operating Leases	\$	73,531	\$	72,649	
Right-of-Use Assets Obtained in Exchange for					
New Operating Lease Liabilities	\$	-	\$	325,691	
Weighted-Average Remaining Lease		2.4 years		3.4 years	
Term - Operating Leases					
Weighted-Average Discount Rate - Operating Leases		1.27%		1.27%	

NOTE 4 LEASES - ASC 842 (CONTINUED)

A maturity analysis of annual undiscounted cash flows for lease liabilities is as follows:

Year Ending December 31,	
2024	\$ 74,708
2025	75,884
2026	38,236
Total Obligations Under Lease Liability - Operating	188,828
Less: Amount Representing Interest	(2,884)
Present Value of Obligations Under	
Lease Liability - Operating	\$ 185,944

NOTE 5 LEASES - ASC 840

The Company elected to apply the provisions of FASB ASC 842 to the beginning of the period of adoption, through a cumulative effect adjustment, with certain practical expedients available. Lease disclosures for the year ended December 31, 2021 are made under prior lease guidance in FASB ASC 840.

The Company leases office space under an operating lease, expiring in June 2026. For the year ended December 31, 2021, rent expense for lease office space was \$70,186, and is included in operating expenses on the statement of income.

The minimum future lease payments required under the lease are as follows:

Year Ending December 31,	 Amount		
2022	\$ 72,649		
2023	73,531		
2024	74,708		
2025	75,884		
2026	 38,236		
Total	\$ 335,008		

NOTE 6 LOANS TO STOCKHOLDERS

The Company has a receivable from stockholders with a balance of \$471,649 as of December 31, 2023, 2022 and 2021, which is included on the balance sheets as Loans to Stockholders.

The loans are due on demand and are not expected to be repaid within the next 12 months. The loans bear interest at rates between 0.14% and 1.73% and require yearly interest-only payments of \$5,096. The Company recorded interest income related to the loans as a component of miscellaneous income on the statements of income of \$5,095, \$5,096 and \$3,178 for the years ended December 31, 2023, 2022 and 2021, respectively.

NOTE 7 LINE OF CREDIT

On May 1, 2021, the Company entered into a credit agreement with a commercial bank for a revolving line of credit which permits borrowings up to \$60,000. The rate of interest charged on borrowings on the line is 9.50%. The agreement is unsecured and subject to periodic credit review and renewal. There was no outstanding balance on the line of credit as of December 31, 2023, 2022 and 2021.

NOTE 8 NOTE PAYABLE

In 2017, the Company secured a promissory note from a previous owner in the amount of \$100,000 to fund the repurchase of stock. The note is secured by all assets of the Company including, but not limited to, all franchisor royalty revenue. The note bears interest at the rate of 3.0% per annum, maturing in December 2027. The agreement stipulates monthly payments of principal and interest of \$959.

Long-term maturities of the note payable are as follows:

Year Ending December 31,	A	Amount		
2024	\$	10,356		
2025		10,671		
2026		10,996		
2027		10,994		
Total	\$	43,017		

NOTE 9 DEFERRED INITIAL FRANCHISE FEES

The Company recognizes the portion of deferred initial franchise fees attributable to the ongoing use of the Company's branding and intellectual property evenly over the term of its franchising agreements, which are typically five years.

Revenue associated with deferred initial franchise fees will be recognized as follows:

Year Ending December 31,	 Amount	
2024	\$ 56,795	
2025	37,102	
2026	15,601	
2027	5,757	
2028	 2,334	
Total	\$ 117,589	

NOTE 10 INCOME TAXES

The provision for income taxes is reconciled to the federal statutory rate as follows as of December 31:

	2023		2022		2021	
Federal Tax at Statutory Rate	\$	142,302	\$	(2,816)	\$	227,832
State Income Taxes, Net of						
Federal Benefit		_		-		17,992
Temporary and Permanent						
Differences, Net		9,598	1	(3,482)		(24,910)
Total Provision for Income Taxes	\$	151,900	\$	(6,298)	\$	220,914

The provision for income taxes for the year ended December 31, 2023 consists of the following:

	Current		Deferred		Total	
Federal Tax	\$ 142,100	\$	9,800	\$	151,900	

The provision for income taxes for the year ended December 31, 2022 consists of the following:

	Current		Deferred		Total	
Federal Tax	\$	(6,398)	\$	100	\$	(6,298)

The provision for income taxes for the year ended December 31, 2021 consists of the following:

	 Current		Deferred		Total	
Federal Tax	\$ 225,240	\$	(27,100)	\$	198,140	
State Tax	22,774		-		22,774	
Total	\$ 248,014	\$	(27,100)	\$	220,914	

The tax effects of temporary differences that give rise to deferred tax assets and liabilities consists of the following at December 31:

	2023		2022		2021	
Excess of Accumulated Tax	•					
Depreciation Over Accumulated						
Book Depreciation	\$	9,500	\$	13,300	\$	13,000
Cash-to-Accrual Adjustments		29,100		35,000		40,200
State Income Taxes Paid			-			(4,800)
Net Deferred Tax Liabilities	\$	38,600	\$	48,300	\$	48,400

THE GLASS GURU ENTERPRISES, INC. NOTES TO FINANCIAL STATEMENTS DECEMBER 31, 2023, 2022 AND 2021

NOTE 11 RELATED PARTY TRANSACTIONS

The Company recognized revenue from a franchisee under common control in the amount of \$176,492, \$133,670 and \$14,585 for the years ended December 31, 2023, 2022 and 2021, respectively. The Company had accounts receivable from this franchisee of \$1,391, \$1,695 and \$-0- as of December 31, 2023, 2022 and 2021, respectively.

NOTE 12 RETIREMENT PLAN

Employees of the Company are covered under a defined contribution plan as defined under Section 401(k) of the U.S. Internal Revenue Code. The safe harbor employer contribution under this plan is to match 100% of employees' contributions up to a maximum match of 4% of the employees' gross wages. The total retirement plan contributions were \$34,333, \$40,940 and \$28,733 for the years ended December 31, 2023, 2022 and 2021, respectively, and is included in operating expenses on the statements of income.

NOTE 13 PAYCHECK PROTECTION PROGRAM LOAN

On June 24, 2020, the Company received a loan from a commercial bank in the amount of \$171,300 to fund payroll, rent, utilities, and interest on mortgages and existing debt through the Paycheck Protection Program (the PPP Loan). The PPP Loan bears interest at a fixed rate of 1.00% per annum, has a term of two years, and is unsecured and guaranteed by the U.S. Small Business Administration (SBA). Payment of principal and interest is deferred until the date on which the amount of forgiveness is remitted to the lender or, if the Company fails to apply for forgiveness within ten months after the covered period, then payment of principal and interest shall begin on that date. These amounts may be forgiven subject to compliance and approval based on the timing and use of these funds in accordance with the program. The covered period from June 24, 2020 to February 19, 2021 is the time that a business has to spend its PPP Loan funds.

During 2021, the SBA processed the Company's PPP Loan forgiveness application and notified the commercial bank that the PPP Loan qualified for full forgiveness. Loan proceeds were received by the bank from the SBA, legally releasing the Company from the debt. The Company recognized extinguishment of debt income in the amount of the loan proceeds as a component of other income and expense on the combined statement of income.

The SBA may review funding eligibility and usage of funds in compliance with the program based on dollar thresholds and other factors. The amount of liability, if any, from potential noncompliance cannot be determined with certainty; however, management is of the opinion that any review will not have a material adverse impact on the Company's financial position.

THE GLASS GURU ENTERPRISES, INC. NOTES TO FINANCIAL STATEMENTS DECEMBER 31, 2023, 2022 AND 2021

NOTE 14 FRANCHISES IN OPERATION

The Company's franchise activity was as follows for December 31:

	2023	2022	2021
Franchises operating - beginning of year	78	77	71
Franchises opened during the year	2	7	11
Franchises closed during the year	7	6	5
Franchises operating - end of year	73	78	77

NOTE 15 COMMITMENTS AND CONTINGENCIES

The Company is occasionally involved in litigation matters that arise in the ordinary course of business. There are no pending significant legal proceedings to which the Company is a party for which management believes the ultimate outcome would have a material adverse effect on the Company's financial position.



EXHIBIT F

TO THE DISCLOSURE DOCUMENT THE GLASS GURU ENTERPRISES, INC.

LIST OF CURRENT AND FORMER FRANCHISEES

ACTIVE FRANCHISEES

During the year ended December 31, 2023 - All Units Are Open for Business

During the year ended December 31, 2023 - All Units Are Open for Business				
Franchise	Franchisee(s)	Phone	City	State
ALABAMA				
East Birmingham	Edward Green	205-433-7058	Moody	AL
Mobile	Wyatt & Sherry Gerald	251-287-1562	Mobile	AL
CALIFORNIA				
Carlsbad	Sherilynn de Maio, Laurel Mayo	760-994-4441	Carlsbad	CA
Brentwood	Ranae Williams	877-654-8507	Antioch	CA
Elk Grove	Steven Snead, Jeremy Goff, Breck Spain, Derek Klinck	916-714-4405	Elk Grove	CA
Fairfield-Vacaville	Chuck & Alex Turner	707-348-5884	Fairfield	CA
Folsom	Vito Lessa	916-517-0952	Orangevale	CA
Redlands	Rod Heggen	909-674-0029	Redlands	CA
Roseville*	Dan Frey, Dave Hull	916-786-4878	Roseville	CA
Stockton	Dennis Freeman	209-625-8156	Lodi	CA
Temecula	Jay Lund	951-719-1000	Temecula	CA
Torrance	Corey Gavin	888-466-7533	Torrance	CA
Yuba City	Greg Lewelling	530-674-3858	Yuba City	CA

COLORADO

Colorado Springs	James Child, Jr.	719-229-2419	Colorado Springs	СО
Fort Collins	Michael & Laurie Bean	970-713-0567	Loveland	СО
Grand Junction	Joe Miller	678-648-0005	Grand Junction	СО
Denver West	James Child, Jr.	720-379-8318	Arvada	СО
Denver South	James Child, Jr.	720-800-9500	Highlands Ranch	СО

FLORIDA

Boca Raton	Oriya Klein	561-218-4581	Boca Raton	FL
Port St. Lucie	Raymond Minter and Alexis Hernandez	772-773-0443	Port St. Lucie	FL
Sarasota	Martin Rivenbark	941-377-4878	Sarasota	FL
SE Orlando	Marco Calvert Barba	407-730-4549	Orlando	FL

GEORGIA

Johns Creek	Charles Gilbreath	770-727-0955	Alpharetta	GA
Warner-Robbins	Kenny Meredith	478-328-0999	Warner Robins	GA

HAWAII

Leeward Oahu	Guerric de Coligny	808-234-4260	Honolulu	НІ
ILLINOIS				
Gurnee	Tom Rowland	847-665-8450	Waukegan	IL

INDIANA

Indy North	Mitch Ritchey	317-222-1592	Fishers	IN			
Indy South	Eric & Daina Buehling	317-350-4138	Greenwood	IN			
KANSAS	KANSAS						
Olathe	Curtis & Brandy Dockett	913-353-2996	Olathe	KS			
Wichita	Michael & Valarie Gomm	316-440-4944	Wichita	KS			
LOUISIANA							
Baton Rouge	Ben Smith	225-271-4890	Denham Springs	LA			
MICHIGAN							
Grand Rapids	Eugene & Cynthia Williams	616-622-3500	Kentwood	МІ			
Traverse City	Steve Bentley	231-510-4464	Traverse City	МІ			
MINNESOTA							
Duluth	Dan Cook	218-624-2119	Duluth	MN			
Blaine	Jonathan Land	763-767-2151	Forest Lake	MN			
West Minneapolis	Peter & Danielle Batman	952-456-6278	Minnetonka	MN			
MISSOURI							
Springfield	Chris Allman	417-708-1231	Springfield	МО			
O'Fallon	Gregg Merkel	636-980-1616	Wentzville	МО			
NEW JERSEY		,	'				
Meadowlands	Jacek & Daria Piczak	201-528-9764	East Rutherford	ИЛ			
L		1	1				

N I	ر ا	A	•	`	\sim	п	1/
Ν	Ė١	/\	•	Y	u	к	ĸ

Albany Paul Schlimgen	518-567-7328 Rensselaer	NY
-----------------------	-------------------------	----

OHIO

East Columbus	David Garrison, Howard Garrison	614-856-4550	Columbus	ОН
West Columbus	David Garrison, Howard Garrison	614-522-0028	Columbus	ОН
Newark	David Garrison	614-522-0028	Columbus	ОН
Macedonia	David Garrison, Howard Garrison, Scott Grooms	330-467-5318	Macedonia	ОН
Mason	John & Lisa Chambers	513-204-2354	Mason	ОН
South Cincinnati	Chris Samples	859-279-3009	Wilder	KY

OKLAHOMA

Edmond	Greg Adams, Austin Adams	405-888-3301	Edmond	ОК
--------	--------------------------	--------------	--------	----

OREGON

Corvallis-Albany	Quentin Winegar	541-205-9752	Albany	OR	

PENNSYLVANIA

Levittown	Yogesh Patel	267-520-7405	Bensalem	PA
-----------	--------------	--------------	----------	----

SOUTH CAROLINA

Charleston	Erik Hilger	843-640-3270	Charleston	SC
Columbia	Daniel Jordon	803-638-4829	Columbia	SC

TENNESEE

West Knoxville	Joe & Jessica Grant	865-217-1013	Madisonville	TN
Chattanooga	Scott Tatum	423-541-3650	Chattanooga	TN
Hendersonville	A. Brant Lanham	615-647-8989	Hendersonville	TN
Memphis	Chris Chambers, Jonathan Conner	901-320-7100	Memphis	TN

TEXAS

Atascocita	Gustavo, Ana, Gustavo Jr. Valbuena	(281) 623-5604	Humble	TX
Austin	Michael Kugler	512-828-6770	Austin	TX
College Station	Craig & Rachael Bell	(979) 485-1443	College Station	TX
West Ft Worth	Blake Boren	817-666-3063	Azle	TX
Katy	Carlos Quintana	346-280-1502	Katy	TX
Longview	Mike Kirkindoll	(903) 686-0333	Longview	TX
Lubbock	Jimmy & Wendy Teeter	806-759-3419	Lubbock	TX
Frisco	Ted & Bonnie Halladay	469-252-3785	Prosper	TX
Plano	Mike & Michele Paulda	469-929-6043	Plano	TX
San Antonio	Michael Kugler	201-686-6151	San Antonio	TX
Sherman	Ted & Bonnie Halladay	903-487-2480	Sherman	TX
Southlake	Kyle & Stephanie Hansen	817-424-3500	Southlake	TX
Spring	Manuel & Isabel Landa	346-224-8380	Spring	TX
Tyler	Gary Kirkindoll	903-504-5646	Tyler	TX
·	·			

WASHINGTON

Bellingham	Su & TJ Randhawa	360-927-9395	Lynden	WA
Everett	Don Dawson	425-470-6728	Everett	WA
Redmond	Don Dawson	425-867-5267	Redmond	WA

WISCONSIN

Sheboygan	Brian Van Wageningen	920-894-4878	Sheboygan	WI	
-----------	----------------------	--------------	-----------	----	--

CANADA

SW Calgary	Frank Lyall	403-975-4527	Calgary	AB
West Edmonton	Kevin Truong	587-735-6969	Edmonton	AB
Ottawa East	Marc Latendresse	613-424-8996	Orleans	ON
North Surrey	Chris Wijaya	604-595-3011	Surrey	ВС
Chatham-Kent	Scott & Jen Bowen	519-627-9119	Wallaceburg	ON

ranchise	Franchisee(s)	Phone	Address	City	State	Postal	
----------	---------------	-------	---------	------	-------	--------	--

ALABAMA

East Birmingham	Edward Green	205-433- 7058	1000 Plantation Pkwy Suite 500	Moody	AL	35004
Mobile	Wyatt & Sherry Gerald	251-287- 1562	220 Portside Blvd Suite A	Mobile	AL	36695

CALIFORNIA

Carlsbad	Sherilynn de Maio, Laurel Mayo	760-994- 4441	2075 Corte DelNogal Suite V	Carlsbad	CA	92011
Brentwood	Ranae Williams	877-654- 8507	2220 A Street, Suite A	Antioch	CA	94509

Elk Grove	Steven Snead, Jeremy Goff, Breck Spain, Derek Klinck	916-714- 4405	9100 Elkmont Way	Elk Grove	CA	95624
Fairfield- Vacaville	Chuck & Alex Tuner	707-348- 5884	1601 Enterprises Drive Unit 14	Fairfield	CA	94533
Folsom	Vito Lessa	916-517- 0952	6243 Main Ave Suite A	Orangevale	CA	95662
Redlands	Rod Heggen	909-674- 0029	370 Alabama Street Suite M	Redlands	CA	92373
Roseville*	Dan Frey, Dave Hull	916-786- 4878	198 Cirby Way, Suite 120	Roseville	CA	95678
Stockton	Dennis Freeman	209-625- 8156	1024 Industrial Way	Lodi	CA	95240
Temecula	Jay Lund	951-719- 1000	43391 Business Park Drive Suite C-8	Temecula	CA	92591
Torrance	Corey Gavin	888-466- 7533	20950 Normandie Ave Unit G	Torrance	CA	90502
Yuba City	Greg Lewelling	530-674- 3858	909 B. North George Washington Blvd.	Yuba City	CA	95993

COLORADO

Colorado Springs	James Child, Jr.	719-229- 2419	204 Mount View Lane #8	Colorado Springs	СО	80907
Fort Collins	Michael & Laurie Bean	970-713- 0567	6712 N. Franklin Avenue	Loveland	СО	80538
Grand Junction	Joe Miller	678-648- 0005	784 Valley Court Unit E	Grand Junction	СО	81505
Denver West	James Child, Jr.	720-379- 8318	6618 Wadsworth Blvd.	Arvada	СО	80003
Denver South	James Child, Jr.	720-800- 9500	9457 S. University #178	Highlands Ranch	СО	80126

FLORIDA

Boca Raton	Oriya Klein	561-218- 4581	10050 Spanish Isles Blvd E2	Boca Raton	FL	33498
Port St. Lucie	Raymond Minter and Alexis Hernandez	772-773- 0443	1766 SW Biltmore Street	Port St. Lucie	FL	34984
Sarasota	Martin Rivenbark	941-377- 4878	4496 McAshton Street	Sarasota	FL	34233

SE Orlando	Marco Calvert Barba	407-730- 4549	6663 Narcoossee Road Suite 130	Orlando	FL	32822				
GEORGIA										
Johns Creek	Charles Gilbreath	770-727- 0955	1360 Union Hill Road Bld 6 Suite I	Alpharetta	GA	30004				
Warner- Robbins	Kenny Meredith	478-328- 0999	100 Hospital Drive	Warner Robins	GA	31088				
HAWAII										
Leeward Oahu	Guerric de Coligny	808-234- 4260	2412 Rose St, Suite 2	Honolulu	НІ	96819				
ILLINOIS										
Gurnee	Tom Rowland	847-665- 8450	1020 S Northpoint Blvd	Waukegan	IL	60085				
INDIANA										
Indy North	Mitch Ritchey	317-222- 1592	10089 Allisonville Road, Suite C	Fishers	IN	46038				
Indy South	Eric & Daina Buehling	317-350- 4138	545 Christy Drive, Suite 2200	Greenwood	IN	46143				
KANSAS										
Olathe	Brandy Dockett	913-353- 2996	715 N Lindenwood Drive Unit E	Olathe	KS	66062				
Wichita	Michael & Valarie Gomm	316-440- 4944	8847 W. Monroe Circle, Suite 200	Wichita	KS	67209				
LOUISIANA										
Baton Rouge	Ben Smith	225-271- 4890	1814 S Range Ave Suite B-	Denham Springs	LA	70726				
MICHIGAN										
Grand Rapids	Eugene & Cynthia Williams	616-622- 3500	4245 44th St SE Suite 9	Kentwood	МІ	49512				
Traverse City	Steve Bentley	231-510- 4464	807A Lake Ave	Traverse City	МІ	49684				
-		•								

MINNESOTA

Duluth	Dan Cook		102 East Central Entrance Suite 6	Duluth	MN	55811
Blaine	Jonathan Land	763-767- 2151	24078 Greenway Road # 11	Forest Lake	MN	55025
West Minneapolis	Peter & Danielle Batman	952-456- 6278	4 Shady Oak Road	Minnetonka	MN	55343

MISSOURI

Springfield	Chris Allman	417-708- 1231	3101 South Scenic Ave. Suite D	Springfield	МО	65807
O'Fallon	Gregg Merkel	636-980- 1616	2182 E Pitman Ave	Wentzville	МО	63385

NEW JERSEY

Meadowlands	Jacek & Daria Piczak	201-528- 9764	130 Hackensack Street	East Rutherford	NJ	07073	
-------------	----------------------	------------------	-----------------------	-----------------	----	-------	--

NEW YORK

Albany	Paul Schlimgen	518-567- 7328	1 Hillview Ave	Rensselaer	NY	12144	
--------	----------------	------------------	----------------	------------	----	-------	--

OHIO

East Columbus	David Garrison, Howard Garrison	614-856- 4550	1010-B Taylor Station Rd	Columbus	ОН	43230
West Columbus	David Garrison, Howard Garrison	614-522- 0028	1010-B Taylor Station Rd	Columbus	ОН	43230
Newark	David Garrison	614-522- 0028	1010-B Taylor Station Rd	Columbus	ОН	43230
Macedonia	David Garrison, Howard Garrison, Scott Grooms	330-467- 5318	7791 Capital Blvd Suite 8B	Macedonia	ОН	44056
Mason	John & Lisa Chambers	513-204- 2354	4457 Bethany Road	Mason	ОН	45040
South Cincinnati	Chris Samples	859-279- 3009	100 Beacon Drive Unit #4	Wilder	KY	41076

OKLAHOMA

Edmond	Greg Adams, Austin Adams	405-888- 3301	1321 Fretz Drive, Suite 140	Edmond	ОК	73013				
OREGON										
Corvallis-Albany	Quentin Winegar	541-205- 9752	1143 Anderson Place SE, Unit 7	Albany	OR	97332				
PENNSYLVANIA										
Levittown	Yogesh Patel	267-520- 7405	3580 Progress Drive, Unit A-1	Bensalem	PA	19020				
SOUTH CAROLINA										
Charleston	Erik Hilger	843-640- 3270	1038 Jenkins Road Ste. 102	Charleston	SC	29407				
Columbia	Daniel Jordon	803-638- 4829	1445 Shop Road Suite B	Columbia	SC	29201				
TENNESEE										
West Knoxville	Joe & Jessica Grant	865-217- 1013	268 Boring Street	Madisonville	TN	37354				
Chattanooga	Scott Tatum	423-541- 3650	Air Office Park, 117 Nowlin Lane, Suite 900	Chattanooga	TN	37421				
Hendersonville	Brant Lanham	615-647- 8989	40 Industiral Park Drive Suite E	Hendersonville	TN	37075				
Memphis	Chris Chambers, Jonathan Conner	901-320- 7100	1690 Shelby Oaks Drive N Suite 4	Memphis	TN	38134				
TEXAS										
Atascocita	Gustavo, Ana, Gustavo Jr. Valbuena	(281) 623- 5604	19737 Lee Road Bldg. A, Ste. 103	Humble	TX	77338				
Austin	Michael Kugler	512-828- 6770	8711 Burnet Rd, Suite B- 34	Austin	TX	78757				
College Station	Craig & Rachael Bell	(979) 485- 1443	3900 State Hwy 6 S Suite 112	College Station	TX	77845				
West Ft Worth	Blake Boren	817-666- 3063	7301 Veal Station Road	Azle	TX	76020				
Katy	Carlos Quintana	346-280- 1502	18502 Clay Road Bldg 2	Katy	TX	77084				

Longview	Mike Kirkindoll	(903) 686- 0333	122 South Ward Street	Longview	TX	75604
Lubbock	Jimmy & Wendy Teeter	806-759- 3419	13405 County Road 1600 Suite 2	Lubbock	TX	79382
Frisco	Ted & Bonnie Halladay	469-252- 3785	450 Business Park Drive Suite 105	Prosper	TX	75041
Plano	Mike & Michelle Paulda	469-929- 6043	849 J Place Suite F	Plano	TX	75074
San Antonio	Michael Kugler	201-686- 6151	7436 Reindeer Trail	San Antonio	TX	78238
Sherman	Ted & Bonnie Halladay	903-487- 2480	158 Meyers Drive	Sherman	TX	75092
Southlake	Kyle & Stephanie Hansen	817-424- 3500	575 Commerce	Southlake	TX	76092
Spring	Manuel & Isabel Landa	346-224- 8380	22131 Rothwood Road 1A1	Spring	TX	77389
Tyler	Gary Kirkindoll	903-504- 5646	6368 Elkton Way	Tyler	TX	75703

WASHINGTON

Bellingham	Su & TJ Randhawa	360-927- 9395	7344 Guide Meridan Road	Lynden	WA	98264
Everett	Don Dawson	425-470- 6728	2207 Everett Ave Suite A	Everett	WA	98201
Redmond	Don Dawson	425-867- 5267	14572 NE 95th St	Redmond	WA	98052

WISCONSIN

Sheboygan Brian Van Wageningen	920-894- 4878	3325 Behrens Parkway	Sheboygan	WI	53081	
--------------------------------	------------------	----------------------	-----------	----	-------	--

CANADA

SW Calgary	Frank Lyall	403-975- 4527	1405 Dufferin Avenue	Calgary	AB	T2G 3Z4
West Edmonton	Kevin Truong	587-735- 6969	14808 116 Ave NW	Edmonton	AB	T5M 3G1
Ottawa East	Marc Latendresse	613-424- 8996	305 Vienna Terrace	Orleans	ON	K4A 0J2

North Surrey	Chris Wijaya	604-595- 3011	17163 102 Avenue	Surrey	ВС	V4N 3L4
Chatham-Kent	Scott &Jen Bowen	519-627- 9119	1405 Dufferin Avenue	Wallaceburg	ON	N8A 2W7

Former Franchisees During the year ended December 31, 2023

Fort Collins	Chad Kegans	970-566-5022	Loveland	CO
Altamonte Springs	Thomas Hewitt	321-972-4951	Longwood	FL
Clarksville	Kristi & Todd King	931-220-9444	Clarksville	TN
Waco	Paul & Paige Thompson	254-749-4520	Lorena	TX
Lehigh Valley	Placido Hernandez	610-249-0039	Easton	PA
Glenwood Springs	Joshua Rice	970-456-6832	New Castle	СО
Columbus	Stewart Jimmerson	706-478-7577	Columbus	GA
Greeley	Chad Kegans	970-566-5022	Loveland	СО

EXHIBIT G

TO THE DISCLOSURE DOCUMENT THE GLASS GURU ENTERPRISES, INC.

MULTI-STATE ADDENDA

ADDENDUM TO THE THE GLASS GURU ENTERPRISES, INC. UNIFORM FRANCHISE DISCLOSURE DOCUMENT

FOR THE STATE OF CALIFORNIA

- 1. THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT 14 DAYS PRIOR TO EXECUTION OF AGREEMENT.
- 2. The registration of this franchise offering by the California Department of Financial Protection and Innovation does not constitute approval, recommendation, or endorsement by the commissioner.
 - 3. Section 31125 of the California Corporations Code requires us to give you a disclosure document, in a form containing the information that the commissioner may by rule or order require, before a solicitation of a proposed material modification of an existing franchise.
- 4. Section 31512.1 of the California Corporations Code requires that any provision of a franchise agreement, franchise disclosure document, acknowledgement, questionnaire, or other writing, including any exhibits thereto, disclaiming or denying any of the following shall be deemed contrary to public policy and shall be void and unenforceable:
 - (a) Representations made by the franchisor or its personnel or agents to a prospective franchisee.
 - (b) Reliance by a franchisee on any representations made by the franchisor or its personnel or agents.
 - (c) Reliance by a franchisee on the franchise disclosure document, including any exhibit thereto.
 - (d) Violates any provision of this division.
- 5. Franchisees and all owners must sign a personal guarantee, making you and your spouse individually liable for your financial obligations under the agreement if you are married, the guaranteed will place you and your spouse's marital and personal assets at risk if your franchise fails.

You must sign a general release if you renew or transfer your franchise. California Corporations Code §31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code §\$31000 through 31516). Business and Professions Code §20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code §\$20000 through 20043).

Neither the franchisor, any person or franchise broker in ITEM 2 of the Disclosure Document is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. 78a et seq., suspending or expelling such persons from membership in such association or exchange.

- 6. ITEM 17 of the Disclosure Document is amended to add the following:
 - The California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination, transfer or non-renewal of a franchise. If the Franchise Agreement contains a provision that is inconsistent with the law, the law will control.

- The Franchise Agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law (11 U.S.C.A. Sec. 101 et seq.).
- The Franchise Agreement contains a covenant not to compete that extends beyond the term of the agreement. This provision might not be enforceable under California law.
- The Franchise Agreement contains a liquidated damages clause. Under California Civil Code Section 1671, certain liquidated damages clauses are unenforceable.
- The Franchise Agreement requires binding arbitration. The arbitration will occur at the forum indicated in ITEM 17 with the costs being borne by the non-prevailing party. Prospective franchisees are encouraged to consult legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of the Franchise Agreement restricting venue to a forum outside the State of California.
- The following URL address is for the franchisor's website: www.theglassguru.com

FRANCHISOR'S WEBSITE HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF BUSINESS OVERSIGHT. ANY COMPLAINTS CONCERNING THE CONTENT OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION AT www.dfpi.ca.gov.

FOR THE STATE OF ILLINOIS

- No action for liability under the Illinois Franchise Disclosure Act shall be maintained unless brought before the expiration of 3 years after the act or transaction constituting the violation upon which it is based, the expiration of 1 year after the franchisee becomes aware of facts or circumstances reasonably indicating that he may have a claim for relief in respect to conduct governed by the Act, or 90 days after delivery to the franchisee of a written notice disclosing the violation, whichever shall first expire.
- Illinois law governs the Franchise Agreement (without regard to conflict of laws), and jurisdiction and venue for court litigation shall be in Illinois.
- Any provision in the Franchise Agreement requiring a general release is void to the extent that such provision requires a waiver of compliance with the Illinois Franchise Disclosure Act.
- Any provision in the Franchise Agreement that designates jurisdiction or venue in a forum outside the State of Illinois is void, provided that a Franchise Agreement may provide for arbitration in a forum outside of Illinois.
- Any condition, stipulation, or provision purporting to bind any person acquiring any franchise to waive
 compliance with any provision of this Act or any other law of this State is void. This Section shall not
 prevent any person from entering into a settlement agreement or executing a general release regarding
 a potential or actual lawsuit filed under any of the provisions of this Act, nor shall it prevent the
 arbitration of any claim under the provisions of Title 9 of the United States Code.
- No statement, questionnaire or acknowledgement signed or agreed to by a franchisee in connection
 with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims
 under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance
 on behalf of the Franchisor. This provision supersedes any other term of any document executed in
 connection with the franchise.

ITEM 17 of the Disclosure Document is amended to add the following:

• The conditions under which a franchise can be terminated and your rights upon non-renewal, as well as the application by which you must bring any claims, may be affected by Sections 19 and 20 of the Illinois Franchise Disclosure Act of 1987,815 ILCS 705/19,705/20.

FOR THE STATE OF MARYLAND

- Any general releases you sign as a condition of renewal and/or assignment/transfer will not apply to claims arising under Maryland Franchise Registration and Disclosure Law.
- The agreement provides for termination upon insolvency. This provision might not be enforceable under federal bankruptcy law (11 U.S.C. Sections 101 et seq.), but we and you agree to enforce it to the extent the law allows.
- You may, subject to your obligations in the Franchise Agreement, bring an action in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.
- Texas law applies, except as otherwise required by applicable law for claims arising under the Maryland Franchise Registration and Disclosure Law.
- Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

Each provision set forth shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Maryland Franchise and Disclosure Law Maryland Franchise and Disclosure Law (MD CODE ANN., BUS. REG. §§ 14-201 through 14-233) are met independently without reference to this Addendum.

FOR THE STATE OF INDIANA

- 1. ITEM 8 of the Disclosure Document is amended to add the following:
 - Under Indiana Code Section 23-2-2.7-1(4), the franchisor will not obtain money, goods, services, or any other benefit from any other person with whom the franchisee does business, on account of, or in relation to, the transaction between the franchisee and the other person, other than for compensation for services rendered by the franchisor, unless the benefit is promptly accounted for, and transmitted by the franchisee.
- 2. ITEMS 6 and 9 of the Disclosure Document is amended to add the following:
 - The franchisee will not be required to indemnify franchisor for any liability imposed upon franchisor as a result of franchisee's reliance upon or use of procedures or products that were required by franchisor, if such procedures or products were utilized by franchisee in the manner required by franchisor.
- 3. ITEM 17 of the Disclosure Document is amended to add the following:
 - Indiana Code 23-2-2.7-1(7) makes unlawful unilateral termination of a franchise unless there is a material violation of the Franchise Agreement and termination is not in bad faith.
 - Indiana Code 23-2-2.7-1(5) prohibits a prospective general release of claims subject to the Indiana Deceptive Franchise Practices Law.
 - ITEM 17(r) is amended subject to Indiana Code 23-2-2.7-1(9) to provide that the post-term non-competition covenant shall have a geographical limitation of the territory granted to Franchisee.
 - ITEM 17(u) is amended to provide that arbitration between a franchisee and franchisor will be conducted in Indiana or a site mutually agreed upon.
 - ITEM 17(v) is amended to provide that Franchisees will be permitted to commence litigation in Indiana for any cause of action under Indiana Law.
 - ITEM 17(w) is amended to provide that in the event of a conflict of law, Indiana Law governs any cause of action that arises under the Indiana Disclosure Law or the Indiana Deceptive Franchise Practices Act.

FOR THE STATE OF MINNESOTA

- 1. ITEM 13 of the Disclosure Document is amended as follows:
 - As required by the Minnesota Franchise Act, Minn. Stat. Sec. 80C.12(g), Franchisor will reimburse the Franchisee for any costs incurred by the Franchisee in the defense of the Franchisee's right to use the Marks, so long as the Franchisee was using the Marks in the manner authorized by Franchisor, and so long as Franchisor is timely notified of the claim and given the right to manage the defense of the claim including the right to compromise, settle or otherwise resolve the claim, and to determine whether to appeal a final determination of the claim.
- 2. ITEM 17 of the Disclosure Document is amended as follows:
 - With respect to franchises governed by Minnesota law, the Franchisor will comply with Minn. Stat. Sec. 80C.14, Subs. 3, 4 and 5, which require, except in certain specified cases, that a Franchisee be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice of non-renewal of the Agreement.
 - ITEM 17 does not provide for a prospective general release of claims against Franchisor that may be subject to the Minnesota Franchise Law. Minn. Rule 2860.4400D prohibits a franchisor from requiring a franchisee to assent to a general release.
 - Minn. Stat. §80C.21 and Minn. Rule 2860.4400J prohibit us from requiring litigation to be conducted outside Minnesota. In addition, nothing in the Disclosure Document or Franchise Agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum or remedies provided for by the laws of the jurisdiction.

FOR THE STATE OF NEW YORK

- 1. All references made herein to an "Disclosure Document" shall be replaced with the term "Offering Prospectus" as used under New York Law.
- 2. The UFOC Cover Page is amended as follows:

REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT THE STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THIS DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND THE NEW YORK STATE DEPARTMENT OF LAW, 120 BROADWAY, NEW YORK, NEW YORK 10271-0332. INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT A OR YOUR PUBLIC LIBRARY FOR SOURCES OF INFORMATION.

THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE CIRCULAR. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS THAT ARE LESS FAVORABLE THAN THOSE DESCRIBED IN THIS CIRCULAR.

- 3. ITEM 3 is amended by the addition of the following language:
 - Neither franchisor, the franchisor's predecessor or an affiliate offering franchises under the franchisor's principal trademark, nor any person identified in ITEM 2 has any administrative, criminal or material civil action (or a significant number of civil actions irrespective of materiality) pending against them alleging a felony; a violation of a franchise, antitrust or securities law; fraud, embezzlement, fraudulent conversion; misappropriation of property; unfair or deceptive practices or comparable civil or misdemeanor allegations. In addition, neither franchisor nor any person identified in ITEM 2 has any pending actions, other than routine litigation incidental to the business, that are significant in the context of the number of franchisees and the size, nature or financial condition of the franchise system or its business operations.
 - Neither franchisor, the franchisor's predecessor or an affiliate offering franchises under the franchisor's principal trademark, nor any person identified in ITEM 2 has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10 year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been held liable in a civil action by final judgment or been the subject of a material compliant or other legal proceeding involving violation of any franchise law, antifraud or securities law; fraud, embezzlement, fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.
 - Neither franchisor, the franchisor's predecessor or an affiliate offering franchises under the franchisor's principal trademark, nor any person identified in ITEM 2 is subject to any injunctive or restrictive order or decree relating to the franchises, or any Federal, State or Canadian franchise, securities, antitrust, trade regulation or trade practice law, as a result of a concluded or pending action or proceeding brought by a publicagency.

4. ITEM 4 is amended to state that:

• Neither the franchisor, nor its predecessor, officers or general partner of the franchisor has, during the ten (10) year period immediately before the date of the Disclosure Document, has: (a) filed as a debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy

Code; (b) obtained a discharge of its debts under the Bankruptcy Code; (c) was a principal officer of any company or a general partner in any partnership that either filed as a debtor (or had filed against it) a petition to start action under the U.S. Bankruptcy Code or that obtained a discharge of its debts under the Bankruptcy Code during or within one (1) year after the officer or general partner of the franchisor held this position in the company or partnership.

- 5. ITEM 5 of the Disclosure Document is amended to add the following:
 - The Franchise Fee will be used to defray franchisor's costs in obtaining and screening franchisees, providing training, training materials, and assisting in opening the Franchise Business for business.
- 6. ITEMS 6 and 11 of the Disclosure Document areamended to add the following:
 - The franchisee will not be required to indemnify franchisor for any liability imposed on franchisor as a result of franchisee's reliance upon or use of procedures or products that were required by franchisor, if such procedures or products were utilized by franchisee in the manner required by franchisor.
- 7. ITEM 17 of the Disclosure Document is amended to add the following:
 - No general release shall be required as a condition of renewal, termination and/or transfer that is intended to exclude claims arising under the New York General Business Law, Article 3, Sections 687.4 and 687.5.
 - ITEM 17(d) is amended to provide that you may terminate the Agreement on any grounds available by law.
 - ITEM 17(j) is amended to state, that no assignment will be made except to an assignee who, in the good faith judgment of Franchisor, is able to assume our obligations under the Agreement.
 - ITEM 17(w) is amended to state that New York Law governs any cause of action that arises under the New York General Business Law, Article 33, Section 680-695.
- 8. Franchisor represents that this Disclosure Document does not knowingly omit anything or contain any untrue statements of a material fact.

FOR THE STATE OF VIRGINIA

In recognition of the restrictions contained in Section 13.1-564 of the Virginia Retail Franchising Act, the Franchise Disclosure Document for The Glass Guru Enterprises, Inc. for use in the Commonwealth of Virginia shall be amended as follows:

Additional Disclosure: The following statements are added to Item 17.h.:

Under Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any ground for default or termination stated in the Franchise Agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

FOR THE STATE OF WASHINGTON

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

Franchisor will defer collection of the initial franchise fee until the franchisor has fulfilled its initial pre-opening obligations to the franchisee and the franchisee is open for business. Please note that the initial fees for the purposes of the deferral include all initial franchisee fees described in Item 5 of the Franchise Disclosure Document

The franchisor may use the services of franchise brokers to assist it in selling franchises. A franchise broker represents the franchisor and is paid a fee for referring prospects to the franchisor and/or selling the franchise. Carefully evaluate any information provided by a franchise broker about a franchise. Do your own investigation by contacting the franchisor's current and former franchisees to ask them about their experience with the franchisor.

The undersigned does	s hereby acknowledge receip	ot of this addendum.	
Dated this	day of		_20
	FRANCHISOR	FRANCHISEE	

EXHIBIT H

TO THE DISCLOSURE DOCUMENT THE GLASS GURU ENTERPRISES, INC.

FINANCING DOCUMENTS

DO NOT DESTROY THIS NOTE:

When paid, this Note must be surrendered for Cancellation

SECURED PROMISSORY NOTE

Principal Amount of Note:	
Interest Rate of Note:	
Monthly Installment Payment of Note:	
FOR VALUE RECEIVED, the undersigned at	
at5550 Granite Parkway, Suite 280, Plan in writing. Payment(s) shall be made in	s Guru Enterprises, Inc., or its assigns (the "Franchisor"), o, TX 75024, or such place as Franchisor may designate lawful money of the United States of America, and in full principal amount first listed above with interest as
6%- Credit Score >700 7%- Credit Score between 650-699 8%- Credit Score between 600-649 9%- Credit Score <600	
	interest are payable in sixty (60) equal monthly o under this Note) in the amount first listed above
	y of each month thereafter until, 20, Il accrued but unpaid interest shall be all due and

Each payment shall be credited first to interest then due, and the remainder, if any, to principal, and interest shall thereupon cease to accrue upon the principal so credited. Should interest not be so paid, it shall thereupon bear like interest as the principal, but such unpaid interest so compounded shall not exceed an amount equal to simple interest on the unpaid principal at the maximum rate permitted by law. Franchisee shall have the right to prepay this note, in whole or in part, at any time without penalty.

<u>Event of Default:</u> In the event Franchisee fails to pay the unpaid principal balance of this note and interest due hereunder within ten (10) days after receipt of written notice by Franchisor ("Event of Default"), Franchisor may proceed to protect and enforce its rights by suit in equity and/or by action at law or by other appropriate proceedings. No delay on the part of Franchisor in the exercise of any power or right under this note, or under any other instrument executed pursuant

thereto shall operate as a waiver thereof, nor shall a single or partial exercise of any other power or right preclude further exercise thereof. Notwithstanding anything to the contrary contained herein, if an Event of Default shall occur, all payments thereafter made hereunder shall be applied, at the option of Franchisor, first to costs of collection, and then to principal.

It is hereby specially agreed that if this note is placed into the hands of an attorney for collection, or if proved, established, or collected in any court, Franchisee agrees to pay to Franchisor an amount equal to all expenses incurred in enforcing or collecting this Note, including court costs and reasonable attorneys' fees.

<u>Late Charge</u>: If any payment of principal or interest due under this Note is not received by Franchisee within ten (10) days after the date due, a late charge of Seventy-Five (\$75) dollars may be charged by Franchisor for the purposes of defraying expenses incident to handling such delinquent payments.

Franchisee specifically agrees that this late charge represents a reasonable sum considering all of the circumstances existing on the date of this Note and represents a fair and reasonable estimate of the costs that will be sustained by Franchisor due to the failure of Franchisee to make timely payments. The parties further agree that proof of actual damages would be costly or inconvenient. The late charges that accrue during any installment period shall be payable upon notice. Failure to demand or collect a late charge for any particular installment shall not waive Franchisor's right to collect late charges at any time.

Acceleration Upon Default: Upon failure to make any payment under this Note when due, the entire unpaid principal and all interest and any late charges under this Note shall immediately become due and payable at the election of the Franchisor, regardless of any prior forbearance, or notice(s) of such election being waived. During the existence of any such default, Franchisor may apply payments received on any amounts due hereunder, as said Franchisor may determine in its sole discretion.

Manner of Notification: Any notice to Franchisee provided for in this Note shall be given by personal delivery or by mailing such notice by first class mail addressed to Franchisee at the address stated in the first paragraph of this Note, or to such other address as Franchisee may designate by written notice to Franchisor. Any notice to Franchisor shall be given by personal delivery or by mailing such notice by first class mail to Franchisor at the address stated in the first paragraph of this Note, or at such other address as may be designated by written notice to Franchisee. Mailed notices shall be deemed delivered and received two (2) days after deposit in the United States mail in the manner provided for in this paragraph.

<u>Security:</u> Obligations under this Note *are personally guaranteed* and are secured by a Security Agreement, as expressed below, (collectively, the "*Related Documents*"). The Related Documents encumber all assets of Franchisee's Franchise Business.

Governing Law: This note shall be governed by and construed in accordance with the laws of the State of Texas and the United States of America. In the event any one or more of the provisions contained herein shall be invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not in any way be affected thereby. Franchisee expressly waives any right to trial by jury.

FRANCHISOR:

THE GLASS GURU ENTERPRISES, INC.

DATE:	
BY:	
NAME:	
TITLE:	
FRANCHISEE:	
Entity Name	
DATE:	
RV·	
ΝΔΜΕ·	
TITI F:	
· · · · · · · · · · · · · · · · · · ·	
DATE:	
BY:	
NAME:	
TITLE:	
DATE:	
BY:	
NAME:	
TITLE:	
DATE:	
BY:	
NAME:	
Title:	

SECURITY AGREEMENT

As a condition for Franchisor to agree to lend Franchisee the funds contemplated herein, Franchisee grants to Franchisor, a security interest in its property, tangible and intangible, including but not limited to: all accounts, now existing or subsequently arising; all contract rights of Franchisee, now existing or subsequently arising; all accounts receivable, now existing or subsequently arising; documents, and instruments related to accounts; inventory, furniture, fixtures, equipment, and supplies now owned or subsequently acquired; and the proceeds, products, and accessions of and to any and all of the foregoing (the "Collateral").

This security interest is granted to secure the debt evidenced by this note and agreement and all costs and expenses incurred by Franchisor in the collection of the debt.

Secured Parties Rights and Remedies

Upon written notice to Franchisee with an opportunity to cure any default, Franchisor shall have all the rights and remedies afforded a Franchisor under the Texas Uniform Commercial Code now in effect and all rights under the Related Documents including without limitation:

- (a) all rights and remedies provided in this Agreement;
- (b) all rights and remedies provided in the Note or other instrument secured by this Agreement; and,
- (c) all rights and remedies provided in any other applicable security agreement and under any applicable Texas law.

Among the rights and remedies of Franchisor are specifically included:

Right of Direct Collection: Franchisor may, at Franchisor's option, notify any account Franchisee of Franchisee or any obligor on any obligation payable to Franchisee and serving as collateral for this Agreement to make payment to Franchisor, as provided in the Texas Uniform Commercial Code.

Right to Control Proceeds: Franchisor may, at Franchisor's option, take control of any and all proceeds to which Franchisor is entitled under the Texas Uniform Commercial Code, and Franchisee agrees to cooperate fully in executing any commercially reasonable direction made in the exercise of this right.

Right to Take Possession of the Collateral: This shall include, but is not limited to:

- (a) Right to Take Possession: Franchisor shall have the right to take possession of the Collateral.
- (b) <u>Franchisee's Cooperation:</u> Franchisee will cooperate fully with Franchisor in the exercise of Franchisor's right to take possession of the collateral. This shall include, but is not limited to, an obligation to assemble and deliver the Collateral or some portion of the Collateral or some part or component of the collateral, on request of Franchisor, to a place designated by Franchisor where it shall be made available to Franchisor. Failure to cooperate shall
- (c) constitute a breach of this Agreement, and Franchisee shall be liable for any and all expenses incident to such failure of cooperation.

Right to Dispose of the Collateral: This shall include, but is not limited to:

- (a) <u>Right of Disposition:</u> Franchisor has a right to dispose of the Collateral by public or private proceeding and by way of one or more contracts. Such sale or other disposition of the Collateral may be made as a unit or in parcels and at any time and place and on any terms, provided only that disposition effected is commercially reasonable. Any actions so taken shall be considered commercially reasonable if made in the good-faith exercise of Franchisor's best business judgment in the matter.
- (b) <u>Place of Disposition:</u> Franchisor has the right to dispose of the Collateral from the premises of Franchisee, and to this end Franchisee agrees to cooperate fully in facilitating such a disposition, which may include, on request, the obligation to assemble the Collateral at some designated location of Franchisee where the Collateral shall be made available to prospective Franchisee.
 - Franchisor remains free to dispose of the Collateral from any other location, provided such location is commercially reasonable. Any location normally employed by Franchisor in the disposition of like goods shall be considered a commercially reasonable location.
- (c) Notice of Disposition: Franchisor shall give Franchisee notice of the time and place of any public sale of the Collateral or, in case of a private sale or disposition, of the time after which such private sale or disposition is intended. It shall be considered commercially reasonable if such notice is sent to Franchisee by first class mail 5 days prior to the public sale or the time after which the private sale or other disposition is intended.
 - There is no need for notice prior to disposition where the Collateral is perishable, or threatens to decline in value quickly, or where the Collateral is of a type customarily sold in a recognized market. In such event, the decision to so dispose of the Collateral shall be considered commercially reasonable provided only that it is made in the good-faith exercise of Franchisor's best business judgment in the matter.
- (d) <u>Proceeds of the Disposition:</u> The proceeds of any disposition shall be applied as provided in the Texas Uniform Commercial Code and shall include any and all expenses provided in this Agreement. They shall also include attorney fees and legal fees to the extent such items are not prohibited by law.
- (e) <u>Deficiency:</u> In the event of any deficiency, Franchisee shall be liable for such deficiency with interest at the rate of 10%, which shall accrue 5 days after receipt of written demand for payment of the deficiency.
- financing Statement: Franchisor, at its discretion, may file one or more financing statements under the Texas Uniform Commercial Code, naming Franchisee as a debtor and Franchisor as secured party and indicating the Collateral specified in this Promissory Note and Security Agreement.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement in duplicate originals by their duly authorized officers or representatives.

FRANCHISOR:	FRANCHISEE:	
THE GLASS GURU ENTERPRISES, INC.	Entity Name	
DATE: BY: NAME: TITLE:	DATE: BY: NAME: TITLE:	
	DATE: BY: NAME: TITLE:	
	DATE: BY: NAME: TITLE:	
	DATE: BY: NAME: TITLF:	

PERSONAL GUARANTY

Guaranty made, 20	, byw	hose
address is	("Guarant	tor"),
to The Glass Guru Enterprises, Inc. whose address is (the "Franchisor").	s 5550 Granite Parkway, Suite 280, Plano, TX 7	5024
Recitals		
Guarantor or shall be the shareholders and the spous	ses of the shareholders of franchisee (the " Franc l	hisee") ; and,
The Franchisee has entered into a promissory note (written above, and other documents, instruments payment to Franchisor of the franchise fee.		
Agreements		
In consideration of Franchisor entering into the No Guaranty unconditionally and irrevocably guarantee payment and performance of all obligations of the F Note, Guarantor will immediately perform all obligation, payment of all amounts due under the Note.	es to Franchisor, their successors and assigns, ranchisee under the Note. If Franchisee default	the prompt s under the
Guarantor will pay to Franchisor all expenses (includ Franchisor rights against Guarantor. This Guaranty v termination, bankruptcy or insolvency of Franchisee representatives, successors and assigns. If more than agrees that their liability is joint and several. Guarant requested by Franchisor in connection with this Guarantee of the connection with th	vill not be discharged or affected by the death, d or Guarantor and will bind Guarantor's heirs, pe one Guarantor has signed this Guaranty, each C tor agrees to provide any and all additional infor	issolution, ersonal Guarantor
THIS GUARANTY IS GOVERNED BY THE LAWS OF jurisdiction of THE state OF TEXAS or ANY federal c Guarantor EXPRESSLY WAIVES ANY RIGHT to A TR	ourt located IN TEXAS AND VENUE IN PLANO,	
FRANCHISOR:	FRANCHISEE:	
THE GLASS GURU ENTERPRISES, INC.	Entity Name	
DATE:	DATE:BY:	
BY:	NAME:	
NAME:	TITLE:	
TITLE:		

DATE: BY: NAME: TITLE:		
DATE:		
BY:		
NAME:		
TITLE:		
DATE:		
BY:		
NAME:		
TITLE:		

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered, or exempt from registration, as of the Effective Date stated below:

State	Effective Date
California	Pending
Hawaii	Pending
Illinois	Pending
Indiana	Pending
Maryland	Pending
Michigan	April 30, 2024
Minnesota	Pending
New York	Pending
North Dakota	Not Registered
Rhode Island	Pending
South Dakota	Not Registered
Virginia	Pending
Washington	Pending
Wisconsin	Pending

Other states may require registration, filing or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

ITEM 23

TO THE DISCLOSURE DOCUMENT THE GLASS GURU ENTERPRISES, INC.

RECEIPTS

RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If The Glass Guru Enterprises, Inc. offers you a franchise, it must provide this disclosure document to you 14 calendar days (and 10 business days in Michigan and Oregon) before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

New York and Rhode Island require that we give you this Disclosure Document at the earlier of the first personal meeting or 10 business days before you sign a binding agreement with or make a payment to us or an affiliate in connection with the proposed franchise sale.

If The Glass Guru Enterprises, Inc. does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the state agencies listed in Exhibit B.

The franchise seller for this offering is: Daniel Frey, 5550 Granite Parkway, Suite 280, Plano, TX 75024 (877) 654-850 Fill in additional franchise sellers below as applicable:	Э7
Issuer as Date: April 20, 2024	

Issuance Date: April 30, 2024

The Glass Guru Enterprises, Inc. authorizes the respective state agencies identified in Exhibit C to receive service of process for it in the particular state.

I received a disclosure document dated ______, 2024, that included the following Exhibits:

A. FRANCHISE AGREEMENT

- 1. SPECIFICS
- 2. GENERAL RELEASE
- 3. NONDISCLOSURE AND NON-COMPETITION AGREEMENT
- 4. GUARANTY AND ASSUMPTION OF OBLIGATIONS
- 5. HOLDERS OF LEGAL OR BENEFICIAL INTEREST IN FRANCHISEE; OFFICERS; DIRECTORS
- 6. ASSIGNMENT OF TELEPHONE NUMBERS AND INTERNTET TOOLS
- 7. ADDENDUM TO LEASE AND COLLATERAL ASSIGNMENT OF LEASE
- 8. MULTI-STATE ADDENDA
- B. LIST OF STATE ADMINISTRATORS
- C. LIST OF STATE AGENTS FOR SERVICE OF PROCESS
- D. TABLE OF CONTENTS OF CONFIDENTIAL OPERATIONS MANUAL
- **E. FINANCIAL STATEMENTS**
- F. LISTS OF CURRENT AND FORMER FRANCHISEES
- G. MULTI-STATE ADDENDA
- H. FINANCING DOCUMENTS

///

Continued on Page 2

Please sign and print your name below, date and return one copy of this receipt to THE GLASS GURU ENTERPRISES, I and keep the other for your records.		
Date of Receipt	Print Name, Title	
	Signature	
	Acorporation or LLC (State of incorporation or organization)	
	LAST PAGE	

RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If The Glass Guru Enterprises, Inc. offers you a franchise, it must provide this disclosure document to you 14 calendar days (and 10 business days in Michigan and Oregon) before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

New York and Rhode Island require that we give you this Disclosure Document at the earlier of the first personal meeting or 10 business days before you sign a binding agreement with or make a payment to us or an affiliate in connection with the proposed franchise sale.

If The Glass Guru Enterprises, Inc. does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the state agencies listed in Exhibit B.

The franchise seller for this offering is:

Daniel Frey, 5550 Granite Parkway, Suite 280, Plano, TX 75024 (877) 654-8507 Fill in additional franchise sellers below as applicable:

Issuance Date: April 30, 2024

The Glass Guru Enterprises, Inc. authorizes the respective state agencies identified in Exhibit C to receive service of process for it in the particular state.

I received a disclosure document dated ______, 2024, that included the following Exhibits:

A. FRANCHISE AGREEMENT

- 1. SPECIFICS
- 2. GENERAL RELEASE
- 3. NONDISCLOSURE AND NON-COMPETITION AGREEMENT
- 4. GUARANTY AND ASSUMPTION OF OBLIGATIONS
- 5. HOLDERS OF LEGAL OR BENEFICIAL INTEREST IN FRANCHISEE; OFFICERS; DIRECTORS
- 6. ASSIGNMENT OF TELEPHONE NUMBERS AND INTERNTET TOOLS
- 7. ADDENDUM TO LEASE AND COLLATERAL ASSIGNMENT OF LEASE
- 8. MULTI-STATE ADDENDA
- B. LIST OF STATE ADMINISTRATORS
- C. LIST OF STATE AGENTS FOR SERVICE OF PROCESS
- D. TABLE OF CONTENTS OF CONFIDENTIAL OPERATIONS MANUAL
- E. FINANCIAL STATEMENTS
- F. LISTS OF CURRENT AND FORMER FRANCHISEES
- G. MULTI-STATE ADDENDA
- H. FINANCING DOCUMENTS

///
///
///
Continued on Page 2

Please sign and print your name below, date and return one copy of this receipt to THE GLASS GURU ENTERPRISES, INC. and keep the other for your records.	
Date of Receipt	Print Name, Title
	Signature
	Acorporation or LLC (State of incorporation or organization)
	LAST PAGE